This Prospectus was approved by the Swedish Financial Supervisory Authority on 9 December 2025. This Prospectus shall be valid for twelve (12) months after the date of its approval. The obligation to supplement this Prospectus in the event of significant new factors, material mistakes or material inaccuracies will not apply when this Prospectus is no longer valid.



SWEDAVIA AB (publ)

PROSPECTUS FOR THE ADMISSION TO TRADING ON NASDAQ STOCKHOLM OF

SEK 500,000,000 Subordinated Perpetual Fixed Rate Capital Securities ISIN: SE0026853285

SEK 750,000,000 Subordinated Perpetual Floating Rate Capital Securities ISIN: SE0026853319

SEK 1,750,000,000 Subordinated Perpetual Floating Rate Capital Securities ISIN: SE0026853327

Important information

In this prospectus, the "Issuer", the "Company" and "Swedavia" means Swedavia AB (publ), Swedish Corporate ID No 556797-0818 and LEI code 529900ZERGNJBEQEX179. The "Group" means the Issuer with all its subsidiaries from time to time (each a "Group Company"). The "Joint Bookrunners" means Nordea Bank Abp and Skandinaviska Enskilda Banken AB (publ) (jointly the "Joint Bookrunners"). "Euroclear Sweden" refers to Euroclear Sweden AB. "Nasdaq Stockholm" refers to Nasdaq Stockholm AB. "SEK" refers to Swedish kronor. "M" refers to million(s).

The Issuer has issued a total of 1,400 subordinated perpetual floating rate callable capital securities with the first call date falling five (5) years after its issue date (the "Floating Rate Capital Securities") in the Total Nominal Amount of SEK 1,750,000,000, a total of 600 subordinated perpetual floating rate callable capital securities with the first call date falling seven (7) years after its issue date (the "Floating Rate Capital Securities") in the Total Nominal Amount of SEK 750,000,000 and a total of 400 subordinated perpetual fixed rate callable capital securities with the first call date falling seven (7) years after its issue date (the "Fixed Rate Capital Securities") in the Total Nominal Amount of SEK 500,000,000 on 19 November 2025 (the "Issue Date"). In this prospectus (the "Prospectus") and except as otherwise indicated, references to the "Capital Securities" are to the Floating Rate Capital Securities (NC5), Floating Rate Capital Securities (NC7) and/or the Fixed Rate Capital Securities, as the context requires, and references to the "Terms and Conditions" are to the terms and conditions of the Floating Rate Capital Securities (NC5)" below), the terms and conditions of the Floating Rate Capital Securities (NC7) (set out in section "Terms and Conditions of the Floating Rate Capital Securities (NC7)" below) and/or the terms and conditions of the Fixed Rate Capital Securities (NC7) (set out in section "Terms and Conditions of the Floating Rate Capital Securities (NC7)" below) and/or the terms and conditions of the Fixed Rate Capital Securities (NC7) when the same meanings when used in this Prospectus, unless expressly stated or otherwise follows from the context.

This Prospectus has been prepared for the admission to trading of the Capital Securities on Nasdaq Stockholm or another regulated market. This Prospectus does not contain and does not constitute an offer or a solicitation to buy or sell Capital Securities. This Prospectus is governed by Swedish law. The courts of Sweden have exclusive jurisdiction to settle any dispute arising out of or in connection herewith.

This Prospectus has been approved and registered by the Swedish Financial Supervisory Authority (*Finansinspektionen*) (the "**SFSA**") pursuant to the provisions of Article 20 of Regulation (EU) 2017/1129 of the European Parliament and of the Council of 14 June 2017 on the prospectus to be published when securities are offered to the public or admitted to trading on a regulated market, and repealing Directive 2003/71/EC (the "**Prospectus Regulation**").

This Prospectus may not be distributed in any jurisdiction where such distribution would require any additional prospectus, registration or measures other than those required under Swedish law, or otherwise would conflict with regulations in such jurisdiction. Persons into whose possession this Prospectus may come are required to inform themselves about, and comply with such restrictions. Any failure to comply with such restrictions may result in a violation of applicable securities regulations. The Capital Securities have not been, and will not be, registered under the United States Securities Act of 1933 or the securities laws of any state or other jurisdiction outside Sweden. Subject to certain exemptions, the Capital Securities may not be offered, sold or delivered within the United States or to, or for the account or benefit of, U.S. persons.

No person has been authorised to provide any information or make any statements other than those contained in this Prospectus. Should such information or statements nevertheless be furnished, it/they must not be relied upon as having been authorised or approved by the Issuer and the Issuer assumes no responsibility for such information or statements. Neither the publication of this Prospectus nor the offering, sale or delivery of any Capital Security implies that the information in this Prospectus is correct and current as at any date other than the date of this Prospectus or that there have not been any changes in the Issuer's or the Group's business since the date of this Prospectus.

Each potential investor in the Capital Securities must in light of its own circumstances determine the suitability of the investment. In particular, each potential investor should:

- (a) have sufficient knowledge and experience to make a meaningful evaluation of the Capital Securities, the merits and risks of investing in the Capital Securities and the information contained or incorporated by reference in this Prospectus or any applicable supplement;
- (b) have access to, and knowledge of, appropriate analytical tools to evaluate, in the context of its particular financial situation, an investment in the Capital Securities and the impact the Capital Securities will have on its overall investment portfolio;
- (c) have sufficient financial resources and liquidity to bear all of the risks of an investment in the Capital Securities;
- (d) understand thoroughly the terms of the Capital Securities and be familiar with the behaviour of any relevant financial markets; and
- (e) be able to evaluate (either alone or with the help of a financial adviser) possible scenarios for economic, interest rate and other factors that may affect its investment and its ability to bear the applicable risks.

Solely for the purposes of the product governance requirements set forth in Directive 2014/65/EU (as amended, "MiFID II"), the Joint Bookrunners (for the purposes of this and the following paragraph, the "manufacturers") have made a target market assessment in respect of the Capital Securities and have concluded that the target group for the Capital Securities is: (a) type of client: clients that are eligible counterparties, professional clients and retail clients, each as defined in MiFID II, (b) knowledge and experience: clients that are (i) informed investors, having one or more of the following characteristics: average knowledge of the relevant financial products (an informed investor can make an informed investment decision based on the offering documentation, together with knowledge and understanding of the specific risk factors/risks highlighted with them only) or some financial industry experience, and (ii) advanced investors, having one, or more of the following characteristics: good knowledge of the relevant financial products and transactions or financial industry experience or accompanied by professional investment advice or included in a discretionary portfolio service, (c) financial situation with a focus on the ability to bear losses: clients that have the ability to tie money up in a perpetual instrument and bear losses of up to 100% of the capital invested in the Capital Securities are typically willing to take more risk than deposit savings or unsubordinated debt securities and do not require a fully guaranteed income or return profile and (e) investment objective: clients whose investment objective is to generate growth of the invested capital and have a long term investment horizon.

Furthermore, the manufacturers have made an assessment as to the negative target market and concluded that the negative target market for the Capital Securities is clients that seek full capital protection or full repayment of the amount invested, are fully risk averse/have no risk tolerance or need a fully guaranteed income or fully predictable return profile. The manufacturers have made an assessment as to the distribution strategy for the Capital Securities, and have concluded that (i) all channels for distribution of the Capital Securities to eligible counterparties and professional clients are appropriate; and (ii) the following channels for distribution of the Capital Securities to retail clients are appropriate – investment advice, portfolio management, non-advised sales and pure execution services, subject to the distributor's suitability and appropriateness obligations under MiFID II, as applicable. Any person subsequently offering, selling or recommending the Capital Securities (a "distributor") should take into consideration the manufacturers' target market assessment. However, a distributor subject to MiFID II is responsible for undertaking its own target market assessment in respect of the Capital Securities (by either adopting or refining the manufacturers' target market assessment) and determining appropriate distribution channels, subject to the distributor's suitability and appropriateness obligations under MiFID II, as applicable.

The Capital Securities are not deemed to fall within the scope of Regulation (EU) No 1286/2014 (as amended) and no key information document (KID) has been prepared.

Forward-looking statements

The Prospectus contains certain forward-looking statements that reflect the Issuer's current views or expectations with respect to future events and financial and operational performance. The words "intend", "estimate", "expect", "may", "plan", "anticipate" or similar expressions regarding indications or forecasts of future developments or trends, which are not statements based on historical facts, constitute forward-looking information. Although the Issuer believes that these statements are based on reasonable assumptions and expectations, the Issuer cannot give any assurances that such statements will materialise. Because these forward-looking statements involve known and unknown risks and uncertainties, the outcome could differ materially from those set out in the forward-looking statement.

Factors that could cause the Issuer's and the Group's actual operations, result or performance to differ from the forward-looking statements include, but are not limited to, those described in "Risk factors". The forward-looking statements included in this Prospectus apply only to the date of the Prospectus. The Issuer undertakes no obligation to publicly update or revise any forward-looking statements, whether as a result of new information, future events or otherwise, other than as required by law. Any subsequent forward-looking information that can be ascribed to the Issuer and the Group or persons acting on the Issuer behalf is subject to the reservations in or referred to in this section.

TABLE OF CONTENTS

RISK FACTORS	4
OVERVIEW OF THE CAPITAL SECURITIES AND USE OF PROCEEDS	14
TERMS AND CONDITIONS OF THE FLOATING RATE CAPITAL SECURITIES (NC5)	20
TERMS AND CONDITIONS OF THE FLOATING RATE CAPITAL SECURITIES (NC7)	51
TERMS AND CONDITIONS OF THE FIXED RATE CAPITAL SECURITIES	85
DESCRIPTION OF SWEDAVIA AB (PUBL)	115
BOARD OF DIRECTORS, GROUP MANAGEMENT AND AUDITOR	118
LEGAL CONSIDERATIONS AND SUPPLEMENTARY INFORMATION	120
ADDRESSES	123

RISK FACTORS

The purpose of this section is to enable a potential investor to assess the relevant risks related to their potential investment in the Capital Securities in order to make an informed investment decision. The risk factors set forth below are therefore limited to risks that are material and specific to Swedavia AB (publ) (the "Issuer" and together with its direct and indirect subsidiaries, the "Group") and the Capital Securities in the opinion of the Issuer in accordance with Regulation (EU) 2017/1129 of the European Parliament and of the Council of 14 June 2017 (the Prospectus Regulation).

In this section, material risk factors are illustrated and discussed, including the Issuer's operational and industry risks, legal risks, financial risks as well as risks relating to the Capital Securities. The Issuer's assessment of the materiality of each risk factor is based on the probability of their occurrence and the expected magnitude of their negative impact. The description of the risk factors below is based on information available and estimates made on the date of this Prospectus.

The risk factors are presented in categories where the most material risk factors in a category are presented first under that category. Subsequent risk factors in the same category are not ranked in order of materiality or probability of occurrence. Where a risk factor may be categorised in more than one category, such risk factor appears only once and in the most relevant category for such risk factor.

Risks relating to the Issuer

Risks relating to the operations and the industry

The Issuer is exposed to risks related to the economic climate and geopolitical conditions

The Issuer is a state-owned company that owns, operates and develops a network of ten airports in Sweden. The Group's commercial revenues with respect to primarily Retail, Food & Beverage, Parking and Entry and Advertising fluctuates in relation to passenger trends. The Issuer is therefore to a significant extent dependant on demand for air transport and airport services, which, in turn, depends on the general economic climate, primarily in Sweden where the Group conducts all of its operations.

The global reach of the aviation industry means that it is sensitive to changes in the international political landscape as well as the global conditions in general, as these could quickly affect demand and conditions for air travel between different countries and regions. Current examples include uncertainties related to global trade wars affecting aircraft manufacturing and the whole supply chain, the war in Ukraine and the security situation in Europe, and the conflict between Israel and Hamas with subsequent uncertainty in the Middle East. In both the short and long term, the uncertain global situation has led to, and may continue to lead to, increased prices for energy, fuel and raw materials, which can negatively affect growth in the world economy, which in turn affects developments in the aviation industry. The aviation industry's dependence on aviation fuel further results in sensitivity to the global oil market, which can be negatively affected by, for example, prolonged conflicts or terrorist attacks in the Middle East or other oil-producing regions. There is a risk that geopolitical tensions and uncertainties in the world affect demand for air travel from both leisure and business travellers and potentially impair airlines' access to fuel or other inputs, which would adversely affect the Group's net revenue and results of operations.

After several years of high inflation, the situation has stabilised to more normal levels. Instead, it is now the low economic growth in Sweden and internationally that can affect the growth of air travel, both within Sweden and globally. High interest rates and high costs for households have led to lower disposable income and less money available for travelling, which has negatively affected traffic recovery.

Accordingly, macro-economic and geopolitical factors, including recession, may significantly affect the Issuer's business, financial position and results of operations.

The Issuer is exposed to risks related to recovery for the aviation industry after the COVID-19 pandemic

The Issuer was affected by the COVID-19 pandemic in 2020, which resulted in a significant decline in passenger volumes, leading to large economic losses for the Issuer. Since 2021, the Issuer has been in a recovery phase where air travel during 2024 amounted to approximately 80 per cent. of the passenger volume prior to the COVID-19 pandemic.

It is not possible to assess with certainty the future demand for air travel. Slow growth in demand for air travel can have a significant impact on the Issuer's business, financial position and results.

The Issuer is exposed to decreases in passenger volumes

Since the Issuer started conducting operations in 2010, passenger volumes have generally increased by 3 per cent. annually, and cargo volumes have also increased as a result of growing trade. However, during the fourth quarter of 2018, a trend break occurred and passenger volumes began to decrease somewhat at the Issuer's airports for the first time in many years. Total air travel decreased, with domestic travel decreasing more than international travel. After the COVID-19 pandemic, the trend has been that domestic travel has recovered to 60 per cent. of pre-COVID-19 pandemic levels, while international travel has recovered to 90 per cent. of pre-COVID-19 pandemic travel. The Issuer's assessment is that demand for domestic travel will continue to be weak.

During 2024, the Issuer's airports had a total of 32.5 million passengers, which is an increase of 1.2 per cent. compared to 2023. However, the number of international passengers increased by 5.0 per cent, whilst domestic passengers decreased by 9.5 per cent. This significant decline in domestic passenger traffic continued into the first half of 2025, with domestic traffic remaining weak. During the third quarter in 2025, the domestic traffic has shown a little growth compared to 2024. Since the pandemic, accessibility and passenger volumes in Sweden have not recovered as strongly as in the rest of Europe. The aviation tax (which was abolished from 1 July 2025) and the concept of "flight shame" have contributed to this development by suppressing demand compared to other regions. Furthermore, the pandemic has led to changed patterns for business travellers, with more meetings held online, which behaviour has remained after the pandemic to some extent, representing a potential structural shift in demand.

There is a significant risk that passenger volumes maintain a slower growth rate than expected or continue to decrease as a result of the environmental impact of the air transport industry and the anti-flying movement affecting passengers' travel behaviour. There is further a risk that passenger volumes are negatively affected by the Issuer's failure to identify new customer needs and, in a timely manner, to find innovative and attractive solutions to such needs in line with customer expectations.

Moreover, the number of passengers traveling to or from any of the Issuer's airports constitutes the basic condition for all revenue the Group generates. Approximately 63 per cent. of the Group's total revenues during 2024 consisted of revenues in the form of airport charges (including passenger and take-off charges). A change in the number of flight movements, the number of landing tons (i.e., the aircraft's weight) and the number of passengers has a direct impact on the Issuer's revenues. The Issuer's other revenues come from Commercial Services, which are revenues from commercial services in connection with the airports such as car parking and rental of premises for offices, logistics and granting of concessions for retail and restaurant operations, and these revenues are thus also linked to passenger volumes. As of 31 December 2024, a change in passenger growth of 1 per cent. would have an impact on the Issuer's total revenue of SEK 42 million on an annualised basis.

In the areas of Retail, Food & Beverage there is currently an ongoing transition work with the introduction of new business models in the agreements, with procured counterparties where the terms are adapted to the current traffic volume. In the event of major changes in traffic volumes, however, there are limited opportunities to secure revenues in the short term. During 2024, the Group's operating result increased to SEK 26 million (SEK -318 million), which was SEK 344 million higher compared to 2023.

Furthermore, in recent years, changes in regulations and additional elements related to safe travel have contributed to increased requirements, which along with changes in consumer behaviour has meant that the Issuer's capacity situation is even more strained. The regulatory requirements lead to adaptations of the airports and flows and affect the capacity of the Issuer's airports both in the short and long term. There is a risk that regulatory changes within, among other things, security and safety will further affect the airports' capacity, which may require additional investments.

Accordingly, since all of the Group's revenue is linked to passenger volumes, a significant decrease in passenger volumes would have a material adverse effect on the Group's business and results of operations.

The Issuer is exposed to disruptions related to serious incidents as well as general business interruptions

In addition to the risks described in the risk factor "The Issuer is exposed to risks related to recovery for the aviation industry after the COVID-19 pandemic" above, the occurrence of extraordinary events such as activity from volcanoes, other natural or man-made disasters or extreme weather conditions, in particular if such events occur in the Swedish airspace or otherwise in the region around any of the destinations reached from the Issuer's

airports, would typically adversely affect the Issuer and its business. Should climate changes result in more extreme weather conditions in Sweden, such as more severe and frequently occurring storms, wildfires, floods and snowstorms, there is a risk that such weather conditions lead to temporary, or permanently, closure of the Issuer's airports, technical interruptions, delayed and cancelled flights, reduced handling capacity and ground transport access, which would interfere with and disrupt the Issuer's operations. For example, the volcanic ash from the Eyjafjallajøkull volcano in Iceland in 2010 led to the entire Swedish airspace being shut down for a few days, as well as the majority of the European airspace, which implied that there was no air traffic from or to the Issuer's airports during that time. The disruption due to the volcanic eruption was estimated to adversely affect the Group's results by SEK 55 million. Accordingly, increased extreme weather conditions as a consequence of climate change would typically negatively affect the Issuer's business, financial condition and results of operations.

Furthermore, acts of terror, political uprisings, armed conflicts and other serious incidents, or any actual or perceived risk thereof, present a significant risk to the aviation industry as a result of consequential reduction in demand for air travel, limitations on the availability of insurance coverage, increase in insurance premiums and costs associated with additional security precautions and the imposition of flight restrictions over conflict zones. Terrorist attacks and terrorist threats occur within Europe on an irregular basis, and airports are considered a strategic important social function that are exposed to terrorist-related threats.

In order to carry out and maintain its operations, the Issuer is dependent on certain equipment, systems, utilities, and processes, which typically are vulnerable to, and can be disrupted by, among other things, equipment breakdown, internal failure, software error, physical damage, sabotage, cyberattacks or other events beyond the Issuer's or its suppliers' control. Any extensive outages or disruptions as a consequence of such events, for example, of the Issuer's equipment and systems to monitor the safety at its airports would also have a material adverse effect on the Issuer's operations. The Issuer has experienced operational incidents in recent years, including incidents involving damage to aircraft from jet bridges and refuelling operations, which have resulted in insurance claims, including one unsettled claim of EUR 1 million relating to a fuel lid incident in 2023.

The degree to which extreme weather conditions, natural disasters, terrorist attacks and other serious incidents as well as general business interruptions may affect the Issuer is uncertain and presents a highly significant risk to demand for travel and the Group's operations.

The Issuer is dependent on maintaining its airline customers

In 2024, the revenue from the Group's Aviation Business, which includes airport related services directed to airlines and ground handling services, amounted to approximately 63 per cent. of the Group's total revenue. The Group generates income from the Aviation Business by airport fees, comprising passenger and take-off charges. These charges are regulated and principally levied on the basis of passenger numbers, the number of flight movements and maximum total aircraft weight. Changes in the number of flight movements, the number of landing tons and the number of passengers directly affect the Issuer's net revenue, and accordingly the Issuer is dependent on maintaining its airline customers. However, the airline industry is highly susceptible to adverse economic development. Demand for airlines' services depends to a significant extent on general economic and industry conditions, such as unemployment levels, consumer confidence and the availability of consumer and corporate credit. Consequently, the airline industry tends to experience significant adverse financial results during general economic downturns.

The effects of the COVID-19 pandemic and the negative macroeconomic climate with increased inflation, rising interest rates, high energy prices, etc. have led to many of the Group's important airlines customers have faced material economic challenges. Several major airline customers are either in or have been in corporate restructurings. Among these, Scandinavian Airlines Systems ("SAS") can be particularly mentioned, which during the period July 2022 to August 2024 carried out a restructuring according to Chapter 11 proceedings (a legal process for financial restructuring) in the United States. The restructuring is now completed and has, among other things, meant that SAS has received new financing and new owners. The Swedish airline Braathens Regional Airlines ("BRA"), which is currently subject to company reorganisation proceedings, wound down its domestic scheduled traffic at the end of 2024 and became a subcontractor to SAS.

Swedavia is dependent on a few major customers and their traffic volumes, including SAS, Norwegian, and Ryanair. Compared to the period before the pandemic, dependency on major airline customers has decreased as the result of an ongoing strategy to attract a broader range of operators. Ryanair has grown and now holds a 10 per cent. market share, and more players have entered the market, including low-cost carriers such as Wizz Air. However, the Issuer remains exposed to concentration risk.

If any of the Group's important airlines customers were to face insolvency or bankruptcy, or for any other reason were to terminate their agreements with the Issuer or reduce the number of flights to and from the Issuer's airports, it would have significant negative consequences for the Group in terms of major traffic disruption and reduced traffic, which would negatively affect the Issuer's profitability. Accordingly, the loss of an important airline customer may significantly affect the Group's revenue and results of operations.

The Issuer is dependent on attracting and retaining key employees

The Issuer is dependent on being able to attract, develop, retain and incentivise employees possessing key skills with respect to, among other things, technical development, safety and security, environmental impact and working environment. For the financial year 2024, the average number of employees totalled 2,769 compared to 2,603 the year before.

The Board of Directors of the Issuer has appointed Mats Johannesson as the new President and CEO who will take up these position by early May 2026. Current CEO Jonas Abrahamsson will stay on as CEO until no later than the end of February 2026. Additionally, The Marketing and Sales Director will leave the company in April 2026, and the recruitment of a replacement has started. Whilst the Issuer does not anticipate material disruption from these changes, there is a risk that transitions in key management positions could temporarily affect strategic decision-making and operational continuity.

When the Issuer cannot deliver efficient flows and punctuality in accordance with passengers' expectations, it risks affecting customers' trust, which in turn can contribute to a negative impact on the Issuer's brand with reduced demand for the Issuer's services as a result.

If the Issuer fails to recruit and retain employees possessing the right skills, experience and values, the Issuer's airport development projects and continued development of the organisation risks being delayed or not successfully implemented, and it may be difficult to comply with more stringent security requirements, which in turn risks to adversely affect the Group's business and competitiveness.

Furthermore, to attract and retain the right personnel, the Issuer might need to increase its remuneration levels, which could adversely affect the Issuer's results of operations. In 2024, the Group's staff expenses totalled SEK 2,097 million. As of 31 December 2024, a 1 per cent. change in staff expenses would have an effect of SEK 21 million on an annualised basis. Conversely, if the Issuer were to offer excessively low remuneration levels, this might lead to employees choosing to terminate their employments, which could result in a lack of resources and competence and would adversely affect the Issuer's current and future operations.

Legal risks

The Issuer is exposed to environmental-related risks

The air transport industry is subject to numerous environmental regulations and laws due to the significant environmental impact of its operations. The primary source of the Group's emissions is exhaust gases from vehicles and the operation of terminals and other buildings. Another significant environmental impact is the perand polyfluoroalkyl substances (so called PFAS) pollutions to land and water primarily of oxygen-consuming substances from the runway de-icing chemicals treatment of runways and de-icing of aircraft.

The areas in the vicinity of the airports are also exposed to aviation noise. As a consequence of its environmental impact, the Issuer's airport operations are considered as environmentally hazardous activities under the Swedish Environmental Code (*Miljöbalken*), and accordingly, the Issuer must hold environmental permits for each of its airports in order to be permitted to operate. The Issuer further carries out certain water operations related to ponds, water treatment facilities and groundwater run-off subject to permit requirements pursuant to the Swedish Environmental Code. As of 31 December 2024, the Issuer was responsible for eight airports that require an environmental permit under the Swedish Environmental Code. For the other two airports, Luleå Airport and Ronneby Airport, the Swedish Armed Forces (*Försvarsmakten*) was responsible for the environmental permits. If the Issuer would fail to obtain required permits, or, to a significant extent, violates terms in permits, it could jeopardise the Group's ability to conduct its operations at the airports in question.

Annual follow-up on conditions for carrying out operations is done to monitor the Issuer's performance against these conditions. The Issuer had recorded 12 breaches out of a total of 156 environmental conditions for its ten airports in 2024. However, there have not been any legal breaches of the permits.

In addition, the Issuer may be held liable to investigate and rectify contamination and emissions at the Group's airports and on property owned or previously owned by the Group, irrespective of whether the Issuer has caused the contamination or whether the operation which caused the contamination was lawful at the time the

contamination occurred. The Issuer may also be subject to claims from public authorities, private individuals, companies or other parties who request compensation for alleged personal injury, property damage or damage to nature caused by contamination or hazardous substances resulting from the Issuer's airport operations. The degree to which environmental risks may affect the Issuer is uncertain and presents a highly significant risk to the Group's operations.

The Issuer is exposed to risks related to legal and administrative procedures

The Issuer's airport operations are primarily regulated by, among others, the Swedish Environmental Code, Swedish Civil Aviation Act, Swedish Civil Aviation Security Act, Swedish Aerodrome Ground Services Act and Swedish Act on Airport Charges. In addition, the Issuer has to comply with numerous other Swedish and European laws and regulations. Accordingly, the Issuer is, and may in the future be, a party to ongoing litigations and disputes as either plaintiff or defendant, or subject of investigations from public authorities. Disputes usually relate to claims arising in the ordinary course of business and, consequently, legal proceedings primarily occur in relation to construction contracts, rental contracts, other business contracts and procurement. Investigations from public authorities occur primarily in relation to aviation law, competition law and environmental law.

Swedavia has encountered contractual disputes with tenants following the termination of their leases. In total, three tenants have filed lawsuits against Swedavia, with the Court of Appeal ruling in Swedavia's favour in one of them, and two rent disputes ongoing. Swedavia has not paid any damages or penalties because of these disputes.

The European Commission announced a final decision in the fourth quarter of 2024 regarding the complaint made to the European Commission regarding the capital contribution of SEK 2,500 million that Swedavia received in 2020 from its owner, the Swedish state. The decision means that SEK 1,418 million of the original SEK 2,500 million can be retained by Swedavia whilst SEK 1,082 million must be repaid plus interest. In May 2025, Swedavia repaid the outstanding amount, including interest, in full.

It may be difficult to foresee the risk, or possible outcome, of litigations, disputes and matters, some of which may be unfavourable for the Issuer and materially adversely affect the Group's results of operations and financial position. In such cases, the Issuer risks to incur significant costs and there is a risk that the measures taken to protect against the impact of such costs may be insufficient. Adverse publicity in connection with legal proceedings further risks damaging the Issuer's reputation. Legal and administrative proceedings are unpredictable in nature, and there is a risk that the actual outcome may differ from the assessments that the Issuer has made. The degree to which legal and administrative procedures may affect the Issuer is uncertain, and presents a highly significant risk to the Issuer's business, results of operations and financial position.

The Issuer is exposed to tax-related risks

The Issuer is exposed for taxes relating to, among others, company tax, real estate tax and energy tax. There is a risk that the Issuer's understanding and interpretation of tax laws, tax treaties and other provisions is not correct in all respects. As a result of the economic losses during the pandemic, the Issuer has an accumulated loss carry forward. During the financial year 2024, the Issuer had a tax income which totalled SEK 27 million with an effective tax rate of 10.5 per cent. There is a risk that the Issuer's prior or present tax position may change as a result of decisions of tax authorities or changes in tax laws and regulations, possibly with retroactive effect, which may have an adverse effect on the Issuer's results of operations and financial position.

The aviation industry is further subject to extensive taxes, and increased taxes or policy decisions relating to the aviation industry could result in higher fares for passengers and declined demand for air travel. The national aviation tax introduced in Sweden in 2018, which according to the Issuer has had negative effects on Swedish growth in the aviation industry, was abolished from 1 July 2025 according to a decision by the Swedish Parliament on 27 November 2024. It is uncertain to what extent taxes or political decisions concerning the aviation industry may affect the Issuer is uncertain, and presents a significant risk to the Issuer's results of operations and financial position.

Financial risks

The Issuer is exposed to interest rate risks

Interest rate risk refers to the risk of a negative effect on results and cash flow in the event of a lasting change in the market rate, which primarily affects the Issuer's interest expenses through changed borrowing costs. As of 31 December 2024, the Issuer had external borrowings of SEK 11,900 million, which was 42 per cent. of the balance sheet. As of 31 December 2024, an increase in interest rates of 1 percentage point would have increased the Issuer's interest expense by SEK 9.5 million, which means that profit before tax would decrease by the same amount. As of 31 December 2023, the average interest rate was 2.8 per cent, and as of 31 December 2024, the

average interest rate was 2.8 per cent. Current lock-in periods and financial instruments are taken into account in calculating the sensitivity analysis. Further increased market interest rates that, to a significant extent, affect the Issuer's interest costs would have a material adverse effect on the Group's results of operations and financial position.

Further, interest changes also affect the Issuer's pension liability. As of 31 December 2024, the Group's provisions for pensions amounted to SEK 736 million. The Issuer's calculation of provisions for pensions is based on a number of assumptions. One of the most critical assumptions for the calculation is the discount rate, which is based on the market rate, and should the market rate decrease, the provisions for pensions would increase. Accordingly, a decrease in the market rate would probably imply that the Issuer must add funds to its provisions for pensions, which would adversely affect the Issuer's expenses and results of operations. As of 31 December 2024, the sensitivity analysis in relation to a change to the discount rate had the following outcome: if the discount rate would decrease with 0.5 per cent, this would increase the Issuer's obligations in the amount of SEK 46 million. If the discount rate would increase with 0.5 per cent, this would decrease the Issuer's obligations in the amount of SEK 42 million.

The Issuer is exposed to credit risks

Credit risk refers to the risk of losses if counterparties fail to perform their obligations. The Issuer's maximum exposure to credit risks corresponds to the book value of financial assets, including derivatives with a positive market value, which as of 30 September 2025 amounted to SEK 1,879 million, of which SEK 127 million related to trade receivables (of which SEK 26 million had fallen due). If the Issuer is unable to collect its trade receivables, it risks having a material adverse effect on its results of operations and financial position.

The exposure to credit risks, in addition to the Issuer's trade receivables, arises when the Group invests liquid assets as well as in the form of counterparty risks when the Group enters into an agreement on financial instruments with banks. There is also a risk that the measures taken to offset the Issuer's credit risks are insufficient or ineffective and that the Group fails to implement and manage any hedging arrangement, which risks having a material adverse effect on its results of operations and financial position.

The Issuer is exposed to liquidity and refinancing risks

Liquidity risk refers to the risk of not having access to cash and cash equivalents or undrawn lines of credits to in order to perform payment obligations. Refinancing risk refers to the risk that refinancing of the Group's capital requirements and outstanding borrowings is rendered more difficult or more expensive.

There is a risk that the Issuer is unable to repay debts as they fall due (as of 30 September 2025, the average maturity on the Issuer's borrowings was 3,2 years (3,2)), which may be due, among other things, to the Group being unable to generate sufficient cash flow from ongoing operations. The ability to obtain financing through loans with favourable terms or at all depends on several factors beyond the Issuer's control, including the conditions prevailing at the time in the international credit and capital markets. The ability to secure capital financing through loans on favourable terms or at all depends on number of factors beyond the Issuer's control, including conditions prevailing at the time on the international credit and capital markets. The Issuer's ability to sell assets (insofar as it is permitted pursuant to terms in loan agreements) and to use the revenues to repay maturing debts also depends on factors beyond the Issuer's control, including there being willing buyers and the value of the assets. If the Issuer fails to repay its existing or future debts, to renew or refinance existing or future credit facilities on acceptable terms or at all, or to perform existing financial obligations, this would have a material adverse effect on the Group's liquidity, results of operations and financial position.

If the Issuer does not succeed in repaying its existing or future debts, renewing or refinancing existing or future credit facilities on acceptable terms or at all, or fulfilling existing financial obligations, this would have a material negative impact on the Group's liquidity, results and financial position. Furthermore, and as described above in the risk factor "*The Issuer is exposed to risks related to legal and administrative proceedings*" above, should an investigation of the capital injection received by the Issuer result in an unfavourable decision against the Issuer, the Issuer could be under an obligation to repay the capital injection which could materially adversely affect the Issuer's financial position. The degree to which liquidity and refinancing risks may affect the Issuer is uncertain and presents a significant risk to the Group's liquidity and financial position.

The Issuer is exposed to exchange rate risks

Exchange rate risk is defined as the risk that movements in currency prices will have a negative impact on the Group's profit, financial position and/or cash flow. The Group's reporting currency is SEK, but the Issuer generates income and purchasing costs in a number of currencies, which exposes the Group to currency risks and exchange

rate fluctuations that affect the Group's operating profit. The currencies to which the Issuer is primarily exposed are EUR and USD.

The Issuer shall, according to its internal policies, hedge its net positions over a countervalue of SEK 5 million with a repayment date within two years by appropriate hedge instruments up to an amount of at least 75 per cent. of the aggregated value. This implies that there is a theoretical exposure of 25 per cent. in total, which per 31 December 2024 constituted a nominal amount of SEK 54 million. An unfavourable exchange difference of 10 per cent. in such scenario would result in a negative effect on the Issuer's result in the amount of SEK 5 million.

The Issuer is also exposed to risks where suitable hedge instruments for the types of risk to which the Group is exposed are not available at a reasonable cost or at all. In addition, there are risk related to the use of such hedging instruments, for example that the Issuer does not have the possibility to use favourable exchange rate conversions. Hedging may thus lead to large losses. These losses may arise for various reasons, for example a counterparty failing to perform its obligations under an applicable hedging agreement, shortcomings in the agreement, or non-compliance with the Issuer's internal hedging policies and procedures, or such policies and procedures failing to function as they should. The degree to which exchange rate risks may affect the Issuer is uncertain and presents a significant risk to its results of operations and financial position.

The Issuer is exposed to risks relating to changes in accounting standards

The Issuer prepares its financial statements in accordance with International Financial Reporting Standards ("IFRS") and International Accounting Standards ("IAS") as adopted by the EU. Future changes in IFRS or IAS may lead to significant changes in the reported financial statements of the Issuer.

The Capital Securities are treated as equity pursuant to IAS 32 Financial Instruments, and consequently, the Capital Securities will not be accounted for as financial liabilities. However, in June 2018, the International Accounting Standard Board (the "IASB") published the discussion paper "Financial Instruments with Characteristics of Equity" (the "Discussion Paper"). The Discussion Paper sets out the IASB's preferred approach to classification of a financial instrument such as the Capital Securities, from the perspective of an issuer, as a financial liability or an equity instrument. The changes to the accounting standards addressed in the Discussion Paper would, if implemented, likely lead to financial instruments such as the Capital Securities being classified as financial liabilities rather than equity as per the current accounting standards. In February 2021, IASB tentatively decided not to change how such instruments should be classified as proposed in the Discussion Paper but instead look to develop presentation and disclosure requirements in relation to them. If the changes to the accounting standard proposed in the Discussion Paper or any similar proposal would be implemented, it would most likely lead to the Capital Securities being classified as financial liabilities of the Issuer which in turn would have a negative impact on the Issuer's financial position and the amount of financial liabilities. Further, if the Capital Securities are classified as financial indebtedness it would lead to the occurrence of an Accounting Event (see "Redemption of the Capital Securities" below).

Credit ratings may not reflect all risks associated with the Capital Securities

One or more independent credit rating agencies may assign credit ratings to the Issuer. The ratings may not reflect the potential impact of all risks related to structure, market, additional factors discussed above, and other factors that may affect the value of the Capital Securities. A credit rating is not a recommendation to buy, sell or hold securities and may be revised, suspended or withdrawn by the rating agency at any time. In general, European regulated investors are restricted under Regulation (EC) No. 1060/2009 (as amended) (the "CRA Regulation") from using credit ratings for regulatory purposes, unless such ratings are issued by a credit rating agency established in the EU and registered under the CRA Regulation (and such registration has not been withdrawn or suspended), subject to transitional provisions that apply in certain circumstances whilst the registration application is pending. Such general restriction will also apply in the case of credit ratings issued by non-EU credit rating agencies, unless the relevant credit ratings are endorsed by an EU-registered credit rating agency or the relevant non-EU rating agency is certified in accordance with the CRA Regulation (and such endorsement action or certification, as the case may be, has not been withdrawn or suspended, subject to transitional provisions that apply in certain circumstances). The list of registered and certified rating agencies published by the European Securities and Markets Authority ("ESMA") on its website in accordance with the CRA Regulation is not conclusive evidence of the status of the relevant rating agency included in such list, as there may be delays between certain supervisory measures being taken against a relevant rating agency and the publication of the updated ESMA list.

Risks relating to the Capital Securities

Risks relating to the nature of the Capital Securities

The Capital Securities are subordinated to most of the Issuer's liabilities

The Capital Securities are intended to constitute deeply subordinated debt obligations of the Issuer. This means that if the Issuer is subject to any dissolution, winding-up, liquidation, company restructuring (företagsrekonstruktion), administrative or other bankruptcy or insolvency proceedings, the Holders normally receive payment after all other creditors have been paid in full. If the Issuer is declared insolvent and a winding up is initiated, it will be required to pay the holders of senior debt and meet its obligations to all its other creditors (including unsecured creditors of all Subordinated Indebtedness) in full before it can make any payments on the Capital Securities. If this occurs, there is a risk that the Issuer does not have sufficient assets remaining after these payments to pay amounts due under the Capital Securities, which presents a significant risk for a single Holder.

In the event of an Issuer Re-Construction (as defined in the Terms and Conditions), unsecured debt could be subject to a mandatory write-down provided that a qualified majority of the unsecured creditors has approved such write-down. There is a risk that claims in respect of the share capital of the Issuer are not subject to loss absorbing measures under an Issuer Re-Construction. Consequently there is a significant risk that the Capital Securities will be, partly or completely, written-off, resulting in Holders not recovering their investment in Capital Securities upon an Issuer Re-Construction and, thereby, presents a significant risk for the Holders.

In the event of a shortfall of funds on an Issuer Winding-up or Issuer Re-Construction, there is a significant risk that a Holder of the Capital Securities will lose all or most of its investment and will not receive any return of the principal amount or any accrued and unpaid interest (including any Deferred Interest). By virtue of such subordination, payments to a Holder will, in the events described in the relevant Terms and Conditions, only be made after all obligations of the Issuer resulting from higher ranking claims have been satisfied. A Holder may therefore recover less than the Holders of unsubordinated or other subordinated liabilities of the Issuer that are senior to the Capital Securities.

Subject to applicable law, no Holder may exercise, claim or plead any right of set-off, compensation or retention in respect of any amount owed to it by the Issuer in respect of, or arising under or in connection with the Capital Securities and each Holder shall, by virtue of its holding of any Capital Security, be deemed to have waived all such rights of set-off, compensation or retention.

Holders of the Capital Securities have very limited rights in relation to the enforcement of payments on the Capital Securities

The Holders' rights of enforcement in respect of payments under the Capital Securities are subject to significant limitations. If a default is made by the Issuer for a period of 30 days or more in relation to the payment of any interest, principal or premium in respect of the Capital Securities which is due and payable, the rights of the Holders in respect of the Capital Securities are limited to instituting proceedings for an Issuer Winding-up, and the Holders may prove and/or claim in respect of the Capital Securities in an Issuer Winding-up. In addition, the Holders shall not be entitled to accelerate payments of interest or principal under the Capital Securities in any circumstances outside an Issuer Winding-up which, presents a risk that the Holders do not recover their investments in Capital Securities.

Furthermore, whilst the Holders may institute other proceedings against the Issuer to enforce the terms of the Capital Securities, the Issuer shall not, by virtue of such proceedings, be obliged to pay any sum or sums sooner than the same would otherwise have been payable by it. Accordingly, the Holders' rights of enforcement in respect of payments under the Capital Securities are very limited, thus presenting a significant risk for a single Holder.

The Capital Securities have no maturity date

The Capital Securities are perpetual meaning that the Capital Securities have no specified maturity date. The Issuer is not obliged to redeem the Capital Securities at any time and Holders have no option to redeem the Capital Securities at any time. The Issuer may only redeem the Capital Securities in the circumstances described in Clause 11 (*Redemption and repurchase of the Capital Securities*) of the Terms and Conditions.

Holders are thus required to bear financial risks of the investment in the Capital Securities for a long period of time and may not recover their investment before a redemption of the Capital Securities (if any) at the discretion of the Issuer (in particular if there is no active trading on the secondary market). Accordingly, there is a risk that Holders may lose the whole, or parts of, their investment in the event the Issuer chooses to not redeem the Capital Securities.

The Issuer may defer interest payments

The Issuer may, at any time and in its sole discretion (except on an Interest Payment Date on which the Capital Securities are to be redeemed), elect to defer any Interest Payment, in whole or in part, which would otherwise be paid on any Interest Payment Date. If interest is deferred in accordance with the Terms and Conditions, the Issuer has no obligation to make such payment on the relevant Interest Payment Date and any such non-payment of interest does not constitute a default or any other breach of obligations under the Capital Securities, and risks leading to the Holders not receiving a return on their investment.

Any actual or anticipated deferral of interest payments will be likely to have an adverse effect on the market price of the Capital Securities. In addition, as a result of such interest deferral provisions of the Capital Securities, the market price of the Capital Securities may be more volatile than the market prices of other debt securities on which interest accrues that are not subject to the above provisions and may be more sensitive generally to adverse changes in the Issuer's financial condition. The degree to which the market price of the Capital Securities may vary is uncertain and presents a significant risk to the value of, and the possibility to trade, the Capital Securities.

Redemption of the Capital Securities

Upon the occurrence of an Accounting Event, a Substantial Repurchase Event, a Tax Event, Rating Methodology Event, a Withholding Tax Event, or a Change of Control Event, the Issuer may redeem the Capital Securities in whole, but not some only, at any time together with any Deferred Interest and any accrued and unpaid interest. If the Capital Securities are redeemed, Holders are entitled to receive a redemption amount, which may exceed the nominal amount of the Capital Securities. There is a risk that the market value of the Capital Securities is higher than the amount received at redemption and that it is not possible for Holders to reinvest such proceeds at an effective interest rate as high as the interest rate on the Capital Securities and may only be able to do so at a significantly lower rate. Accordingly, this presents a significant risk for a single Holder.

Substitution or variation of the Capital Securities

Subject to Clause 11.6 (Substitution or variation due to a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event) of the Terms and Conditions, the Issuer may, at its option and without the permission or approval of the relevant Holders, elect to substitute or vary the terms of all (but not some only) outstanding Capital Securities for, or so that they become or remain, as applicable, Qualifying Securities if a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event occurs.

There is a risk that, due to the particular circumstances of each Holder, any Qualifying Securities will be less favourable to each Holder in all respects or that a particular Holder would not make the same determination as the Issuer as to whether the terms of the relevant Qualifying Securities are not materially less favourable to Holders than the terms of the relevant Notes (as reasonably determined by the Issuer in consultation with an independent investment bank, independent financial adviser or legal counsel). The substitution or variation of the Capital Securities may thus lead to changes in the Capital Securities that have effects that are less favourable to the Holders. The Issuer bears no responsibility towards the Holders for any adverse effects of such substitution or variation (including, without limitation, with respect to any adverse tax consequence suffered by any Holder). The degree to which the Capital Securities may be substituted or varied is uncertain and presents a highly significant risk on the price of, and/or the market for, the Capital Securities, and that there might be adverse tax consequences for any Holders arising from such substitution or variation.

Risks relating to the admission of the Capital Securities to trading

There has been no active trading market for the Capital Securities and an established trading market for the Capital Securities may not develop

Pursuant to the Terms and Conditions, the Issuer shall use its best efforts (without assuming any legal or contractual obligation) to apply for the Capital Securities to be admitted to trading on the corporate bond list of Nasdaq Stockholm (or another Regulated Market) but there is a risk that such application will not be accepted or that the Capital Securities will not be so admitted. A failure to admit the Capital Securities to trading risks having a negative impact on the market value of the Capital Securities and would not provide a Holder with a right of prepayment (put option) of its Capital Securities.

Prior to any admission to trading, there has been no public market for the Capital Securities. Even if an admission to trading will occur, there is a risk that an active trading market for the Capital Securities will not evolve or, if evolved, will not be sustained. The nominal amount of the Capital Securities may not be indicative of their market value after being admitted for trading on Nasdaq Stockholm (or another Regulated Market). Furthermore, following an admission to trading of the Capital Securities, the liquidity and trading price of the Capital Securities

may vary substantially as a result of numerous factors, including market fluctuations and general economic conditions and irrespective of the performance of the Issuer and the Group. Capital Securities may be acquired by the Issuer and subsequently be cancelled by the Issuer, provided that the aggregate principal amount of the Capital Securities subject to such cancellation represents eighty (80) per cent. or more of the aggregate principal amount of the Capital Securities issued (which shall include, for these purposes, any Subsequent Capital Securities). The degree to which the liquidity and the trading price of the Capital Securities may vary is uncertain, and risks leading to the Holders not recovering their investments in the Capital Securities. In addition, transaction costs in any secondary market may be high, which also presents a risk to the Holders not recovering their investments in the Capital Securities.

Therefore, Holders may not be able to sell their Capital Securities at the desired time or at a price level that will provide them with a yield comparable to similar investments that have a developed secondary market. Accordingly, the purchase of Capital Securities is suitable only for Holders who can bear the risks associated with a lack of liquidity in the Capital Securities and the financial and other risks associated with an investment in the Capital Securities. The degree to which the market value of the Capital Securities may vary is uncertain, and presents a significant risk for Holders' investment in the Capital Securities.

Further, if the Issuer fails to procure the admission to trading in time, Holders holding Capital Securities on an investment savings account (*investeringssparkonto*) will no longer be able to hold the Capital Securities on such account and, thus, presents a significant risk to such Holder's tax situation.

Other risks relating to the Capital Securities

European Benchmarks Regulation

In order to ensure the reliability of reference rates (such as STIBOR), legislative action at EU level has been taken. Hence, the so-called Benchmark Regulation (Regulation (EU) 2016/1011) entered into force on 1 January 2018. The Benchmark Regulation regulates the provision of reference values, reporting of data bases for reference values and use of reference values within the EU. The effects of the Benchmark Regulation so far are difficult to assess. However, there are future risks that the Benchmark Regulation affects how certain reference rates are determined and how they are developed. This in conjunction with increased administrative requirements is likely to lead to a reduced number of entities involved in the determination of reference rates, which, in such case, would lead to a certain reference interest ceasing to be published. If this is the case for STIBOR, and e.g. the relevant fall-back solution evident from the Terms and Conditions should not work properly or negatively for either or both of the Issuer or the Holders, this may lead to difficulties with determination and calculating interest which in turn risks leading to costly and time consuming discussions (and maybe even disputes) in respect of the matter, which in each case risks having an adverse effect on the Capital Securities, the Issuer and/or the Holders. There is furthermore a risk that such alternative interest calculation mechanism results in interest payments less advantageous for the investors compared to similar securities investments, or that such interest payment do not meet market interest rate expectations.

OVERVIEW OF THE CAPITAL SECURITIES AND USE OF PROCEEDS

This section is only intended to serve as an introduction to the Capital Securities. Any decision to invest in Capital Securities shall be based on an assessment of all information contained in this Prospectus as well as all documents incorporated herein by reference, the Terms and Conditions of the Floating Rate Capital Securities (NC5) are found on page 20 in this Prospectus, the Terms and Conditions of the Floating Rate Capital Securities (NC7) are found on page 51 in this Prospectus, and the Terms and Conditions of the Fixed Rate Capital Securities are found on page 85 in this Prospectus.

The Capital Securities

The Capital Securities have a Nominal Amount of SEK 1,250,000 each and are denominated in SEK. The aggregate nominal amount of the Capital Securities is SEK 3,000,000,000. In total, 2,400 Capital Securities have been issued. All Capital Securities are issued on a fully paid basis at an issue price of 100 per cent. of the Nominal Amount.

ISIN and common code

The Floating Rate Capital Securities (NC5) have been allocated the ISIN code SE0026853327. The Floating Rate Capital Securities (NC7) have been allocated the ISIN code SE0026853319. The Fixed Rate Capital Securities have been allocated the ISIN code SE0026853285. The Capital Securities will also be allocated a trading code upon admission to trading. Such trading code has not been allocated at the date of this Prospectus.

Form of the Capital Securities

The Capital Securities are issued in dematerialized book-entry form and registered on a Securities Account (*värdepapperskonto*) on behalf of the relevant Holder. Hence, no physical securities have been issued. The Capital Securities are registered in accordance with the Central Securities Depositories and Financial Instruments Accounts Act (*lagen* (1998:1479) om värdepapperscentraler och kontoföring av finansiella instrument) and registration requests relating to the Capital Securities shall be directed to an Account Operator. The Capital Securities are governed by Swedish law and are unilateral debt instruments intended for public trading (*ensidig skuldförbindelse avsedd för allmän omsättning*) as set out in Chapter 1, section 3 of the Central Securities Depositories and Financial Instruments Accounts Act.

The Capital Securities are freely transferable, but the Holders may be subject to purchase or transfer restrictions with regard to the Capital Securities, as applicable, under local laws to which a Holder may be subject. Each Holder must ensure compliance with such restrictions at its own cost and expense.

Status of the Capital Securities

The Capital Securities are constituted by the Terms and Conditions. The Issuer undertakes to make payments in relation to the Capital Securities and to comply with the Terms and Conditions.

The Capital Securities, including the obligation to pay interest thereon, constitute direct, general, unconditional, unsecured and subordinated obligations of the Issuer. The rights and claims of the Holders in respect of the Capital Securities against the Issuer, are subordinated as described in Clause 3 (*Status of the Capital Securities*) of the Terms and Conditions.

In short, this means that (i) in the event of an Issuer Winding-up, the right of the Holders to receive payments in respect of the Capital Securities will rank *pari passu* without any preference among themselves and with any present or future claims in respect of obligations of the Issuer in respect of Parity Securities; and (ii) in the event of a company re-construction (*företagsrekonstruktion*) of the Issuer under the Company Reorganisation Act (*lagen* (2022:964) om *företagsrekonstruktion*) the right of the Holders to receive payments in respect of the Capital Securities will rank *pari passu* without any preference among themselves and with any present or future claims in respect of obligations of the Issuer in respect of Parity Securities.

See further in Clause 3 (Status of the Capital Securities) of the Terms and Conditions.

Issuance, repurchase, redemption and calculation

Issue Date and Final Maturity Date

The Capital Securities were issued on 19 November 2025. The Capital Securities are perpetual and have no specified maturity date. The Issuer may only redeem the Capital Securities in the circumstances described in

Clause 11 (*Redemption and repurchase of the Capital Securities*) of the Terms and Conditions, as described below. Capital Securities are not redeemable at the option of the Holders at any time.

Purchase of Capital Securities by Group Companies

The Issuer or any other Group Company may, subject to applicable regulations, at any time and at any price purchase Capital Securities on the market or in any other way.

Capital Securities held by the Issuer or a Group Company may at such Group Company's discretion be retained or sold or, if held by the Issuer, cancelled by the Issuer, provided that the aggregate principal amount of the Capital Securities subject to such cancellation represents eighty (80) per cent. or more of the aggregate principal amount of the Capital Securities issued.

Voluntary total redemption (call option)

The Issuer may redeem all, but not some only, of the outstanding Capital Securities in full on the First Call Date, which is the date falling five years after the Issue Date (in respect of the Floating Rate Capital Securities (NC5)) or seven years after the First Issue Date (in respect of the Floating Rate Capital Securities (NC7) and the Fixed Rate Capital Securities), or on any Interest Payment Date thereafter at an amount per Capital Security equal to 100 per cent. of the Nominal Amount together with any Deferred Interest and any other accrued and unpaid interest up to (and including) the Redemption Date.

Voluntary redemption due to a Special Event

Upon the occurrence of a Special Event which is continuing the Issuer may, subject to Clause 12 (*Preconditions to Special Event Redemption or Change of Control Event Redemption*) of the Terms and Conditions, redeem all, but not some only, of the Capital Securities at an amount equal to:

- (a) 101 per cent. of their principal amount, where such redemption occurs before the First Call Date; or
- (b) 100 per cent. of their principal amount, where such redemption occurs on or after the First Call Date,

together, in each case, with any Deferred Interest and any other accrued and unpaid interest up to (and including) the Redemption Date.

Change of Control Event

Upon the occurrence of a Change of Control Event, the Issuer may, no later than the Change of Control Step-up Date, subject to Clause 12 (*Precondition to Special Event Redemption or Change of Control Event Redemption*) of the Terms and Conditions, redeem all, but not some only, of the Capital Securities at an amount equal to:

- (a) 101 per cent. of their principal amount, where such redemption occurs before the First Call Date; or
- (b) 100 per cent. of their principal amount, where such redemption occurs on or after the First Call Date,

together, in each case, with any Deferred Interest and any other accrued and unpaid interest up to (and including) the Redemption Date.

Substitution or variation due to a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event

Upon the occurrence of a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event, the Issuer may, at its option but subject to giving notice in accordance with Clause 11.7 (*Notice of redemption, substitution or variation*), substitute or vary the terms of all (but not some only) of the outstanding Capital Securities without any requirement for the consent or approval of the Holders, so that they become or remain, as applicable, Qualifying Securities, provided that such substitution or variation does not itself give rise to any right of the Issuer to redeem, substitute or vary the terms of the Capital Securities in accordance with Clause 11.6 (*Substitution or variation due to a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event*) in relation to the Qualifying Securities so substituted or varied.

Notice of redemption

Redemption in accordance with Clauses 11.3 (*Voluntary total redemption (call option*)), 11.4 (*Voluntary redemption due to a Special Event*) or 11.5 (*Voluntary redemption due to a Change of Control Event*) of the Terms and Conditions shall be made by the Issuer giving not less than 30, and not more than 60 Business Days' notice to the Holders and the Agent, in each case calculated from the effective date of the notice.

See further in Clause 11.6 (Notice of redemption, substitution or variation) of the Terms and Conditions.

Cancellation of Capital Securities

All Capital Securities which are redeemed pursuant to Clause 11 (*Redemption and repurchase of the Capital Securities*) of the Terms and Conditions and all Capital Securities purchased and elected to be cancelled pursuant to Clause 11.7 (*Cancellation of Capital Securities*) of the Terms and Conditions will be cancelled and may not be reissued or resold.

See further in Clause 11.8 (Cancellation of Capital Securities) of the Terms and Conditions.

Payments in respect of the Capital Securities

Any payment or repayment under the Terms and Conditions, or any amount due in respect of a repurchase of any Capital Securities, shall be made to the person whom is registered as a Holder on the Record Date prior to an Interest Payment Date or other relevant payment date, or to such other person who is registered with the CSD on such Record Date as being entitled to receive the relevant payment, repayment or repurchase amount.

See further in Clause 8 (Payments in respect of the Capital Securities) of the Terms and Conditions.

Interest, default interest and deferral interest

Interest for Floating Rate Capital Securities

Each Floating Rate Capital Security (NC5) carries Interest at the Interest Rate applied to the Nominal Amount from (but excluding) the Issue Date up to (and including) the relevant Redemption Date.

Each Floating Rate Capital Security (NC7) carries Interest at the Interest Rate applied to the Nominal Amount from (but excluding) the First Issue Date up to (and including) the relevant Redemption Date.

Interest accrues during an Interest Period. Subject to Clause 10 (*Optional Interest Deferral*) of the Terms and Conditions, Payment of Interest in respect of the Capital Securities shall be made to the Holders on each Interest Payment Date for the preceding Interest Period.

Interest shall be calculated on the basis of the actual number of days in the Interest Period in respect of which payment is being made divided by 360 (actual/360-days basis).

The Interest Rate is calculated as the Base Rate (STIBOR three (3) month or any reference rate replacing STIBOR three (3) month in accordance with Clause 17 (*Replacement of Base Rate*) of the Terms and Conditions of the Floating Rate Capital Securities) plus the applicable Margin, as adjusted by any application of Clause 17 (*Replacement of Base Rate*) of the Terms and Conditions of the Floating Rate Capital Securities (and if any such total rate is below zero then the Interest Rate will be deemed to be zero).

The applicable Margin for the Floating Rate Capital Securities (NC5) is:

- (a) in respect of the period from (but excluding) the First Issue Date to (and including) the First Call Date, 1.70 per cent. *per annum*; and
- (b) in respect of the period from (but excluding) the Margin Step-up Date, 2.70 per cent. per annum.

The applicable Margin for the Floating Rate Capital Securities (NC7) is:

- (a) in respect of the period from (but excluding) the Issue Date to (and including) the First Call Date, 2.20 per cent. *per annum*; and
- (b) in respect of the period from (but excluding) the Margin Step-up Date, 3.20 per cent. *per annum*.

Interest for Fixed Rate Capital Securities

Each Fixed Rate Capital Security carries Interest at the Interest Rate applied to the Nominal Amount from (but excluding) the First Issue Date up to (and including) the relevant Redemption Date.

Interest accrues during an Interest Period. Subject to Clause 10 (*Optional Interest Deferral*) of the Terms and Conditions, Payment of Interest in respect of the Capital Securities shall be made to the Holders on each Interest Payment Date for the preceding Interest Period.

Interest shall be calculated on the basis of a 360-day year comprised of twelve (12) months of thirty (30) days each and, in case of an incomplete month, the actual number of days elapsed (30/360-days basis).

The Interest Rate is:

- in respect of the period from (but excluding) the First Issue Date to (and including) the Interest Step-up Date, a fixed rate of 4.815 per cent. *per annum*; and
- (d) in respect of the period from (but excluding) the Interest Rate Step-up Date, a fixed rate of 5.815 per cent. *per annum*.

Step-up after a Change of Control Event

Notwithstanding any other provision of Clause 9 (*Interest*) of the Terms and Conditions, if the Issuer does not elect to redeem the Capital Securities in accordance with Clause 11.5 (*Voluntary redemption due to a Change of Control Event*) following the occurrence of a Change of Control Event, the then prevailing Interest Rate, and each subsequent Interest Rate otherwise determined in accordance with the provisions of Clause 9 (*Interest*) of the Terms and Conditions, on the Capital Securities shall be increased by 500 basis points *per annum* with effect from (but excluding) the Change of Control Step-up Date up to (and including) the relevant Redemption Date.

Default interest

If the Issuer fails to pay any amount payable by it pursuant to Clause 10.3 (Mandatory settlement of Deferred Interest) or Clause 11 (Redemption and repurchase of the Capital Securities) (except for Clause 11.1 (No maturity), Clause 11.2 (Purchase of Capital Securities by Group Companies) and Clause 11.8 (Cancellation of Capital Securities)) of the Terms and Conditions on its due date, default interest shall accrue on the overdue amount from (but excluding) the relevant due date up to (and including) the date of actual payment at a rate which is 200 basis points per annum higher than the Interest Rate. Accrued default interest shall not be capitalised. No default interest shall accrue where the failure to pay was solely attributable to the Agent or the CSD, in which case the Interest Rate shall apply instead.

Deferral of Interest Payments

The Issuer may, at any time and at its sole discretion, elect to defer any Interest Payment, in whole or in part, on any Interest Payment Date (except on an Interest Payment Date on which the Capital Securities are to be redeemed) by giving notice of such election to the Holders, the Issuing Agent and the Agent in accordance with Clause 23 (*Notices*) of the Terms and Conditions of the Floating Rate Capital Securities (NC5), Clause 23 (*Notices*) of the Terms and Conditions of the Floating Rate Capital Securities (NC7) or Clause 22 (*Notices*) of the Terms and Conditions of the Fixed Rate Capital Securities not less than seven Business Days prior to the relevant Interest Payment Date. Notwithstanding the foregoing, failure to give such notice shall not prejudice the right of the Issuer to defer any Interest Payment as described above.

See further in Clause 10 (Optional Interest deferral) of the Terms and Conditions.

Use of benchmarks

Interest payable for the Floating Rate Capital Securities will be calculated by reference to the benchmark STIBOR (or any reference rate replacing STIBOR in accordance with Clause 17 (*Replacement of Base Rate*) of the Terms and Conditions of the Floating Rate Capital Securities (NC5) or Clause 17 (*Replacement of Base Rate*) of the Terms and Conditions of the Floating Rate Capital Securities (NC7)). STIBOR is provided by Swedish Financial Benchmark Facility which operates as an approved administrator of STIBOR pursuant to Article 36 of Regulation (EU) 2016/1011 on indices used as benchmarks in financial instruments and financial contracts or to measure the performance of investment funds (the "Benchmark Regulation").

Admission to trading of the Capital Securities

The Issuer has the intention and shall use its best efforts (without assuming any legal or contractual obligation) to ensure (a) that the Capital Securities are admitted to trading on the corporate bond list of Nasdaq Stockholm or, if such admission to trading is not possible to obtain or maintain, admitted to trading on another Regulated Market within thirty (30) days after the Issue Date, (b) that the Capital Securities, once admitted to trading on Nasdaq Stockholm (or, if applicable, any other Regulated Market), continue being listed thereon. The aforementioned shall however not apply from and including the last day on which the admission reasonably can, pursuant to the then applicable regulations of the Regulated Market and the CSD, subsist.

It is estimated that the Issuer's costs in conjunction with the admission to trading will be no higher than SEK 150,000.

Decisions by Holders

A request by the Agent for a decision by the Holders on a matter relating to the Terms and Conditions shall (at the option of the Agent) be dealt with at a Holders' Meeting or by way of a Written Procedure.

Only a Holder, or a person who has been provided with a power of attorney or other authorisation pursuant to Clause 7 (*Right to act on behalf of a Holder*) of the Terms and Conditions from a Holder:

- (e) on the Business Day specified in the notice pursuant to Clause 15.2.2 of the Terms and Conditions, in respect of a Holders' Meeting or
- (f) on the Business Day specified in the communication pursuant to Clause 15.3.2 of the Terms and Conditions, in respect of a Written Procedure;

may exercise voting rights as a Holder at such Holders' Meeting or in such Written Procedure, provided that the relevant Capital Securities are included in the Adjusted Nominal Amount.

A matter decided at a duly convened and held Holders' Meeting or by way of Written Procedure is binding on all Holders, irrespective of them being present or represented at the Holders' Meeting or responding in the Written Procedure. The Holders that have not adopted or voted for a decision shall not be liable for any damages that this may cause the Issuer or the other Holders.

Information about decisions taken at a Holders' Meeting or by way of a Written Procedure shall promptly be published on the websites of the Issuer and the Agent, provided that a failure to do so shall not invalidate any decision made or voting result achieved. The minutes from the relevant Holders' Meeting or Written Procedure shall at the request of a Holder be sent to it by the Issuer or the Agent, as applicable.

See further in Clause 15 (Decisions by Holders) of the Terms and Conditions.

No direct action by Holders

Subject to certain exemptions set out in the Terms and Conditions, a Holder may not take any steps whatsoever against the Issuer to enforce or recover any amount due or owing to it pursuant to the Terms and Conditions, or to initiate, support or procure the winding-up, dissolution, liquidation, company reorganisation (företagsrekonstruktion) or bankruptcy (konkurs) in any jurisdiction of any Group Company in relation to any of the obligations and liabilities of the Issuer or any Subsidiary under the Terms and Conditions.

Time-bar

The right to receive repayment of the principal of the Capital Securities shall be time-barred and become void ten (10) years from the relevant Redemption Date. The right to receive payment of interest (excluding any capitalised interest) shall be time-barred and become void three (3) years from the relevant due date for payment. The Issuer is entitled to any funds set aside for payments in respect of which the Holders' right to receive payment has been time-barred and has become void.

Governing law

The Terms and Conditions of the Capital Securities and any non-contractual obligations arising out of or in connection therewith shall be governed by and construed in accordance with the laws of Sweden. The Issuer submits to the non-exclusive jurisdiction of the District Court of Stockholm (*Stockholms tingsrätt*).

The CSD

Euroclear Sweden AB, Swedish Corporate ID No. 556112-8074, P.O. Box 191, SE-101 23 Stockholm, Sweden, is initially acting as Central Securities Depositary (CSD) and registrar in respect of the Capital Securities.

The Agent and the Agency Agreement

Nordic Trustee & Agency AB (publ), Swedish Corporate ID. 556882-1879, P.O. Box 7329, SE-103 90 Stockholm, Sweden, is initially acting as Agent on behalf of the Holders in accordance with the Terms and Conditions.

Pursuant to the Agency Agreement that was entered into on or before the Issue Date between the Issuer and the Agent, the Agent has undertaken to represent the Holders in accordance with the Terms and Conditions. The Issuer has undertaken to, among other things, pay certain fees to the Agent.

The Issuing Agent

Nordea Bank Abp, filial i Sverige, Swedish Corporate ID. 516411-1683, Smålandsgatan 17, SE-105 71 Stockholm, Sweden, is initially acting as Issuing Agent in accordance with the Terms and Conditions of the Capital Securities.

Use of proceeds

The Issuer shall use the Net Proceeds from the issue of Capital Securities for its general corporate purposes, including refinancing financial indebtedness and financing acquisitions and investments.

The Issuer shall use the Net Proceeds from the issue of any Subsequent Capital Securities for its general corporate purposes, including financing acquisitions and investments.

TERMS AND CONDITIONS OF THE FLOATING RATE CAPITAL SECURITIES (NC5)



TERMS AND CONDITIONS FOR SWEDAVIA AB (PUBL)

SEK 1,750,000,000 SUBORDINATED PERPETUAL FLOATING RATE CALLABLE CAPITAL SECURITIES

ISIN: SE0026853327

Issue Date: 19 November 2025

SELLING RESTRICTIONS

No action is being taken that would or is intended to permit a public offering of the Capital Securities or the possession, circulation or distribution of this document or any other material relating to the Issuer or the Capital Securities in any jurisdiction other than Sweden, where action for that purpose is required. Persons into whose possession this document comes are required by the Issuer to inform themselves about, and to observe, any applicable restrictions.

PRIVACY NOTICE

The Issuer, the Agent and the Issuing Agent may collect and process personal data relating to the Holders, the Holders' representatives or agents, and other persons nominated to act on behalf of the Holders pursuant to the Terms and Conditions (name, contact details and, when relevant, holding of Capital Securities). The personal data relating to the Holders is primarily collected from the registry kept by the CSD. The personal data relating to other persons is primarily collected directly from such persons.

The personal data collected will be processed by the Issuer, the Agent and the Issuing Agent for the following purposes:

- (a) to exercise their respective rights and fulfil their respective obligations under the Terms and Conditions;
- (b) to manage the administration of the Capital Securities and payments under the Capital Securities;
- (c) to enable the Holders' to exercise their rights under the Terms and Conditions; and
- (d) to comply with their obligations under applicable laws and regulations.

The processing of personal data by the Issuer, the Agent and the Issuing Agent in relation to items (a) to (c) (inclusive) is based on their legitimate interest to exercise their respective rights and to fulfil their respective obligations under the Terms and Conditions. In relation to item (d), the processing is based on the fact that such processing is necessary for compliance with a legal obligation incumbent on the Issuer, the Agent or the Issuing Agent. Unless otherwise required or permitted by law, the personal data collected will not be kept longer than necessary given the purpose of the processing.

Personal data collected may be shared with third parties, such as the CSD, when necessary to fulfil the purpose for which such data is processed.

Subject to any legal preconditions, the applicability of which have to be assessed in each individual case, data subjects have the rights as follows. Data subjects have right to get access to their personal data and may request the same in writing at the address of the Issuer, the Agent and the Issuing Agent, respectively. In addition, data subjects have the right to (i) request that personal data is rectified or erased, (ii) object to specific processing, (iii) request that the processing be restricted and (iv) receive personal data provided by themselves in machine-readable format.

Data subjects are also entitled to lodge complaints with the relevant supervisory authority if dissatisfied with the processing carried out.

The Issuer's, the Agent's and the Issuing Agent's addresses, and the contact details for their respective Data Protection Officers (if applicable), are found on their websites www.swedavia.se, www.nordictrustee.com and www.nordea.com.

TABLE OF CONTENTS

1.	DEFINITIONS AND CONSTRUCTION	1
2.	THE CAPITAL SECURITIES	8
3.	STATUS OF THE CAPITAL SECURITIES	9
4.	USE OF PROCEEDS	10
5.	CONDITIONS PRECEDENT	10
6.	CAPITAL SECURITIES IN BOOK-ENTRY FORM	11
7.	RIGHT TO ACT ON BEHALF OF A HOLDER	11
8.	PAYMENTS IN RESPECT OF THE CAPITAL SECURITIES	12
9.	INTEREST	12
10.	OPTIONAL INTEREST DEFERRAL	13
11.	REDEMPTION AND REPURCHASE OF THE CAPITAL SECURITIES	14
12.	PRECONDITIONS TO SPECIAL EVENT REDEMPTION OR CHANGE OF CEVENT REDEMPTION	
13.	ADMISSION TO TRADING	16
14.	DEFAULT AND ENFORCEMENT	17
15.	DECISIONS BY HOLDERS	17
16.	AMENDMENTS AND WAIVERS	22
17.	REPLACEMENT OF BASE RATE	23
18.	THE AGENT	27
19.	THE ISSUING AGENT	31
20.	THE CSD	31
21.	NO DIRECT ACTIONS BY HOLDERS	31
22.	TIME-BAR	32
23.	NOTICES	32
24.	FORCE MAJEURE	33
25.	GOVERNING LAW AND JURISDICTION	33
SCH	HEDULE	
1.	CONDITIONS PRECEDENT	34

1. DEFINITIONS AND CONSTRUCTION

1.1 **Definitions**

In these terms and conditions (the "Terms and Conditions"):

- "Account Operator" means a bank or other party duly authorised to operate as an account operator pursuant to the Central Securities Depositories and Financial Instruments Accounts Act and through which a Holder has opened a Securities Account in respect of its Capital Securities.
- "Accounting Event" means the receipt by the Issuer of an opinion of an independent and authorised accountant (auktoriserad revisor) in Sweden (experienced in such matters) to the effect that, as a result of a change in the Accounting Principles or interpretation thereof, the equity treatment of the Capital Securities as "equity" in full in the Issuer's consolidated financial statements has or will cease.
- "Accounting Principles" means the international financial reporting standards (IFRS) within the meaning of Regulation 1606/2002/EC (or as otherwise adopted or amended from time to time).
- "Adjusted Nominal Amount" means the Total Nominal Amount less the aggregate Nominal Amount of all Capital Securities owned by a Group Company, irrespective of whether such person is directly registered as owner of such Capital Securities.
- "Agency Agreement" means the agency agreement entered into before the Issue Date, between the Issuer and the Agent, or any replacement agency agreement entered into after the Issue Date between the Issuer and an agent.
- "Agent" means Nordic Trustee & Agency AB (publ), Swedish Reg. No. 556882-1879, or another party replacing it, as Holders' agent, in accordance with these Terms and Conditions.
- "Base Rate" means STIBOR or any reference rate replacing STIBOR in accordance with Clause 17 (Replacement of Base Rate).
- "Base Rate Administrator" means the Swedish Financial Benchmark Facility AB (SFBF) or any person replacing it as administrator of the Base Rate.
- "Business Day" means a day in Sweden other than a Sunday or other public holiday. Saturdays, Midsummer Eve (midsommarafton), Christmas Eve (julafton) and New Year's Eve (nyårsafton) shall for the purpose of this definition be deemed to be public holidays.
- "Business Day Convention" means the first following day that is a Business Day unless that day falls in the next calendar month, in which case that date will be the first preceding day that is a Business Day.
- "Capital Security" means a debt instrument (skuldförbindelse) for the Nominal Amount and of the type set forth in Chapter 1 Section 3 of the Central Securities Depositories and Financial Instruments Accounts Act and which are governed by and issued by the Issuer under these Terms and Conditions.
- "Central Securities Depositories and Financial Instruments Accounts Act" means the Swedish Central Securities Depositories and Financial Instruments Accounts Act (lag (1998:1479) om värdepapperscentraler och kontoföring av finansiella instrument).
- "Change of Control Event" means an event where the Government of Sweden ceases to, directly or indirectly, own or control less than 100 per cent. of the shares and the votes in the Issuer.
- "Change of Control Step-up Date" means the date falling six (6) months after the date on which a Change of Control Event has occurred.
- "CSD" means the Issuer's central securities depository and registrar in respect of the Capital Securities, Euroclear Sweden AB, Swedish Reg. No. 556112-8074, P.O. Box 191, 101 23 Stockholm, Sweden, or another party replacing it, as CSD, in accordance with these Terms and Conditions.
- "CSD Regulations" means the CSD's rules and regulations applicable to the Issuer, the Agent and the Capital Securities from time to time.

"**Debt Register**" means the debt register (*skuldbok*) kept by the CSD in respect of the Capital Securities in which a Holder is registered.

"Deferred Interest" has the meaning ascribed to it in Clause 10.1.3.

"Deferred Interest Payment Event" means any one or more of the following events:

- (a) declaration or payment of any distribution or dividend or any other payment made by the Issuer on its share capital or any other obligation of the Issuer which ranks or is expressed by its terms to rank junior to the Capital Securities or any Parity Securities;
- (b) declaration or payment of any distribution or dividend or any other payment made by the Issuer or any Subsidiary of the Issuer, as the case may be, on any Parity Securities;
- (c) redemption, repurchase, repayment, cancellation, reduction or other acquisition by the Issuer or any Subsidiary of the Issuer of any shares of the Issuer or any other obligation of the Issuer which ranks or is expressed by its terms to rank junior to the Capital Securities or any Parity Securities; and/or
- (d) redemption, repurchase, repayment, cancellation, reduction or other acquisition by the Issuer or any Subsidiary of the Issuer of any Parity Securities,

save for:

- (i) in each case, any compulsory distribution, dividend, other payment, redemption, repurchase, repayment, cancellation, reduction or other acquisition required by the terms of such securities or by mandatory operation of applicable law; and
- (ii) in the case of paragraph (c) above only, any redemption, repurchase, repayment, cancellation, reduction or other acquisition that is executed in connection with, or for the purpose of:
 - (A) any reduction of the quota value of the share capital of the Issuer without a corresponding return of cash, capital or assets to shareholders of the Issuer; or
 - (B) any share buyback programme then in force and duly approved by the shareholders' general meeting of the Issuer or the relevant Subsidiary of the Issuer (as applicable) or any existing or future stock option plan or free share allocation plan or other incentive plan, in all cases, reserved for directors, officers and/or employees of the Issuer or the relevant Subsidiary of the Issuer or any associated hedging transaction.

"First Call Date" means the date falling five (5) years after the Issue Date or, to the extent such day is not a Business Day, the Business Day following from an application of the Business Day Convention.

"Force Majeure Event" has the meaning set forth in Clause 24.1.

"Group" means the Issuer and its Subsidiaries from time to time (each a "Group Company").

"Holder" means the person who is registered on a Securities Account as direct registered owner (direktregistrerad ägare) or nominee (förvaltare) with respect to a Capital Security.

"Holders' Meeting" means a meeting among the Holders held in accordance with Clauses 15.1 (Request for a decision), 15.2 (Convening of Holders' Meeting) and 15.4 (Majority, quorum and other provisions).

"Insolvent" means, in respect of a relevant person, that it is deemed to be insolvent, or admits inability to pay its debts as they fall due, in each case within the meaning of Chapter 2, Sections 7-9 of the Swedish Bankruptcy Act (konkurslagen (1987:672)) (or its equivalent in any other relevant jurisdiction).

"Interest" means the interest on the Capital Securities calculated in accordance with Clauses 9.1.1 to 9.1.3.

- "Interest Payment" means, in respect the payment of interest on an Interest Payment Date, the amount of interest payable for the relevant Interest Period in accordance with Clause 9 (Interest).
- "Interest Payment Date" means, subject to Clause 10 (*Optional Interest Deferral*), 19 February, 19 May, 19 August and 19 November of each year or, to the extent such day is not a Business Day, the Business Day following from an application of the Business Day Convention. The first Interest Payment Date for the Capital Securities shall be 19 February 2026 and the last Interest Payment Date shall be the relevant Redemption Date.
- "Interest Period" means (i) in respect of the first Interest Period, the period from (but excluding) the Issue Date to (and including) the first Interest Payment Date or, if the Capital Securities are redeemed before, the Redemption Date, and (ii) in respect of subsequent Interest Periods, the period from (but excluding) an Interest Payment Date to (and including) the next succeeding Interest Payment Date or, if the Capital Securities are redeemed before, the Redemption Date.
- "**Interest Rate**" means the Base Rate plus the applicable Margin, as adjusted by any application of Clause 17 (*Replacement of Base Rate*). For the avoidance of doubt, if any such total rate is below zero then the Interest Rate will be deemed to be zero (0).
- "Issue Date" means 19 November 2025.
- "Issuer" means Swedavia AB (publ), a public limited liability company incorporated under the laws of Sweden with Reg. No. 556797-0818.
- "Issuer Winding-up" has the meaning set forth in Clause 3.2.
- "Issuer Re-Construction" has the meaning set forth in Clause 3.2.
- "Issuing Agent" means, initially, Nordea Bank Abp, filial i Sverige, Reg. No. 516411-1683, and thereafter each other party appointed as Issuing Agent in accordance with these Terms and Conditions and the CSD Regulations.
- "Margin" means:
- (a) in respect of the period from (but excluding) the Issue Date to (and including) the Margin Step-up Date, 1.70 per cent. per annum; and
- (b) in respect of the period from (but excluding) the Margin Step-up Date, 2.70 per cent. per annum.
- "Margin Step-up Date" means the Interest Payment Date falling in November 2035.
- "Net Proceeds" means the gross proceeds from the offering of the relevant Capital Securities, minus the costs incurred by the Issuer in conjunction with the issuance and listing thereof.
- "Nominal Amount" has the meaning set forth in Clause 2.3.
- "Parity Securities" means any obligations of:
- (a) the Issuer, issued directly or indirectly by it, which rank, or are expressed to rank, pari passu with the Capital Securities; and
- (b) any Subsidiary of the Issuer having the benefit of a guarantee or support agreement from the Issuer which ranks or is expressed to rank pari passu with the Capital Securities.
- "Qualifying Securities" means securities issued directly by the Issuer following a substitution or variation of the Capital Securities in accordance with Clause 11.6 (Substitution or variation due to a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event) that have terms not materially less favourable to investors, certified by the Issuer acting reasonably (having consulted with an independent investment bank or independent financial adviser), than the terms of the Capital Securities (immediately prior to the relevant substitution or variation), provided that they:
- (a) shall include a ranking at least equal to that of the Capital Securities;
- (b) shall have at least the same Interest Rate and the same Interest Payment Dates as those applying to the Capital Securities;
- (c) shall have the same redemption rights as the Capital Securities;

- (d) shall preserve any existing rights under the Capital Securities to any accrued interest which has not been paid but which has not been cancelled in respect of the period from (and including) the Interest Payment Date last preceding the date of the relevant substitution or variation of the Capital Securities;
- (e) if the Capital Securities were admitted to trading on a Regulated Market immediately prior to the relevant substitution or variation, the Issuer shall use reasonable efforts to ensure that the Capital Securities are admitted to trading on a Regulated Market within thirty (30) days from their issuance (noting that no investor in the relevant Qualifying Securities (or its representative) has the right to accelerate the relevant Qualifying Securities or otherwise request a prepayment or redemption of the relevant Qualifying Securities upon a failure to admit the relevant Qualifying Securities to trading).
- "Quotation Day" means, in relation to any period for which an interest rate is to be determined, two (2) Business Days before the first day of that period.
- "Rating Agency" means Moody's and any other rating agency of equivalent international standing requested by the Issuer to grant a corporate credit rating to the Issuer and, in each case, their successors or affiliates.
- "Rating Methodology Event" shall be deemed to occur if the Issuer has received confirmation from any Rating Agency either directly or via a publication by such Rating Agency, that due to any amendment to, clarification of or change in hybrid capital methodology or a change in the interpretation thereof, in each case occurring or becoming effective after the Issue Date:
- (a) all or any of the Capital Securities will no longer be eligible for the same or a higher amount of "equity credit" (or such other nomenclature that the relevant Rating Agency may then use to describe the degree to which an instrument exhibits the characteristics of an ordinary share) attributed to the Capital Securities at the Issue Date (or, if "equity credit" is not assigned to the Capital Securities by the relevant Rating Agency on the Issue Date, at the date on which "equity credit" is assigned by such Rating Agency for the first time); or
- (b) the length of time the Capital Securities are assigned a particular level of "equity credit" by the relevant Rating Agency is shortened as compared to the length of time that the Capital Securities were assigned that level of "equity credit" by the relevant Rating Agency under its prevailing methodology on the Issue Date.

"Record Date" means the fifth (5) Business Day prior to:

- (a) an Interest Payment Date;
- (b) a Redemption Date; or
- (c) another relevant date,

or in each case such other Business Day falling prior to a relevant date if generally applicable on the Swedish bond market.

- "Redemption Date" means the date on which the relevant Capital Securities are to be redeemed or repurchased in accordance with Clause 11 (Redemption and repurchase of the Capital Securities).
- "Regulated Market" means any regulated market (as defined in Directive 2014/65/EU on markets in financial instruments).
- "Securities Account" means the account for dematerialised securities (avstämningsregister) maintained by the CSD pursuant to the Central Securities Depositories and Financial Instruments Accounts Act in which:
- (a) an owner of such security is directly registered; or
- (b) an owner's holding of securities is registered in the name of a nominee.
- "Security" means a mortgage, charge, pledge, lien, security assignment or other security interest securing any obligation of any person, or any other agreement or arrangement having a similar effect.

"Special Event" means any of an Accounting Event, a Substantial Repurchase Event, a Tax Deductibility Event, a Rating Methodology Event. a Withholding Tax Event or any combination of the foregoing.

"STIBOR" means:

- (c) the Stockholm interbank offered rate (STIBOR) administered by the Swedish Financial Benchmark Facility AB (or any person replacing it as administrator) for Swedish Kronor and for a period equal to the relevant Interest Period, as published by Swedish Financial Benchmark Facility AB (or any other person which takes over the publication of that rate) as of or around 11.00 a.m. on the Quotation Day;
- (d) if no rate as described in paragraph (a) above is available for the relevant Interest Period, the rate determined by the Issuing Agent by linear interpolation between the two closest rates for STIBOR fixing, as published by Swedish Financial Benchmark Facility AB (or any other person which takes over the publication of that rate), as of or around 11.00 a.m. on the Quotation Day for Swedish Kronor;
- (e) if no rate as described in paragraph (a) or (b) above is available for the relevant Interest Period, the arithmetic mean of the Stockholm interbank offered rates (rounded upwards to four decimal places) as supplied to the Issuing Agent at its request quoted by leading banks in the Stockholm interbank market reasonably selected by the Issuing Agent for deposits of SEK 100,000,000 for the relevant period; or
- (f) if no rate as described in paragraph (a) or (b) above is available for the relevant Interest Period and no quotation is available pursuant to paragraph (c) above, the interest rate which according to the reasonable assessment of the Issuing Agent best reflects the interest rate for deposits in Swedish Kronor offered in the Stockholm interbank market for the relevant period.
- "Subordinated Indebtedness" means any obligation of the Issuer, whether or not having a fixed maturity, which by its terms is, or is expressed to be, subordinated in the event of an Issuer Windingup to the claims of all unsubordinated creditors of the Issuer but senior to the Capital Securities or to the obligations of the Issuer in respect of any Parity Securities.
- "Subsidiary" means, in relation to any person, any Swedish or foreign legal entity (whether incorporated or not), which at the time is a subsidiary (*dotterföretag*) to such person, directly or indirectly, as defined in the Swedish Companies Act (*aktiebolagslagen* (2005:551)).
- "Substantial Repurchase Event" shall be deemed to occur if the Issuer and/or any of its Subsidiaries repurchases and cancels or has at any time repurchased and cancelled, a principal amount of Capital Securities equal to or greater than eighty (80) per cent. of the aggregate principal amount of the Capital Securities issued.
- "Swedish Kronor" and "SEK" means the lawful currency of Sweden.
- "Tax Deductibility Event" means the receipt by the Issuer of an opinion of independent and well-reputed counsel in Sweden (experienced in such matters) to the effect that, as a result of a Tax Law Change, any interest payments under the Capital Securities were, but are no longer, tax-deductible by the Issuer for Swedish tax purposes to the same extent as any unsubordinated obligations of the Issuer.

"Tax Law Change" means:

- (a) any amendment to, clarification of, or change in, the laws or treaties (or any regulations thereunder) of Sweden, or any political subdivision or any authority thereof or therein having the power to tax, affecting taxation;
- (b) any governmental action; or
- (c) any amendment to, clarification of, or change in the application, official position or the interpretation of such law, treaty (or regulations thereunder) or governmental action or any interpretation, decision or pronouncement that provides for a position with respect to such law, treaty (or regulations thereunder) or governmental action that differs from the theretofore generally accepted position,

in each case, by any legislative body, court, governmental authority or regulatory body in Sweden, irrespective of the manner in which such amendment, clarification, change, action, pronouncement, interpretation or decision is made known, which amendment, clarification or change is effective or such governmental action, pronouncement, interpretation or decision is announced on or after the date falling five (5) Business Days prior to the Issue Date.

"**Total Nominal Amount**" means the total aggregate Nominal Amount of the Capital Securities outstanding at the relevant time.

"Withholding Tax Event" shall be deemed to occur if, as a result of any Tax Law Change, in making any payments on the Capital Securities, the Issuer has paid or will or would on the next Interest Payment Date be required to pay additional amounts on the Capital Securities and the Issuer cannot avoid the foregoing by taking reasonable measures available to it as evidenced by an opinion of an independent and well-reputed counsel in Sweden (experienced in such matters) to that effect.

"Written Procedure" means the written or electronic procedure for decision making among the Holders in accordance with Clauses 15.1 (Request for a decision), 15.3 (Instigation of Written Procedure) and 15.4 (Majority, quorum and other provisions).

1.2 Construction

- 1.2.1 Unless a contrary indication appears, any reference in these Terms and Conditions to:
 - (a) "assets" includes present and future properties, revenues and rights of every description;
 - (b) any agreement or instrument is a reference to that agreement or instrument as supplemented, amended, novated, extended, restated or replaced from time to time;
 - (c) a "**regulation**" includes any law, regulation, rule or official directive (whether or not having the force of law) of any governmental, intergovernmental or supranational body, agency or department;
 - (d) a provision of regulation is a reference to that provision as amended or re-enacted; and
 - (e) a time of day is a reference to Stockholm time.
- 1.2.2 When ascertaining whether a limit or threshold specified in Swedish Kronor has been attained or broken, an amount in another currency shall be counted on the basis of the rate of exchange for such currency against Swedish Kronor for the previous Business Day, as published by the Swedish Central Bank (*Riksbanken*) on its website (www.riksbank.se). If no such rate is available, the most recently published rate shall be used instead.
- 1.2.3 A notice shall be deemed to be sent by way of press release if it is made available to the public within Sweden promptly and in a non-discriminatory manner.
- 1.2.4 No delay or omission of the Agent or of any Holder to exercise any right or remedy under the Terms and Conditions shall impair or operate as a waiver of any such right or remedy.
- 1.2.5 The selling restrictions, the privacy notice and any other information contained in this document before the table of contents section do not form part of these Terms and Conditions and may be updated without the consent of the Holders and the Agent.

2. THE CAPITAL SECURITIES

2.1 The Capital Securities are denominated in Swedish Kronor and each Capital Security is constituted by these Terms and Conditions. The Issuer undertakes to make payments in relation to the Capital Securities (subject to and in accordance with these Terms and Conditions) and to comply with these Terms and Conditions.

- 2.2 By subscribing for Capital Securities, each initial Holder agrees that the Capital Securities shall benefit from and be subject to the Terms and Conditions and by acquiring Capital Securities, each subsequent Holder confirms such agreement.
- 2.3 The nominal amount of each Capital Security is SEK 1,250,000 (the "Nominal Amount"). The Total Nominal Amount of the Capital Securities as at the Issue Date is SEK 1,750,000,000. All Capital Securities are issued on a fully paid basis at an issue price of one hundred (100) per cent. of the Nominal Amount.
- 2.4 The ISIN for the Capital Securities is SE0026853327.
- 2.5 The Capital Securities are freely transferable but the Holders may be subject to purchase or transfer restrictions with regard to the Capital Securities, as applicable, under local regulation to which a Holder may be subject. Each Holder must ensure compliance with such restrictions at its own cost and expense.

3. STATUS OF THE CAPITAL SECURITIES

3.1 The Capital Securities, including the obligation to pay interest thereon, constitute direct, general, unconditional, unsecured and subordinated obligations of the Issuer. The rights and claims of the Holders in respect of the Capital Securities against the Issuer, are subordinated as described in Clause 3.2 below.

3.2 In the event of:

- (a) a voluntary or involuntary liquidation (*likvidation*) or bankruptcy (*konkurs*) of the Issuer (each an "**Issuer Winding-up**"), the Holders (or the Agent on their behalf) shall, in respect of their Capital Securities, have a claim for the principal amount of their Capital Securities and any accrued and unpaid interest (including any Deferred Interest) thereon and such claims will rank:
 - (i) pari passu without any preference among themselves and with any present or future claims in respect of obligations of the Issuer in respect of Parity Securities;
 - (ii) in priority to all present and future claims in respect of:
 - (A) the ordinary shares of the Issuer; and
 - (B) any other obligation of the Issuer expressed by its terms as at its original issue date to rank junior to the Capital Securities or any Parity Securities; and
 - (iii) junior in right of payment to any present or future claims of:
 - (A) all unsubordinated obligations of the Issuer; and
 - (B) all Subordinated Indebtedness; or
- (b) a company re-construction (företagsrekonstruktion) of the Issuer under the Swedish Company Reorganisation Act (*lag (2022:964) om företagsrekonstruktion*) (an "**Issuer Reconstruction**"), the Holders (or the Agent on their behalf) shall, in respect of their Capital Securities, have a claim for the principal amount of their Capital Securities and any accrued and unpaid interest (including any Deferred Interest) thereon and such claims will rank:
 - (i) pari passu without any preference among themselves and with any present or future claims in respect of obligations of the Issuer in respect of Parity Securities; and
 - (ii) junior in right of payment to any present or future claims of:
 - (A) all unsubordinated obligations of the Issuer; and
 - (B) all Subordinated Indebtedness.

Claims in respect of the share capital of the Issuer are not subject to loss absorbing measures under an Issuer Re-construction.

3.3 Subject to applicable law, no Holder may exercise, claim or plead any right of set-off, compensation or retention in respect of any amount owed to it by the Issuer in respect of, or arising under or in connection with the Capital Securities and each Holder shall, by virtue of its holding of any Capital Security, be deemed to have waived all such rights of set-off, compensation or retention.

4. USE OF PROCEEDS

The Issuer shall use the Net Proceeds from the issue of Capital Securities for its general corporate purposes, including refinancing financial indebtedness and financing acquisitions and investments.

5. CONDITIONS PRECEDENT

5.1 Conditions precedent to the Issue Date

- 5.1.1 The Issuer shall provide to the Agent, prior to the Issue Date (or such later time as agreed by the Agent), each document and other evidence listed in <u>Schedule 1</u> (*Conditions Precedent*) in form and substance satisfactory to the Agent.
- 5.1.2 The Agent shall confirm to the Issuing Agent when it is satisfied that the conditions in Clause 5.1.1 have been fulfilled (or amended or waived in accordance with Clause 16 (*Amendments and waivers*)).

5.2 Settlement and disbursement

Following receipt by the Issuing Agent of the confirmation in accordance with Clause 5.1.2, the Issuing Agent shall settle the issuance of the Capital Securities and pay the Net Proceeds to the Issuer on the Issue Date.

5.3 No responsibility for documentation

The Agent may assume that the documentation and evidence delivered to it pursuant to this Clause 5 is accurate, legally valid, enforceable, correct, true and complete unless it has actual knowledge to the contrary and the Agent does not have to verify or assess the contents of any such documentation or evidence. The Agent does not have any obligation to review the documentation and evidence set out in this Clause 5 from a legal or commercial perspective of the Holders.

6. CAPITAL SECURITIES IN BOOK-ENTRY FORM

- The Capital Securities will be registered for the Holders on their respective Securities Accounts and no physical Capital Securities will be issued. Accordingly, the Capital Securities will be registered in accordance with the Central Securities Depositories and Financial Instruments Accounts Act. Registration requests relating to the Capital Securities shall be directed to an Account Operator. The Debt Register shall constitute conclusive evidence of the persons who are Holders and their holdings of Capital Securities.
- Those who according to assignment, Security, the provisions of the Swedish Children and Parents Code (*föräldrabalken* (1949:381)), conditions of will or deed of gift or otherwise have acquired a right to receive payments in respect of a Capital Security shall register their entitlements to receive payment in accordance with the Central Securities Depositories and Financial Instruments Accounts Act.
- 6.3 The Issuer and the Agent shall at all times be entitled to obtain information from the Debt Register. At the request of the Agent, the Issuer shall promptly obtain such information and provide it to the Agent. For the purpose of carrying out any administrative procedure that arises out of the Terms and Conditions, the Issuing Agent shall be entitled to obtain information from the Debt Register.
- 6.4 The Issuer shall issue any necessary power of attorney to such persons employed by the Agent, as notified by the Agent, in order for such individuals to independently obtain information directly from

the Debt Register. The Issuer may not revoke any such power of attorney unless directed by the Agent or unless consent thereto is given by the Holders.

6.5 The Issuer and the Agent may use the information referred to in Clause 6.3 only for the purposes of carrying out their duties and exercising their rights in accordance with the Terms and Conditions and the Agency Agreement and shall not disclose such information to any Holder or third party unless necessary for such purposes.

7. RIGHT TO ACT ON BEHALF OF A HOLDER

- 7.1 If any person other than a Holder (including the owner of a Capital Security, if such person is not the Holder), wishes to exercise any rights under the Terms and Conditions, it must obtain a power of attorney or other authorisation from the Holder or a successive, coherent chain of powers of attorney or authorisations starting with the Holder and authorising such person.
- 7.2 A Holder may issue one or several powers of attorney or other authorisations to third parties to represent it in relation to some or all of the Capital Securities held by it. Any such representative may act independently under the Terms and Conditions in relation to the Capital Securities for which such representative is entitled to represent the Holder.
- 7.3 The Agent shall only have to examine the face of a power of attorney or other authorisation that has been provided to it pursuant to Clause 7.2 and may assume that such document has been duly authorised, is valid, has not been revoked or superseded and that it is in full force and effect, unless otherwise is apparent from its face or the Agent has actual knowledge to the contrary.
- 7.4 These Terms and Conditions shall not affect the relationship between a Holder who is the nominee (*förvaltare*) with respect to a Capital Security and the owner of such Capital Security, and it is the responsibility of such nominee to observe and comply with any restrictions that may apply to it in this capacity.

8. PAYMENTS IN RESPECT OF THE CAPITAL SECURITIES

- 8.1 Any payment or repayment under the Terms and Conditions, or any amount due in respect of a repurchase of any Capital Securities, shall be made to such person who is registered as a Holder on the Record Date prior to an Interest Payment Date or other relevant payment date, or to such other person who is registered with the CSD on such Record Date as being entitled to receive the relevant payment, repayment or repurchase amount.
- 8.2 Provided that a Holder has registered an income account (avkastningskonto) for the relevant Securities Account on the applicable Record Date, the CSD shall procure that principal, interest and other payments under the Capital Securities are deposited to such income account on the relevant payment date. If an income account has not been registered on the Record Date for the payment, no payment will be effected by the CSD to such Holder. The outstanding amount will instead be held by the Issuer until the person that was registered as a Holder on the relevant Record Date has made a valid request for such amount. Should the CSD, due to a delay on behalf of the Issuer or some other obstacle, not be able to effect payments as aforesaid, the Issuer shall procure that such amounts are paid to the persons who are registered as Holders on the relevant Record Date as soon as possible after such obstacle has been removed.
- 8.3 If, due to any obstacle for the CSD, the Issuer cannot make a payment or repayment, such payment or repayment may be postponed until the obstacle has been removed.
- 8.4 If payment or repayment is made in accordance with this Clause 8, the Issuer shall be deemed to have fulfilled its obligation to pay, irrespective of whether such payment was made to a person not entitled to receive such amount (unless the Issuer has actual knowledge of the fact that the payment was made to the wrong person).
- 8.5 The Issuer is not liable to gross-up any payments under the Terms and Conditions by virtue of any withholding tax, public levy or the similar.

9. INTEREST

9.1 **Interest**

- 9.1.1 Each Capital Security carries Interest at the applicable Interest Rate applied to the Nominal Amount from (but excluding) the Issue Date up to (and including) the relevant Redemption Date.
- 9.1.2 Interest accrues during an Interest Period. Subject to Clause 10 (*Optional Interest Deferral*), payment of Interest in respect of the Capital Securities shall be made to the Holders on each Interest Payment Date for the preceding Interest Period.
- 9.1.3 Interest shall be calculated on the basis of the actual number of days in the Interest Period in respect of which payment is being made divided by 360 (actual/360-days basis).
- 9.1.4 Payment of Interest in respect of each Interest Period may be deferred in accordance with Clause 10 (Optional Interest Deferral).

9.2 Step-up after a Change of Control Event

Notwithstanding any other provision of this Clause 9, if the Issuer does not elect to redeem the Capital Securities in accordance with Clause 11.5 (*Voluntary redemption due to* Change of Control Event) following the occurrence of a Change of Control Event, the then prevailing Interest Rate, and each subsequent Interest Rate otherwise determined in accordance with the provisions of this Clause 9, on the Capital Securities shall be increased by 500 basis points *per annum* with effect from (but excluding) the Change of Control Step-up Date up to (and including) the relevant Redemption Date.

9.3 **Default interest**

If the Issuer fails to pay any amount payable by it pursuant to Clause 10.3 (Mandatory settlement of Deferred Interest) or Clause 11 (Redemption and repurchase of the Capital Securities) (except for Clause 11.1 (No maturity), Clause 11.2 (Purchase of Capital Securities by Group Companies) and Clause 11.8 (Cancellation of Capital Securities)) on its due date, default interest shall accrue on the overdue amount from (but excluding) the due date up to (and including) the date of actual payment at a rate which is 200 basis points per annum higher than the Interest Rate. Accrued default interest shall not be capitalised. No default interest shall accrue where the failure to pay was solely attributable to the Agent or the CSD, in which case the Interest Rate shall apply instead.

10. OPTIONAL INTEREST DEFERRAL

10.1 **Deferral of Interest Payments**

- 10.1.1 The Issuer may, at any time and at its sole discretion, elect to defer any Interest Payment, in whole or in part, which is otherwise scheduled to be paid on an Interest Payment Date (except on any Interest Payment Date on which the Capital Securities are to be redeemed) by giving notice of such election to the Holders, the Issuing Agent and the Agent in accordance with Clause 23 (*Notices*) not less than seven (7) Business Days prior to the relevant Interest Payment Date. Notwithstanding the foregoing, failure to give such notice shall not prejudice the right of the Issuer to defer any Interest Payment as described above.
- 10.1.2 If the Issuer makes only a partial payment of interest on an Interest Payment Date, such amount shall be applied equally to each Capital Security.
- Any Interest Payment so deferred pursuant to this Clause 10 shall, from (but excluding) the Interest Payment Date on which such Interest Payment would (but for its deferral) have been payable to (and including) the date on which it is paid in full, itself bear interest at the Interest Rate prevailing from time to time (which interest shall compound on each Interest Payment Date) and, for so long as the same remains unpaid, such deferred interest (together with the interest accrued thereon) shall constitute "Deferred Interest".

10.1.4 The deferral of an Interest Payment in accordance with this Clause 10 shall not constitute a default pursuant to Clause 14 (*Default and enforcement*) by the Issuer under the Capital Securities or for any other purpose.

10.2 **Optional Settlement of Deferred Interest**

Deferred Interest may be paid, in whole or in part, at any time at the option of the Issuer following delivery of a notice to such effect given by the Issuer to the Holders, the Issuing Agent and the Agent in accordance with Clause 23 (*Notices*) not less than seven (7) Business Days prior to the date (to be specified in such notice) on which the Issuer will pay such Deferred Interest. Any such notice shall state the date fixed for the payment and the relevant Record Date and be irrevocable.

10.3 **Mandatory settlement of Deferred Interest**

- 10.3.1 The Issuer shall pay any Deferred Interest, in whole but not in part, on the first to occur of the following dates:
 - (a) the tenth (10th) Business Day following the date on which a Deferred Interest Payment Event occurs;
 - (b) any Interest Payment Date in respect of which the Issuer does not elect to defer all of the interest accrued in respect of the relevant Interest Period; and
 - (c) the date on which the Capital Securities are redeemed or repaid in accordance with Clause 11 (Redemption and repurchase of the Capital Securities) or Clause 14 (Default and Enforcement).
- 10.3.2 Notice of any Deferred Interest Payment Event shall be given by the Issuer to the Holders, the Issuing Agent and the Agent in accordance with Clause 23 (*Notices*) within three (3) Business Days of such event.

11. REDEMPTION AND REPURCHASE OF THE CAPITAL SECURITIES

11.1 No maturity

The Capital Securities are perpetual and have no specified maturity date. The Issuer may only redeem the Capital Securities in the circumstances described in this Clause 11 (*Redemption and repurchase of the Capital Securities*). The Capital Securities are not redeemable at the option of the Holders at any time.

11.2 Purchase of Capital Securities by Group Companies

- 11.2.1 The Issuer or any other Group Company may, subject to applicable regulations, at any time and at any price purchase Capital Securities on the market or in any other way.
- 11.2.2 Capital Securities held by a Group Company may at such Group Company's discretion be retained or sold but, if held by the Issuer, not cancelled except in connection with a Substantial Repurchase Event.

11.3 Voluntary total redemption (call option)

The Issuer may redeem all, but not some only, of the outstanding Capital Securities in full on the First Call Date or on any Interest Payment Date thereafter at an amount per Capital Security equal to one

hundred (100) per cent. of the Nominal Amount together with any Deferred Interest and any other accrued and unpaid interest up to (and including) the Redemption Date.

11.4 Voluntary redemption due a Special Event

Upon the occurrence of a Special Event which is continuing, the Issuer may, subject to Clause 12 (*Preconditions to Special Event Redemption or Change of Control Event Redemption*), redeem all, but not some only, of the Capital Securities at an amount equal to:

- (a) one hundred and one (101) per cent. of their principal amount, where such redemption occurs before the First Call Date; or
- (b) one hundred (100) per cent. of their principal amount, where such redemption occurs on or after the First Call Date.

together, in each case, with any Deferred Interest and any other accrued and unpaid interest up to (and including) the Redemption Date.

11.5 Voluntary redemption due to Change of Control Event

- 11.5.1 Upon the occurrence of a Change of Control Event, the Issuer may, no later than the Change of Control Step-up Date, subject to Clause 12 (*Preconditions to Special Event Redemption or Change of Control Event Redemption*), redeem all, but not some only, of the Capital Securities at an amount equal to:
 - one hundred and one (101) per cent. of their principal amount, where such redemption occurs before the First Call Date; or
 - (b) one hundred (100) per cent. of their principal amount, where such redemption occurs on or after the First Call Date,

together, in each case, with any Deferred Interest and any other accrued and unpaid interest up to (and including) the Redemption Date.

- 11.5.2 Immediately upon the Issuer becoming aware that a Change of Control Event has occurred, the Issuer shall give notice to the Agent, the Issuing Agent and the Holders in accordance with Clause 23 (*Notices*) specifying the nature of the Change of Control Event.
- Substitution or variation due to a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event

If a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event occurs, the Issuer may, at its option but subject to giving notice in accordance with Clause 11.7 (*Notice of redemption, substitution or variation*), substitute or vary the terms of all (but not some only) of the outstanding Capital Securities without any requirement for the consent or approval of the Holders, so that they become or remain, as applicable, Qualifying Securities, provided that such substitution or variation does not itself give rise to any right of the Issuer to redeem, substitute or vary the terms of the Capital Securities in accordance with this Clause 11.6 in relation to the Qualifying Securities so substituted or varied.

11.7 Notice of redemption, substitution or variation

Redemption in accordance with Clauses 11.3, 11.4, or 11.5.1 shall be made by the Issuer giving not less than thirty (30) Business Days' notice and not more than sixty (60) Business Days' notice to the Holders and the Agent, in each case calculated from the effective date of the notice. The Notice from the Issuer shall specify the Redemption Date and also the Record Date on which a person shall be registered as a Holder to receive the amounts due on such Redemption Date. The notice is irrevocable but may in the case of a redemption in accordance with Clause 11.3, at the Issuer's discretion, contain one or more conditions precedent that shall be satisfied prior to the Record Date. Upon fulfilment of the conditions precedent (if any), the Issuer shall redeem the Capital Securities in full at the applicable amount on the specified Redemption Date.

11.8 Cancellation of Capital Securities

All Capital Securities which are redeemed pursuant to this Clause 11 and all Capital Securities purchased and elected to be cancelled pursuant to Clause 11.2 (*Purchase of Capital Securities by Group Companies*) will be cancelled and may not be reissued or resold. The Issuer shall promptly inform the Holders, the Agent and the Issuing Agent in accordance with Clause 23 (*Notices*) of any such cancellation and for so long as the Capital Securities are admitted to trading on Nasdaq Stockholm or another Regulated Market, as the case may be, and the rules of such exchange so require, the Issuer shall promptly inform Nasdaq Stockholm or the Regulated Market, as the case may be, of the cancellation of any Capital Securities under this Clause 11.

12. PRECONDITIONS TO SPECIAL EVENT REDEMPTION OR CHANGE OF CONTROL EVENT REDEMPTION

- Prior to the publication of any notice of redemption pursuant to Clause 11 (*Redemption and repurchase of the Capital Securities*) (other than redemption pursuant to Clause 11.3 (*Voluntary total redemption (call option)*)), the Issuer shall deliver to the Agent a certificate signed by two authorised signatories of the Issuer stating that the relevant requirement or circumstance giving rise to the right to redeem the Capital Securities is satisfied.
- In addition, in the case of an Accounting Event, a Tax Deductibility Event or a Withholding Tax Event, the Issuer shall deliver to the Agent and the Issuing Agent the relevant opinion of the independent legal, accounting or other tax advisers (as applicable) to the effect that the relevant requirement or circumstance giving rise to the right to redeem is satisfied (save, in the case of a Withholding Tax Event, as to whether reasonable measures to avoid paying additional amounts are available to the Issuer). Such certificate and, if applicable, opinion, shall be conclusive and binding on the Holders provided that the opinion complies with the opinion requirements set out in this Clause 12.2.
- 12.3 Any redemption of the Capital Securities in accordance with Clause 11 (*Redemption and repurchase of the Capital Securities*) shall be conditional on all Deferred Interest being paid in full in accordance with the provisions of Clause 10.3 (*Mandatory settlement of Deferred Interest*) on or prior to the date of such redemption.

13. ADMISSION TO TRADING

The Issuer has the intention and shall use its best efforts (without assuming any legal or contractual obligation) to ensure that the Capital Securities:

- (a) are admitted to trading on the corporate bond list of Nasdaq Stockholm or, if such admission to trading is not possible to obtain or maintain, admitted to trading on another Regulated Market within thirty (30) days after the Issue Date; and
- (b) once admitted to trading on the relevant Regulated Market, continue being listed thereon (however, taking into account the rules and regulations of the relevant Regulated Market and the CSD (as amended from time to time) preventing trading in the Capital Securities in close connection to the redemption of the Capital Securities).

14. DEFAULT AND ENFORCEMENT

14.1 **Proceedings**

Without prejudice to the Issuer's right to defer the payment of interest under Clause 10 (*Optional Interest Deferral*), if a default is made by the Issuer for a period of thirty (30) days or more in relation to the payment of any interest, principal or premium in respect of the Capital Securities which is due and payable (a "**Default**"), then the Issuer shall be deemed to be in default under the Capital Securities and the Agent (acting on instructions of the Holders in accordance with these Terms and Conditions)

or (subject to Clause 21 (*No direct action by Holders*)) any Holder may institute proceedings for an Issuer Winding-up provided that such Default is still continuing.

14.1.2 In the event of an Issuer Winding-up, a Holder may, provided such Holder does not contravene a previously adopted resolution in accordance with Clause 15.4.2 or 15.4.3 (if any), either independently or through the Agent prove and/or claim in such Issuer Winding-up in respect of the Capital Securities, such claim being for such amount, and being subordinated in such manner, as is provided under Clause 3.2.

14.2 **Enforcement**

The Agent (acting on the instructions of the Holders in accordance with these Terms and Conditions) may institute such proceedings against the Issuer as it may think fit to enforce any term or condition binding on the Issuer under the Capital Securities but in no event shall the Issuer, by virtue of the institution of any such proceedings, be obliged to pay any sum or sums sooner than the same would otherwise have been payable by it.

14.3 Extent of Holders' Remedy

No remedy against the Issuer, other than as referred to in this Clause 14, shall be available to the Agent and the Holders in relation to the Capital Securities, whether for the recovery of amounts owing in respect of the Capital Securities or in respect of any breach by the Issuer of any of its other obligations under or in respect of the Capital Securities.

15. DECISIONS BY HOLDERS

15.1 Request for a decision

- 15.1.1 A request by the Agent for a decision by the Holders on a matter relating to the Terms and Conditions shall (at the option of the Agent) be dealt with at a Holders' Meeting or by way of a Written Procedure.
- Any request from the Issuer or a Holder (or Holders) representing at least ten (10) per cent. of the Adjusted Nominal Amount (such request shall, if made by several Holders, be made by them jointly) for a decision by the Holders on a matter relating to the Terms and Conditions shall be directed to the Agent and dealt with at a Holders' Meeting or by way a Written Procedure, as determined by the Agent. The person requesting the decision may suggest the form for decision making, but if it is in the Agent's opinion more appropriate that a matter is dealt with at a Holders' Meeting than by way of a Written Procedure, it shall be dealt with at a Holders' Meeting.
- 15.1.3 The Agent may refrain from convening a Holders' Meeting or instigating a Written Procedure if:
 - (a) the suggested decision must be approved by any person in addition to the Holders and such person has informed the Agent that an approval will not be given; or
 - (b) the suggested decision is not in accordance with applicable regulations.
- 15.1.4 The Agent shall not be responsible for the content of a notice for a Holders' Meeting or a communication regarding a Written Procedure unless and to the extent it contains information provided by the Agent.
- 15.1.5 Should the Agent not convene a Holders' Meeting or instigate a Written Procedure in accordance with these Terms and Conditions, without Clause 15.1.3 being applicable, the Issuer or the Holder(s) requesting a decision by the Holders may convene such Holders' Meeting or instigate such Written Procedure, as the case may be, instead. The Issuer or the Issuing Agent shall upon request provide the convening Holder(s) with the information available in the Debt Register in order to convene and hold the Holders' Meeting or instigate and carry out the Written Procedure, as the case may be. The Issuer or Holder(s), as applicable, shall supply to the Agent a copy of the dispatched notice or communication.

- 15.1.6 Should the Issuer want to replace the Agent, it may (i) convene a Holders' Meeting in accordance with Clause 15.2 (Convening of Holders' Meeting) or (ii) instigate a Written Procedure by sending communication in accordance with Clause 15.3 (Instigation of Written Procedure). After a request from the Holders pursuant to Clause 18.4.3, the Issuer shall no later than ten (10) Business Days after receipt of such request (or such later date as may be necessary for technical or administrative reasons) convene a Holders' Meeting in accordance with Clause 15.2. The Issuer shall inform the Agent before a notice for a Holders' Meeting or communication relating to a Written Procedure where the Agent is proposed to be replaced is sent and supply to the Agent a copy of the dispatched notice or communication.
- 15.1.7 Should the Issuer or any Holder(s) convene a Holders' Meeting or instigate a Written Procedure pursuant to Clause 15.1.5 or 15.1.6, then the Agent shall no later than five (5) Business Days' prior to dispatch of such notice or communication be provided with a draft thereof. The Agent may further append information from it together with the notice or communication, provided that the Agent supplies such information to the Issuer or the Holder(s), as the case may be, no later than one (1) Business Day prior to the dispatch of such notice or communication.

15.2 Convening of Holders' Meeting

- 15.2.1 The Agent shall convene a Holders' Meeting by way of notice to the Holders as soon as practicable and in any event no later than five (5) Business Days after receipt of a complete notice from the Issuer or the Holder(s) (or such later date as may be necessary for technical or administrative reasons).
- 15.2.2 The notice pursuant to Clause 15.2.1 shall include:
 - (a) time for the meeting;
 - (b) place for the meeting;
 - (c) a specification of the Record Date on which a person must be registered as a Bondholder in order to be entitled to exercise voting rights;
 - (d) a form of power of attorney;
 - (e) the agenda for the meeting;
 - (f) any applicable conditions precedent and conditions subsequent;
 - (g) the reasons for, and contents of, each proposal;
 - (h) if the proposal concerns an amendment to the Terms and Conditions, the details of such proposed amendment;
 - (i) if a notification by the Holders is required in order to attend the Holders' Meeting, information regarding such requirement; and
 - (j) information on where additional information (if any) will be published.
- 15.2.3 The Holders' Meeting shall be held no earlier than ten (10) Business Days and no later than thirty (30) Business Days after the effective date of the notice.
- 15.2.4 Without amending or varying these Terms and Conditions, the Agent may prescribe such further regulations regarding the convening and holding of a Holders' Meeting as the Agent may deem appropriate. Such regulations may include a possibility for Holders to vote without attending the meeting in person.

15.3 **Instigation of Written Procedure**

- 15.3.1 The Agent shall instigate a Written Procedure by way of sending a communication to the Holders as soon as practicable and in any event no later than five (5) Business Days after receipt of a complete communication from the Issuer or the Holder(s) (or such later date as may be necessary for technical or administrative reasons).
- 15.3.2 A communication pursuant to Clause 15.3.1 shall include:
 - (a) specification of the Record Date on which a person must be registered as a Holder in order to be entitled to exercise voting rights:
 - (b) instructions and directions on where to receive a form for replying to the request (such form to include an option to vote yes or no for each request) as well as a form of power of attorney;
 - (c) the stipulated time period within which the Holder must reply to the request (such time period to last at least ten (10) Business Days and not longer than thirty (30) Business Days from the effective date of the communication pursuant to Clause 15.3.1);
 - (d) any applicable conditions precedent and conditions subsequent;
 - (e) the reasons for, and contents of, each proposal;
 - (f) if a proposal concerns an amendment to the Terms and Conditions, the details of such proposed amendment;
 - (g) if the voting is to be made electronically, instructions for such voting; and
 - (h) information on where additional information (if any) will be published.
- 15.3.3 If so elected by the person requesting the Written Procedure and provided that it is also disclosed in the communication pursuant to Clause 15.3.1, when consents from Holders representing the requisite majority of the total Adjusted Nominal Amount pursuant to Clauses 15.4.2 and 15.4.3 have been received in a Written Procedure, the relevant decision shall be deemed to be adopted pursuant to Clause 15.4.2 or 15.4.3, as the case may be, even if the time period for replies in the Written Procedure has not yet expired.
- 15.3.4 The Agent may, during the Written Procedure, provide information to the Issuer by way of updates whether or not quorum requirements have been met and about the eligible votes received by the Agent, including the portion consenting or not consenting to the proposal(s) or refraining from voting (as applicable).

15.4 Majority, quorum and other provisions

- Only a Holder, or a person who has been provided with a power of attorney or other authorisation pursuant to Clause 7 (Right to act on behalf of a Holder) from a Holder:
 - (a) on the Record Date specified in the notice pursuant to Clause 15.2.2, in respect of a Holders' Meeting, or
 - (b) on the Record Date specified in the communication pursuant to Clause 15.3.2, in respect of a Written Procedure,

may exercise voting rights as a Holder at such Holders' Meeting or in such Written Procedure, provided that the relevant Capital Securities are included in the Adjusted Nominal Amount. Each whole Capital Security entitles to one vote and any fraction of a Capital Security voted for by a person shall be disregarded. Such Business Day specified pursuant to paragraph (a) or (b) above must fall no earlier than one (1) Business Day after the effective date of the notice or communication, as the case may be.

The following matters shall require the consent of Holders representing at least sixty-six and two thirds (66%) per cent. of the Adjusted Nominal Amount for which Holders are voting at a Holders'

Meeting or for which Holders reply in a Written Procedure in accordance with the instructions given pursuant to Clause 15.3.2:

- (a) a change of Issuer;
- (b) a change to the currency, denomination, status or transferability of the Capital Securities;
- (c) a change to the Interest Rate (including, for the avoidance of doubt, changes to the Margin or the Base Rate (other than as a result of an application of Clause 17 (*Replacement of Base Rate*)) or the Nominal Amount;
- (d) any delay of the due date for payment of any interest on the Capital Securities other than as permitted pursuant to Clause 10 (*Optional Interest Deferral*);
- (e) a reduction of the premium payable upon the redemption or repurchase of any Capital Security pursuant to Clause 11 (*Redemption and repurchase of the Capital Securities*);
- (f) a change to the terms dealing with the requirements for Holders' consent set out in this Clause 15.4 (*Majority, quorum and other provisions*);
- (g) an extension of the tenor of the Capital Securities or any delay of the due date for payment of any principal or interest on the Capital Securities;
- (h) a mandatory exchange of the Capital Securities for other securities; and
- (i) early redemption of the Capital Securities, other than as otherwise permitted or required by these Terms and Conditions.
- Any matter not covered by Clause 15.4.2, including for the avoidance of doubt the initiation of an Issuer Winding-up, shall require the consent of Holders representing more than fifty (50) per cent. of the Adjusted Nominal Amount for which Holders are voting at a Holders' Meeting or for which Holders reply in a Written Procedure in accordance with the instructions given pursuant to Clause 15.3.2. This includes, but is not limited to, any amendment to, or waiver of, the terms of the Terms and Conditions that does not require a higher majority (other than an amendment permitted pursuant to Clause 16.1(a) or (c)).
- 15.4.4 Quorum at a Holders' Meeting or in respect of a Written Procedure only exists if a Holder (or Holders) representing at least fifty (50) per cent. of the Adjusted Nominal Amount in case of a matter pursuant to Clause 15.4.2, and otherwise twenty (20) per cent. of the Adjusted Nominal Amount:
 - (a) if at a Holders' Meeting, attend the meeting in person or by other means prescribed by the Agent pursuant to Clause 15.2.4 (or appear through duly authorised representatives); or
 - (b) if in respect of a Written Procedure, reply to the request.
- 15.4.5 If a quorum exists for some but not all of the matters to be dealt with at a Holders' Meeting or by a Written Procedure, decisions may be taken in the matters for which a quorum exists.
- 15.4.6 If a quorum does not exist at a Holders' Meeting or in respect of a Written Procedure, the Agent or the Issuer shall convene a second Holders' Meeting (in accordance with Clause 15.2.1) or initiate a second Written Procedure (in accordance with Clause 15.3.1), as the case may be, provided that the person(s) who initiated the procedure for Holders' consent has confirmed that the relevant proposal is not withdrawn. For the purposes of a second Holders' Meeting or second Written Procedure pursuant to this Clause 15.4.6, the date of request of the second Holders' Meeting pursuant to Clause 15.2.1 or second Written Procedure pursuant to Clause 15.3.1, as the case may be, shall be deemed to be the relevant date when the quorum did not exist. The quorum requirement in Clause 15.4.4 shall not apply to such second Holders' Meeting or Written Procedure.
- 15.4.7 Any decision which extends or increases the obligations of the Issuer or the Agent, or limits, reduces or extinguishes the rights or benefits of the Issuer or the Agent, under the Terms and Conditions shall be subject to the Issuer's or the Agent's consent, as applicable.
- 15.4.8 A Holder holding more than one Capital Security need not use all its votes or cast all the votes to which it is entitled in the same way and may in its discretion use or cast some of its votes only.
- 15.4.9 The Issuer may not, directly or indirectly, pay or cause to be paid any consideration to or for the benefit of any owner of Capital Securities (irrespective of whether such person is a Holder) for or as

inducement to any consent under these Terms and Conditions, unless such consideration is offered to all Holders that vote in respect of the proposal at the relevant Holders' Meeting or in a Written Procedure within the time period stipulated for the consideration to be payable (such time period to be no less than ten (10) Business Days).

- 15.4.10 The Issuer may not, directly or indirectly, pay or cause to be paid any consideration to or for the benefit of any owner of Capital Securities (irrespective of whether such person is a Holder), nor make an offer to repurchase any Capital Securities, if receipt of such consideration or participation in such tender offer (as applicable) is conditional upon the Holder's consent to a proposal at a Holder's Meeting or in a Written Procedure.
- 15.4.11 A matter decided at a duly convened and held Holders' Meeting or by way of Written Procedure is binding on all Holders, irrespective of them being present or represented at the Holders' Meeting or responding in the Written Procedure. The Holders that have not adopted or voted for a decision shall not be liable for any damages that this may cause the Issuer or the other Holders.
- 15.4.12 All costs and expenses incurred by the Issuer or the Agent for the purpose of convening a Holders' Meeting or for the purpose of carrying out a Written Procedure, including reasonable fees to the Agent, shall be paid by the Issuer.
- 15.4.13 If a decision is to be taken by the Holders on a matter relating to the Terms and Conditions, the Issuer shall promptly at the request of the Agent provide the Agent with a certificate specifying the number of Capital Securities owned by Group Companies as per the Record Date for voting, irrespective of whether such person is a Holder. The Agent shall not be responsible for the accuracy of such certificate or otherwise be responsible for determining whether a Capital Security is owned by a Group Company.
- 15.4.14 Information about decisions taken at a Holders' Meeting or by way of a Written Procedure shall promptly be published on the websites of the Issuer and the Agent, provided that a failure to do so shall not invalidate any decision made or voting result achieved. The minutes from the relevant Holders' Meeting or Written Procedure shall at the request of a Holder be sent to it by the Issuer or the Agent, as applicable.

16. AMENDMENTS AND WAIVERS

- The Issuer and the Agent (acting on behalf of the Holders) may agree in writing to amend or waive any provision in the Terms and Conditions or any other document relating to the Capital Securities, provided that the Agent is satisfied that such amendment or waiver:
 - (a) is not detrimental to the interest of the Holders as a group;
 - (b) is made solely for the purpose of rectifying obvious errors and mistakes;
 - (c) is required by any applicable regulation, a court ruling or a decision by a relevant authority;
 - (d) is necessary for the purpose of having the Capital Securities admitted to trading on Nasdaq Stockholm (or any other Regulated Market, as applicable), provided that such amendment or waiver does not materially adversely affect the rights of the Holders;
 - (e) is made pursuant to Clause 17 (Replacement of Base Rate); or
 - (f) has been duly approved by the Holders in accordance with Clause 15 (*Decisions by Holders*) and it has received any conditions precedent specified for the effectiveness of the approval by the Holders.
- The Issuer may substitute or vary the terms of all (but not some only) of the outstanding Capital Securities without any requirement for the consent or approval of the Holders, so that they become or remain, as applicable, Qualifying Securities, provided that such substitution or variation does not itself give rise to any right of the Issuer to substitute or vary the terms of the Capital Securities in accordance with Clause 11.6 (Substitution or variation due to a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event) in relation to the Qualifying Securities so substituted or varied.

- The consent of the Holders is not necessary to approve the particular form of any amendment or waiver to these Terms and Conditions. It is sufficient if such consent approves the substance of the amendment or waiver.
- The Agent shall promptly notify the Holders of any amendments or waivers made in accordance with Clause 16.1, setting out the date from which the amendment or waiver will be effective, and ensure that any amendments to these Terms and Conditions are available on the websites of the Issuer and the Agent. The Issuer shall ensure that any amendments to these Terms and Conditions are duly registered with the CSD and each other relevant organisation or authority.
- An amendment to the Terms and Conditions shall take effect on the date determined by the Holders' Meeting, in the Written Procedure or by the Agent, as the case may be.

17. REPLACEMENT OF BASE RATE

17.1 General

- 17.1.1 Any determination or election to be made by an Independent Adviser, the Issuer or the Holders in accordance with the provisions of this Clause 17 shall at all times be made by such Independent Adviser, the Issuer or the Holders (as applicable) acting in good faith, in a commercially reasonable manner and by reference to relevant market data.
- 17.1.2 If a Base Rate Event has occurred, this Clause 17 shall take precedent over the fallbacks set out in paragraphs (a) to (d) of the definition of STIBOR.

17.2 **Definitions**

In this Clause 17:

- "Adjustment Spread" means a spread (which may be positive, negative or zero) or a formula or methodology for calculating a spread, or a combination thereof to be applied to a Successor Base Rate and that is:
- (a) formally recommended by any Relevant Nominating Body in relation to the replacement of the Base Rate; or
- (b) if paragraph (a) is not applicable, the adjustment spread that the Independent Adviser determines is reasonable to use in order to eliminate, to the extent possible, any transfer of economic value from one party to another as a result of a replacement of the Base Rate and is customarily applied in comparable debt capital market transactions.

"Base Rate Amendments" has the meaning set forth in Clause 17.3.4.

"Base Rate Event" means one or several of the following circumstances:

- (a) the Base Rate (for the relevant Interest Period) has ceased to exist or ceased to be published for at least five (5) consecutive Business Days as a result of the Base Rate (for the relevant Interest Period) ceasing to be calculated or administered;
- (b) a public statement or publication of information by (i) the supervisor of the Base Rate Administrator or (ii) the Base Rate Administrator that the Base Rate Administrator ceases to provide the applicable Base Rate (for the relevant Interest Period) permanently or indefinitely and, at the time of the statement or publication, no successor administrator has been appointed or is expected to be appointed to continue to provide the Base Rate;
- (c) a public statement or publication of information in each case by the supervisor of the Base Rate Administrator that the Base Rate (for the relevant Interest Period) is no longer representative of the underlying market which the Base Rate is intended to represent and the representativeness of the Base Rate will not be restored in the opinion of the supervisor of the Base Rate Administrator;

- (d) a public statement or publication of information in each case by the supervisor of the Base Rate Administrator with the consequence that it is unlawful for the Issuer or the Issuing Agent to calculate any payments due to be made to any Holder using the applicable Base Rate (for the relevant Interest Period) or it has otherwise become prohibited to use the applicable Base Rate (for the relevant Interest Period);
- (e) a public statement or publication of information in each case by the bankruptcy trustee of the Base Rate Administrator or by the trustee under the bank recovery and resolution framework (*krishanteringsregelverket*) containing the information referred to in paragraph (b) above; or
- (f) a Base Rate Event Announcement has been made and the announced Base Rate Event as set out in paragraphs (b) to (e) above will occur within six (6) months.

"Base Rate Event Announcement" means a public statement or published information as set out in paragraph (b) to (e) of the definition of Base Rate Event that any event or circumstance specified therein will occur.

"Independent Adviser" means an independent financial institution or adviser of repute in the debt capital markets where the Base Rate is commonly used.

"Relevant Nominating Body" means, subject to applicable law, firstly any relevant supervisory authority, secondly any applicable central bank, or any working group or committee of any of them, or thirdly the Financial Stability Council (*Finansiella stabilitetsrådet*) or any part thereof.

"Successor Base Rate" means:

- (a) a screen or benchmark rate, including the methodology for calculating term structure and calculation methods in respect of debt instruments with similar interest rate terms as the Capital Securities, which is formally recommended as a successor to or replacement of the Base Rate by a Relevant Nominating Body; or
- (b) if there is no such rate as described in paragraph (a), such other rate as the Independent Adviser determines is most comparable to the Base Rate.

For the avoidance of doubt, in the event that a Successor Base Rate ceases to exist, this definition shall apply *mutatis mutandis* to such new Successor Base Rate.

17.3 Determination of Base Rate, Adjustment Spread and Base Rate Amendments

- 17.3.1 Without prejudice to Clause 17.3.2, upon a Base Rate Event Announcement, the Issuer may, if it is possible to determine a Successor Base Rate at such point of time, at any time before the occurrence of the relevant Base Rate Event at the Issuer's expense appoint an Independent Adviser to initiate the procedure to determine a Successor Base Rate, the Adjustment Spread and any Base Rate Amendments for purposes of determining, calculating and finally deciding the applicable Base Rate. For the avoidance of doubt, the Issuer will not be obliged to take any such actions until obliged to do so pursuant to Clause 17.3.2.
- 17.3.2 If a Base Rate Event has occurred, the Issuer shall use all commercially reasonable endeavours to, as soon as reasonably practicable and at the Issuer's expense, appoint an Independent Adviser to initiate the procedure to determine, as soon as commercially reasonable, a Successor Base Rate, the Adjustment Spread and any Base Rate Amendments for purposes of determining, calculating, and finally deciding the applicable Base Rate.
- 17.3.3 If the Issuer fails to appoint an Independent Adviser in accordance with Clause 17.3.2, the Holders shall, if so decided at a Holders' Meeting or by way of Written Procedure, be entitled to appoint an Independent Adviser (at the Issuer's expense) for the purposes set forth in Clause 17.3.2. If the Issuer fails to carry out any other actions set forth in Clause 17.3 to 17.6, the Agent (acting on the instructions of the Holders) may to the extent necessary effectuate any Base Rate Amendments without the Issuer's cooperation.

- 17.3.4 The Independent Adviser shall also initiate the procedure to determine any technical, administrative or operational changes required to ensure the proper operation of a Successor Base Rate or to reflect the adoption of such Successor Base Rate in a manner substantially consistent with market practice ("Base Rate Amendments").
- 17.3.5 Provided that a Successor Base Rate, the applicable Adjustment Spread and any Base Rate Amendments have been finally decided no later than prior to the relevant Quotation Day in relation to the next succeeding Interest Period, they shall become effective with effect from and including the commencement of the next succeeding Interest Period, always subject to any technical limitations of the CSD and any calculations methods applicable to such Successor Base Rate.

17.4 **Interim measures**

- 17.4.1 If a Base Rate Event set out in any of the paragraphs (a) to (e) of the Base Rate Event definition has occurred but no Successor Base Rate and Adjustment Spread have been finally decided prior to the relevant Quotation Day in relation to the next succeeding Interest Period or if such Successor Base Rate and Adjustment Spread have been finally decided but due to technical limitations of the CSD, cannot be applied in relation to the relevant Quotation Day, the Interest Rate applicable to the next succeeding Interest Period shall be:
 - (a) if the previous Base Rate is available, determined pursuant to the terms that would apply to the determination of the Base Rate as if no Base Rate Event had occurred; or
 - (b) if the previous Base Rate is no longer available or cannot be used in accordance with applicable law or regulation, equal to the Interest Rate determined for the immediately preceding Interest Period.
- 17.4.2 For the avoidance of doubt, Clause 17.4.1 shall apply only to the relevant next succeeding Interest Period and any subsequent Interest Periods are subject to the subsequent operation of, and to adjustments as provided in, this Clause 17. This will however not limit the application of Clause 17.4.1 for any subsequent Interest Periods, should all relevant actions provided in this Clause 17 have been taken, but without success.

17.5 Notices etc.

Prior to the Successor Base Rate, the applicable Adjustment Spread and any Base Rate Amendments become effective the Issuer shall promptly, following the final decision by the Independent Adviser of any Successor Base Rate, Adjustment Spread and any Base Rate Amendments, give notice thereof to the Agent, the Issuing Agent and the Holders in accordance with Clause 23 (*Notices*) and the CSD. The notice shall also include information about the effective date of the amendments. If the Capital Securities are admitted to trading on a stock exchange, the Issuer shall also give notice of the amendments to the relevant stock exchange.

17.6 Variation upon replacement of Base Rate

- 17.6.1 No later than giving the Agent notice pursuant to Clause 17.5, the Issuer shall deliver to the Agent a certificate signed by the Independent Adviser and the CEO, CFO or any other duly authorised signatory of the Issuer (subject to Clause 17.3.3) confirming the relevant Successor Base Rate, the Adjustment Spread and any Base Rate Amendments, in each case as determined and decided in accordance with the provisions of this Clause 17. The Successor Base Rate the Adjustment Spread and any Base Rate Amendments (as applicable) specified in such certificate will, in the absence of manifest error or bad faith in any decision, be binding on the Issuer, the Agent, the Issuing Agent and the Holders.
- 17.6.2 Subject to receipt by the Agent of the certificate referred to in Clause 17.6.1, the Issuer and the Agent shall, at the request and expense of the Issuer, without the requirement for any consent or approval of the Holders, without undue delay effect such amendments to the Finance Documents as may be required by the Issuer in order to give effect to this Clause 17.

17.6.3 The Agent and the Issuing Agent shall always be entitled to consult with external experts prior to amendments are effected pursuant to this Clause 17. Neither the Agent nor the Issuing Agent shall be obliged to concur if in the reasonable opinion of the Agent or the Issuing Agent (as applicable), doing so would impose more onerous obligations upon it or expose it to any additional duties, responsibilities or liabilities or reduce or amend the protective provisions afforded to the Agent or the Issuing Agent in the Finance Documents.

17.7 Limitation of liability for the Independent Adviser

Any Independent Adviser appointed pursuant to Clause 17.3 shall not be liable whatsoever for damage or loss caused by any determination, action taken or omitted by it under or in connection with any Finance Document, unless directly caused by its gross negligence or wilful misconduct. The Independent Adviser shall never be responsible for indirect or consequential loss.

18. THE AGENT

18.1 **Appointment of the Agent**

- 18.1.1 By subscribing for Capital Securities, each initial Holder appoints the Agent to act as its agent in all matters relating to the Capital Securities and the Terms and Conditions, and authorises the Agent to act on its behalf (without first having to obtain its consent, unless such consent is specifically required by these Terms and Conditions) in any legal or arbitration proceedings relating to the Capital Securities held by such Holder, including the winding-up, dissolution, liquidation, company reorganisation (*företagsrekonstruktion*) or bankruptcy (*konkurs*) (or its equivalent in any other jurisdiction) of the Issuer and in relation to any mandatory exchange of the Capital Securities for other securities (including, for the avoidance of doubt, a right for the Agent to subscribe for any such new securities on behalf of the relevant Holder). By acquiring Capital Securities, each subsequent Holder confirms such appointment and authorisation for the Agent to act on its behalf.
- 18.1.2 Each Holder shall immediately upon request provide the Agent with any such documents, including a written power of attorney (in form and substance satisfactory to the Agent), that the Agent deems necessary for the purpose of exercising its rights and/or carrying out its duties under the Terms and Conditions. The Agent is under no obligation to represent a Holder which does not comply with such request.
- 18.1.3 The Issuer shall promptly upon request provide the Agent with any documents and other assistance (in form and substance satisfactory to the Agent), that the Agent deems necessary for the purpose of exercising its rights and/or carrying out its duties under the Terms and Conditions.
- 18.1.4 The Agent is entitled to fees for all its work in such capacity and to be indemnified for costs, losses and liabilities on the terms set out in the Terms and Conditions and the Agency Agreement and the Agent's obligations as Agent under the Terms and Conditions are conditioned upon the due payment of such fees and indemnifications.
- 18.1.5 The Agent may act as agent or trustee for several issues of securities or other loans issued by or relating to the Issuer and other Group Companies notwithstanding potential conflicts of interest.

18.2 **Duties of the Agent**

- 18.2.1 The Agent shall represent the Holders in accordance with the Terms and Conditions.
- 18.2.2 When acting pursuant to the Terms and Conditions, the Agent is always acting with binding effect on behalf of the Holders. The Agent is never acting as an advisor to the Holders or the Issuer. Any advice or opinion from the Agent does not bind the Holders or the Issuer.
- 18.2.3 When acting pursuant to the Terms and Conditions, the Agent shall carry out its duties with reasonable care and skill in a proficient and professional manner.
- 18.2.4 The Agent shall treat all Holders equally and, when acting pursuant to the Terms and Conditions, act with regard only to the interests of the Holders as a group and shall not be required to have regard to

- the interests or to act upon or comply with any direction or request of any other person, other than as explicitly stated in the Terms and Conditions.
- 18.2.5 The Agent is always entitled to delegate its duties to other professional parties and to engage external experts when carrying out its duties as agent, without having to first obtain any consent from the Holders or the Issuer. The Agent shall however remain liable for any actions of such parties if such parties are performing duties of the Agent under the Terms and Conditions.
- 18.2.6 The Issuer shall on demand by the Agent pay all costs for external experts engaged by it in connection with (i) matters relating to the Issuer or the Terms and Conditions which the Agent reasonably believes may be detrimental to the interests of the Holders under the Terms and Conditions, (ii) any Holders' Meeting or Written Procedure, or (iii) any amendment (whether contemplated by the Terms and Conditions or not) or waiver under the Terms and Conditions.
- 18.2.7 The Agent shall, as applicable, enter into agreements with the CSD, and comply with such agreement and the CSD Regulations applicable to the Agent, as may be necessary in order for the Agent to carry out its duties under the Terms and Conditions.
- 18.2.8 Other than as specifically set out in the Terms and Conditions, the Agent shall not be obliged to monitor (i) the financial condition of the Issuer and the Group, (ii) the performance, default or any breach by the Issuer or any other party of its obligations under the Terms and Conditions, or (iii) whether any other event specified in the Terms and Conditions has occurred or is expected to occur. Should the Agent not receive such information, the Agent is entitled to assume that no such event or circumstance exists or can be expected to occur, provided that the Agent does not have actual knowledge of such event or circumstance.
- The Agent shall ensure that it receives evidence satisfactory to it that the Terms and Conditions are duly authorised and executed. The Issuer shall promptly upon request provide the Agent with such documents and evidence as the Agent reasonably considers necessary for the purpose of being able to comply with this Clause 18.2.9. Other than as set out above, the Agent shall neither be liable to the Issuer or the Holders for damage due to any documents and information delivered to the Agent not being accurate, correct and complete, unless it has actual knowledge to the contrary, nor be liable for the content, validity, perfection or enforceability of such documents.
- 18.2.10 Notwithstanding any other provision of the Terms and Conditions to the contrary, the Agent is not obliged to do or omit to do anything if it would or might in its reasonable opinion constitute a breach of any regulation.
- 18.2.11 If in the Agent's reasonable opinion the cost, loss or liability which it may incur (including reasonable fees to the Agent) in complying with instructions of the Holders, or taking any action at its own initiative, will not be covered by the Issuer, the Agent may refrain from acting in accordance with such instructions, or taking such action, until it has received such funding or indemnities (or adequate Security has been provided therefore) as it may reasonably require.
- 18.2.12 The Agent shall give a notice to the Holders:
- 18.2.13 before it ceases to perform its obligations under the Terms and Conditions by reason of the non-payment by the Issuer of any fee or indemnity due to the Agent under the Terms and Conditions or the Agency Agreement; or
- 18.2.14 if it refrains from acting for any reason described in Clause 18.2.11.
- 18.2.15 Subject to applicable regulations, the Agent shall promptly upon the reasonable request by a Holder forward by post any information from such Holder to the Holders which relates to the Capital Securities. The Agent may require that the requesting Holder reimburses any costs or expenses incurred, or to be incurred, by it in doing so (including a reasonable fee for its work).
- 18.2.16 Subject to the restrictions of a non-disclosure agreement entered into by the Agent in connection with these Terms and Conditions, the Agent is entitled to disclose to the Holders any document, information, event or circumstance directly or indirectly relating to the Issuer or the Capital Securities. Notwithstanding the foregoing, the Agent may if it considers it to be beneficial to the interests of the Holders delay disclosure or refrain from disclosing certain information.
- 18.2.17 If a committee representing the Holders' interests under the Terms and Conditions has been appointed by the Holders in accordance with Clause 15 (*Decisions by Holders*), the members of such committee

may agree with the Issuer not to disclose information received from the Issuer, provided that it, in the reasonable opinion of such members, is beneficial to the interests of the Holders. The Agent shall be a party to such agreement and receive the same information from the Issuer as the members of the committee.

18.3 **Liability for the Agent**

- 18.3.1 The Agent will not be liable to the Holders for damage or loss caused by any action taken or omitted by it under or in connection with these Terms and Conditions, unless directly caused by its negligence or wilful misconduct. The Agent shall never be responsible for indirect or consequential loss.
- 18.3.2 The Agent shall not be considered to have acted negligently if it has acted in accordance with advice from or opinions of reputable external experts provided to the Agent or if the Agent has acted with reasonable care in a situation when the Agent considers that it is detrimental to the interests of the Holders to delay the action in order to first obtain instructions from the Holders.
- 18.3.3 The Agent shall not be liable for any delay (or any related consequences) in crediting an account with an amount required pursuant to the Terms and Conditions to be paid by the Agent to the Holders, provided that the Agent has taken all necessary steps as soon as reasonably practicable to comply with the regulations or operating procedures of any recognised clearing or settlement system used by the Agent for that purpose.
- 18.3.4 The Agent shall have no liability to the Issuer or the Holders for damage caused by the Agent acting in accordance with instructions of the Holders given in accordance with the Terms and Conditions.
- 18.3.5 Any liability towards the Issuer which is incurred by the Agent in acting under, or in relation to, the Terms and Conditions shall not be subject to set-off against the obligations of the Issuer to the Holders under the Terms and Conditions.

18.4 **Replacement of the Agent**

- Subject to Clause 18.4.6, the Agent may resign by giving notice to the Issuer and the Holders, in which case the Holders shall appoint a successor Agent at a Holders' Meeting convened by the retiring Agent or by way of Written Procedure initiated by the retiring Agent.
- Subject to Clause 18.4.6, if the Agent is Insolvent, the Agent shall be deemed to resign as Agent and the Issuer shall within ten (10) Business Days appoint a successor Agent which shall be an independent financial institution or other reputable company which regularly acts as agent under debt issuances.
- 18.4.3 A Holder (or Holders) representing at least ten (10) per cent. of the Adjusted Nominal Amount may, by notice to the Issuer (such notice shall, if given by several Holders, be given by them jointly), require that a Holders' Meeting is held for the purpose of dismissing the Agent and appointing a new Agent. The Issuer may, at a Holders' Meeting convened by it or by way of Written Procedure initiated by it, propose to the Holders that the Agent be dismissed and a new Agent appointed.
- 18.4.4 If the Holders have not appointed a successor Agent within ninety (90) days after:
 - (a) the earlier of the notice of resignation was given or the resignation otherwise took place; or
 - (b) the Agent was dismissed through a decision by the Holders,

the Issuer shall within thirty (30) days thereafter appoint a successor Agent which shall be an independent financial institution or other reputable company with the necessary resources to act as agent under debt issuances.

18.4.5 The retiring Agent shall, at its own cost, make available to the successor Agent such documents and records and provide such assistance as the successor Agent may reasonably request for the purposes of performing its functions as Agent under the Terms and Conditions.

- 18.4.6 The Agent's resignation or dismissal shall only take effect upon the earlier of:
 - (a) the appointment of a successor Agent and acceptance by such successor Agent of such appointment and the execution of all necessary documentation to effectively substitute the retiring Agent; and
 - (b) the period pursuant to Clause 18.4.4 having lapsed.
- 18.4.7 Upon the appointment of a successor, the retiring Agent shall be discharged from any further obligation in respect of the Terms and Conditions but shall remain entitled to the benefit of the Terms and Conditions and remain liable under the Terms and Conditions in respect of any action which it took or failed to take whilst acting as Agent. Its successor, the Issuer and each of the Holders shall have the same rights and obligations amongst themselves under the Terms and Conditions as they would have had if such successor had been the original Agent.
- In the event that there is a change of the Agent in accordance with this Clause 18.4, the Issuer shall execute such documents and take such actions as the new Agent may reasonably require for the purpose of vesting in such new Agent the rights, powers and obligation of the Agent and releasing the retiring Agent from its further obligations under the Terms and Conditions and the Agency Agreement. Unless the Issuer and the new Agent agree otherwise, the new Agent shall be entitled to the same fees and the same indemnities as the retiring Agent.

19. THE ISSUING AGENT

- 19.1 The Issuer shall when necessary appoint an Issuing Agent to manage certain specified tasks under these Terms and Conditions and in accordance with the legislation, rules and regulations applicable to and/or issued by the CSD and relating to the Capital Securities. The Issuing Agent shall be a commercial bank or securities institution approved by the CSD.
- The Issuer shall ensure that the Issuing Agent enters into agreements with the CSD, and comply with such agreement and the CSD Regulations applicable to the Issuing Agent, as may be necessary in order for the Issuing Agent to carry out its duties relating to the Capital Securities.
- 19.3 The Issuing Agent will not be liable to the Holders for damage or loss caused by any action taken or omitted by it under or in connection with the Terms and Conditions, unless directly caused by its gross negligence or wilful misconduct. The Issuing Agent shall never be responsible for indirect or consequential loss.

20. THE CSD

- The Issuer has appointed the CSD to manage certain tasks under these Terms and Conditions and in accordance with the CSD Regulations and the other regulations applicable to the Capital Securities.
- The CSD may retire from its assignment or be dismissed by the Issuer, provided that the Issuer has effectively appointed a replacement CSD that accedes as CSD at the same time as the old CSD retires or is dismissed and provided also that the replacement does not have a negative effect on any Holder or the admission to trading of the Capital Securities on the Regulated Market. The replacing CSD must be authorised to professionally conduct clearing operations pursuant to the Central Securities Depository Regulation (Regulation (EU) No 909/2014) and be authorised as a central securities depository in accordance with the Central Securities Depositories and Financial Instruments Accounts Act.

21. NO DIRECT ACTIONS BY HOLDERS

- A Holder may not take any steps whatsoever against the Issuer to enforce or recover any amount due or owing to it pursuant to the Terms and Conditions, or to initiate, support or procure the winding-up, dissolution, liquidation, company reorganisation or bankruptcy in any jurisdiction of any Group Company in relation to any of the obligations and liabilities of the Issuer or any Subsidiary under the Terms and Conditions. Such steps may only be taken by the Agent.
- Clause 21.1 shall not apply if the Agent has been instructed by the Holders in accordance with the Terms and Conditions to take certain actions but fails for any reason to take, or is unable to take (for any reason other than a failure by a Holder to provide documents in accordance with Clause 17.1.2), such actions within a reasonable period of time and such failure or inability is continuing. However, if the failure to take certain actions is caused by the non-payment of any fee or indemnity due to the Agent under the Terms and Conditions or the Agency Agreement or by any reason described in Clause 17.2.11, such failure must continue for at least forty (40) Business Days after notice pursuant to Clause 17.2.12 before a Holder may take any action referred to in Clause 21.1.

22. TIME-BAR

- The right to receive repayment of the principal of the Capital Securities shall be time-barred and become void ten (10) years from the relevant Redemption Date. The right to receive payment of interest (excluding any capitalised interest) shall be time-barred and become void three (3) years from the relevant due date for payment. The Issuer is entitled to any funds set aside for payments in respect of which the Holders' right to receive payment has been time-barred and has become void.
- If a limitation period is duly interrupted in accordance with the Swedish Act on Limitations (preskriptionslag (1981:130)), a new limitation period of ten (10) years with respect to the right to receive repayment of the principal of the Capital Securities, and of three (3) years with respect to receive payment of interest (excluding capitalised interest) will commence, in both cases calculated from the date of interruption of the limitation period, as such date is determined pursuant to the provisions of the Swedish Act on Limitations.

23. NOTICES

- 23.1 Any notice or other communication to be made under or in connection with the Terms and Conditions:
 - if to the Agent, shall be given at the address specified on its website www.nordictrustee.com on the Business Day prior to dispatch or, if sent by email by the Issuer, to the email address notified by the Agent to the Issuer from time to time;
 - (b) if to the Issuer, shall be given at the address registered with the Swedish Companies Registration Office on the Business Day prior to dispatch or, if sent by email by the Agent, to the email address notified by the Issuer to the Agent from time to time; and
 - (c) if to the Holders, shall be given at their addresses registered with the CSD on a date selected by the sending person which falls no more than five (5) Business Days prior to the date on which the notice or communication is sent, and by either courier delivery (if practically possible) or letter for all Holders. A Notice to the Holders shall also be published on the websites of the Issuer and the Agent.
- Any notice or other communication made by one person to another under or in connection with the Terms and Conditions shall be sent by way of courier, personal delivery or letter, or, if between the Issuer and the Agent, by email, and will only be effective:
 - (a) in case of courier or personal delivery, when it has been left at the address specified in Clause 23.1;

- (b) in case of letter, three (3) Business Days after being deposited postage prepaid in an envelope addressed to the address specified in Clause 23.1; or
- (c) in case of email, when received in readable form by the email recipient.
- 23.3 Any notice or other communication pursuant to the Terms and Conditions shall be in English.
- Failure to send a notice or other communication to a Holder or any defect in it shall not affect its sufficiency with respect to other Holders.

24. FORCE MAJEURE

- 24.1 Neither the Agent nor the Issuing Agent shall be held responsible for any damage arising out of any legal enactment, or any measure taken by a public authority, or war, strike, lockout, boycott, blockade, natural disaster, insurrection, civil commotion, terrorism or any other similar circumstance (a "Force Majeure Event"). The reservation in respect of strikes, lockouts, boycotts and blockades applies even if the Agent or the Issuing Agent itself takes such measures, or is subject to such measures.
- 24.2 Should a Force Majeure Event arise which prevents the Agent or the Issuing Agent from taking any action required to comply with these Terms and Conditions, such action may be postponed until the obstacle has been removed.
- 24.3 The provisions in this Clause 24 apply unless they are inconsistent with the provisions of the Central Securities Depositories and Financial Instruments Accounts Act which provisions shall take precedence.

25. GOVERNING LAW AND JURISDICTION

- These Terms and Conditions, and any non-contractual obligations arising out of or in connection therewith, shall be governed by and construed in accordance with the laws of Sweden.
- The Issuer submits to the non-exclusive jurisdiction of the District Court of Stockholm (*Stockholms tingsrätt*).

SCHEDULE 1

CONDITIONS PRECEDENT

1. CORPORATE DOCUMENTS

- (a) Copies of the articles of association and an up-to-date certificate of registration of the Issuer.
- (b) A copy of a resolution from the board of directors of the Issuer:
 - (i) approving the issue of the Capital Securities, the terms of the Terms and Conditions and the Agency Agreement, and resolving to enter into such documents and any other documents necessary in connection therewith;
 - (ii) authorising a specified person or persons to execute the Terms and Conditions and the Agency Agreement; and
 - (iii) authorising a specified person or persons, on its behalf, to execute all documents and notices to be executed by it or in connection with the Terms and Conditions or the Agency Agreement
- (c) A copy of a passport or driver's license which includes a specimen of the signature of each person authorised by the resolution referred to in paragraph (b) above and actually signing the Terms and Conditions and/or the Agency Agreement.

2. FINANCE DOCUMENTS

- (a) A copy of the Terms and Conditions duly executed by the Issuer.
- (b) A copy of the Agency Agreement duly executed by the Issuer.

3. OTHER DOCUMENTS AND EVIDENCE

Such other documents and evidence as is agreed between the Agent and the Issuer.

TERMS AND CONDITIONS OF THE FLOATING RATE CAPITAL SECURITIES (NC7)



TERMS AND CONDITIONS FOR SWEDAVIA AB (PUBL)

UP TO SEK 3,000,000,000 SUBORDINATED PERPETUAL FLOATING RATE CALLABLE CAPITAL SECURITIES

ISIN: SE0026853319

First Issue Date: 19 November 2025

SELLING RESTRICTIONS

No action is being taken that would or is intended to permit a public offering of the Capital Securities or the possession, circulation or distribution of this document or any other material relating to the Issuer or the Capital Securities in any jurisdiction other than Sweden, where action for that purpose is required. Persons into whose possession this document comes are required by the Issuer to inform themselves about, and to observe, any applicable restrictions.

PRIVACY NOTICE

The Issuer, the Agent and the Issuing Agent may collect and process personal data relating to the Holders, the Holders' representatives or agents, and other persons nominated to act on behalf of the Holders pursuant to the Terms and Conditions (name, contact details and, when relevant, holding of Capital Securities). The personal data relating to the Holders is primarily collected from the registry kept by the CSD. The personal data relating to other persons is primarily collected directly from such persons.

The personal data collected will be processed by the Issuer, the Agent and the Issuing Agent for the following purposes:

- (e) to exercise their respective rights and fulfil their respective obligations under the Terms and Conditions:
- (f) to manage the administration of the Capital Securities and payments under the Capital Securities;
- (g) to enable the Holders' to exercise their rights under the Terms and Conditions; and
- (h) to comply with their obligations under applicable laws and regulations.

The processing of personal data by the Issuer, the Agent and the Issuing Agent in relation to items (a) to (c) (inclusive) is based on their legitimate interest to exercise their respective rights and to fulfil their respective obligations under the Terms and Conditions. In relation to item (d), the processing is based on the fact that such processing is necessary for compliance with a legal obligation incumbent on the Issuer, the Agent or the Issuing Agent. Unless otherwise required or permitted by law, the personal data collected will not be kept longer than necessary given the purpose of the processing.

Personal data collected may be shared with third parties, such as the CSD, when necessary to fulfil the purpose for which such data is processed.

Subject to any legal preconditions, the applicability of which have to be assessed in each individual case, data subjects have the rights as follows. Data subjects have right to get access to their personal data and may request the same in writing at the address of the Issuer, the Agent and the Issuing Agent, respectively. In addition, data subjects have the right to (i) request that personal data is rectified or erased, (ii) object to specific processing, (iii) request that the processing be restricted and (iv) receive personal data provided by themselves in machine-readable format.

Data subjects are also entitled to lodge complaints with the relevant supervisory authority if dissatisfied with the processing carried out.

The Issuer's, the Agent's and the Issuing Agent's addresses, and the contact details for their respective Data Protection Officers (if applicable), are found on their websites www.swedavia.se, www.nordictrustee.com and www.nordea.com.

TABLE OF CONTENTS

1.	DEFINITIONS AND CONSTRUCTION	1
2.	THE CAPITAL SECURITIES	8
3.	STATUS OF THE CAPITAL SECURITIES	9
4.	USE OF PROCEEDS	10
5.	CONDITIONS PRECEDENT	10
6.	CAPITAL SECURITIES IN BOOK-ENTRY FORM	11
7.	RIGHT TO ACT ON BEHALF OF A HOLDER	12
8.	PAYMENTS IN RESPECT OF THE CAPITAL SECURITIES	12
9.	INTEREST	13
10.	OPTIONAL INTEREST DEFERRAL	14
11.	REDEMPTION AND REPURCHASE OF THE CAPITAL SECURITIES	15
12.	PRECONDITIONS TO SPECIAL EVENT REDEMPTION OR CHANGE OF CONT EVENT REDEMPTION	
13.	ADMISSION TO TRADING	17
14.	DEFAULT AND ENFORCEMENT	17
15.	DECISIONS BY HOLDERS	18
16.	AMENDMENTS AND WAIVERS	23
17.	REPLACEMENT OF BASE RATE	24
18.	THE AGENT	28
19.	THE ISSUING AGENT	31
20.	THE CSD	32
21.	NO DIRECT ACTIONS BY HOLDERS	32
22.	TIME-BAR	32
23.	NOTICES	33
24.	FORCE MAJEURE	34
25.	GOVERNING LAW AND JURISDICTION	34
SCH	HEDULE	
1.	CONDITIONS PRECEDENT	35

1. DEFINITIONS AND CONSTRUCTION

1.1 **Definitions**

In these terms and conditions (the "Terms and Conditions"):

- "Account Operator" means a bank or other party duly authorised to operate as an account operator pursuant to the Central Securities Depositories and Financial Instruments Accounts Act and through which a Holder has opened a Securities Account in respect of its Capital Securities.
- "Accounting Event" means the receipt by the Issuer of an opinion of an independent and authorised accountant (*auktoriserad revisor*) in Sweden (experienced in such matters) to the effect that, as a result of a change in the Accounting Principles or interpretation thereof, the equity treatment of the Capital Securities as "equity" in full in the Issuer's consolidated financial statements has or will cease.
- "Accounting Principles" means the international financial reporting standards (IFRS) within the meaning of Regulation 1606/2002/EC (or as otherwise adopted or amended from time to time).
- "Adjusted Nominal Amount" means the Total Nominal Amount less the aggregate Nominal Amount of all Capital Securities owned by a Group Company, irrespective of whether such person is directly registered as owner of such Capital Securities.
- "Agency Agreement" means the agency agreement entered into before the First Issue Date, between the Issuer and the Agent, or any replacement agency agreement entered into after the First Issue Date between the Issuer and an agent.
- "Agent" means Nordic Trustee & Agency AB (publ), Swedish Reg. No. 556882-1879, or another party replacing it, as Holders' agent, in accordance with these Terms and Conditions.
- "Base Rate" means STIBOR or any reference rate replacing STIBOR in accordance with Clause 17 (Replacement of Base Rate).
- "Base Rate Administrator" means the Swedish Financial Benchmark Facility AB (SFBF) or any person replacing it as administrator of the Base Rate.
- "Business Day" means a day in Sweden other than a Sunday or other public holiday. Saturdays, Midsummer Eve (*midsommarafton*), Christmas Eve (*julafton*) and New Year's Eve (*nyårsafton*) shall for the purpose of this definition be deemed to be public holidays.
- "Business Day Convention" means the first following day that is a Business Day unless that day falls in the next calendar month, in which case that date will be the first preceding day that is a Business Day.
- "Capital Security" means a debt instrument (*skuldförbindelse*) for the Nominal Amount and of the type set forth in Chapter 1 Section 3 of the Central Securities Depositories and Financial Instruments Accounts Act and which are governed by and issued by the Issuer under these Terms and Conditions, including the Initial Capital Securities and any Subsequent Capital Securities.
- "Central Securities Depositories and Financial Instruments Accounts Act" means the Swedish Central Securities Depositories and Financial Instruments Accounts Act (lag (1998:1479) om värdepapperscentraler och kontoföring av finansiella instrument).
- "Change of Control Event" means an event where the Government of Sweden ceases to, directly or indirectly, own or control less than 100 per cent. of the shares and the votes in the Issuer.
- "Change of Control Step-up Date" means the date falling six (6) months after the date on which a Change of Control Event has occurred.
- "CSD" means the Issuer's central securities depository and registrar in respect of the Capital Securities, Euroclear Sweden AB, Swedish Reg. No. 556112-8074, P.O. Box 191, 101 23 Stockholm, Sweden, or another party replacing it, as CSD, in accordance with these Terms and Conditions.
- "CSD Regulations" means the CSD's rules and regulations applicable to the Issuer, the Agent and the Capital Securities from time to time.

"**Debt Register**" means the debt register (*skuldbok*) kept by the CSD in respect of the Capital Securities in which a Holder is registered.

"Deferred Interest" has the meaning ascribed to it in Clause 10.1.3.

"Deferred Interest Payment Event" means any one or more of the following events:

- (a) declaration or payment of any distribution or dividend or any other payment made by the Issuer on its share capital or any other obligation of the Issuer which ranks or is expressed by its terms to rank junior to the Capital Securities or any Parity Securities;
- (b) declaration or payment of any distribution or dividend or any other payment made by the Issuer or any Subsidiary of the Issuer, as the case may be, on any Parity Securities;
- (c) redemption, repurchase, repayment, cancellation, reduction or other acquisition by the Issuer or any Subsidiary of the Issuer of any shares of the Issuer or any other obligation of the Issuer which ranks or is expressed by its terms to rank junior to the Capital Securities or any Parity Securities; and/or
- (d) redemption, repurchase, repayment, cancellation, reduction or other acquisition by the Issuer or any Subsidiary of the Issuer of any Parity Securities,

save for:

- (i) in each case, any compulsory distribution, dividend, other payment, redemption, repurchase, repayment, cancellation, reduction or other acquisition required by the terms of such securities or by mandatory operation of applicable law; and
- (ii) in the case of paragraph (c) above only, any redemption, repurchase, repayment, cancellation, reduction or other acquisition that is executed in connection with, or for the purpose of:
 - (A) any reduction of the quota value of the share capital of the Issuer without a corresponding return of cash, capital or assets to shareholders of the Issuer; or
 - (B) any share buyback programme then in force and duly approved by the shareholders' general meeting of the Issuer or the relevant Subsidiary of the Issuer (as applicable) or any existing or future stock option plan or free share allocation plan or other incentive plan, in all cases, reserved for directors, officers and/or employees of the Issuer or the relevant Subsidiary of the Issuer or any associated hedging transaction.

"First Call Date" means the date falling seven (7) years after the First Issue Date or, to the extent such day is not a Business Day, the Business Day following from an application of the Business Day Convention.

"First Issue Date" means 19 November 2025.

"Force Majeure Event" has the meaning set forth in Clause 24.1.

"Group" means the Issuer and its Subsidiaries from time to time (each a "Group Company").

"Holder" means the person who is registered on a Securities Account as direct registered owner (direktregistrerad ägare) or nominee (förvaltare) with respect to a Capital Security.

"Holders' Meeting" means a meeting among the Holders held in accordance with Clauses 15.1 (Request for a decision), 15.2 (Convening of Holders' Meeting) and 15.4 (Majority, quorum and other provisions).

"Initial Capital Securities" means the Capital Securities issued on the First Issue Date, being in a Nominal Amount of SEK 750,000,000.

- "Insolvent" means, in respect of a relevant person, that it is deemed to be insolvent, or admits inability to pay its debts as they fall due, in each case within the meaning of Chapter 2, Sections 7-9 of the Swedish Bankruptcy Act (konkurslagen (1987:672)) (or its equivalent in any other relevant jurisdiction).
- "Interest" means the interest on the Capital Securities calculated in accordance with Clauses 9.1.1 to 9.1.3.
- "Interest Payment" means, in respect the payment of interest on an Interest Payment Date, the amount of interest payable for the relevant Interest Period in accordance with Clause 9 (Interest).
- "Interest Payment Date" means, subject to Clause 10 (*Optional Interest Deferral*), 19 February, 19 May, 19 August and 19 November of each year or, to the extent such day is not a Business Day, the Business Day following from an application of the Business Day Convention. The first Interest Payment Date for the Capital Securities shall be 19 February 2026 and the last Interest Payment Date shall be the relevant Redemption Date.
- "Interest Period" means (i) in respect of the first Interest Period, the period from (but excluding) the First Issue Date to (and including) the first Interest Payment Date or, if the Capital Securities are redeemed before, the Redemption Date, and (ii) in respect of subsequent Interest Periods, the period from (but excluding) an Interest Payment Date to (and including) the next succeeding Interest Payment Date or, if the Capital Securities are redeemed before, the Redemption Date.
- "**Interest Rate**" means the Base Rate plus the applicable Margin, as adjusted by any application of Clause 17 (*Replacement of Base Rate*). For the avoidance of doubt, if any such total rate is below zero then the Interest Rate will be deemed to be zero (0).
- "Issuer" means Swedavia AB (publ), a public limited liability company incorporated under the laws of Sweden with Reg. No. 556797-0818.
- "Issuer Winding-up" has the meaning set forth in Clause 3.2.
- "Issuer Re-Construction" has the meaning set forth in Clause 3.2.
- "Issuing Agent" means, initially, Nordea Bank Abp, filial i Sverige, Reg. No. 516411-1683, and thereafter each other party appointed as Issuing Agent in accordance with these Terms and Conditions and the CSD Regulations.
- "Margin" means:
- in respect of the period from (but excluding) the First Issue Date to (and including) the Margin Step-up Date, 2.20 per cent. *per annum*; and
- (b) in respect of the period from (but excluding) the Margin Step-up Date, 3.20 per cent. *per annum*.
- "Margin Step-up Date" means the Interest Payment Date falling in November 2035.
- "Net Proceeds" means the gross proceeds from the offering of the relevant Capital Securities, minus the costs incurred by the Issuer in conjunction with the issuance and listing thereof.
- "Nominal Amount" has the meaning set forth in Clause 2.3.
- "Parity Securities" means any obligations of:
- (a) the Issuer, issued directly or indirectly by it, which rank, or are expressed to rank, *pari passu* with the Capital Securities; and
- (b) any Subsidiary of the Issuer having the benefit of a guarantee or support agreement from the Issuer which ranks or is expressed to rank *pari passu* with the Capital Securities.
- "Qualifying Securities" means securities issued directly by the Issuer following a substitution or variation of the Capital Securities in accordance with Clause 11.6 (Substitution or variation due to a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event) that have terms not materially less favourable to investors, certified by the Issuer acting reasonably (having consulted

with an independent investment bank or independent financial adviser), than the terms of the Capital Securities (immediately prior to the relevant substitution or variation), provided that they:

- (a) shall include a ranking at least equal to that of the Capital Securities;
- (b) shall have at least the same Interest Rate and the same Interest Payment Dates as those applying to the Capital Securities;
- (c) shall have the same redemption rights as the Capital Securities;
- (d) shall preserve any existing rights under the Capital Securities to any accrued interest which has not been paid but which has not been cancelled in respect of the period from (and including) the Interest Payment Date last preceding the date of the relevant substitution or variation of the Capital Securities;
- (e) if the Capital Securities were admitted to trading on a Regulated Market immediately prior to the relevant substitution or variation, the Issuer shall use reasonable efforts to ensure that the Capital Securities are admitted to trading on a Regulated Market within thirty (30) days from their issuance (noting that no investor in the relevant Qualifying Securities (or its representative) has the right to accelerate the relevant Qualifying Securities or otherwise request a prepayment or redemption of the relevant Qualifying Securities upon a failure to admit the relevant Qualifying Securities to trading).
- "Quotation Day" means, in relation to any period for which an interest rate is to be determined, two (2) Business Days before the first day of that period.
- "Rating Agency" means Moody's and any other rating agency of equivalent international standing requested by the Issuer to grant a corporate credit rating to the Issuer and, in each case, their successors or affiliates.
- "Rating Methodology Event" shall be deemed to occur if the Issuer has received confirmation from any Rating Agency either directly or via a publication by such Rating Agency, that due to any amendment to, clarification of or change in hybrid capital methodology or a change in the interpretation thereof, in each case occurring or becoming effective after the Issue Date:
- all or any of the Capital Securities will no longer be eligible for the same or a higher amount of "equity credit" (or such other nomenclature that the relevant Rating Agency may then use to describe the degree to which an instrument exhibits the characteristics of an ordinary share) attributed to the Capital Securities at the Issue Date (or, if "equity credit" is not assigned to the Capital Securities by the relevant Rating Agency on the Issue Date, at the date on which "equity credit" is assigned by such Rating Agency for the first time); or
- (b) the length of time the Capital Securities are assigned a particular level of "equity credit" by the relevant Rating Agency is shortened as compared to the length of time that the Capital Securities were assigned that level of "equity credit" by the relevant Rating Agency under its prevailing methodology on the Issue Date.

"Record Date" means the fifth (5) Business Day prior to:

- (a) an Interest Payment Date;
- (b) a Redemption Date; or
- (c) another relevant date,

or in each case such other Business Day falling prior to a relevant date if generally applicable on the Swedish bond market.

"Redemption Date" means the date on which the relevant Capital Securities are to be redeemed or repurchased in accordance with Clause 11 (*Redemption and repurchase of the Capital Securities*).

- "Regulated Market" means any regulated market (as defined in Directive 2014/65/EU on markets in financial instruments).
- "Securities Account" means the account for dematerialised securities (avstämningsregister) maintained by the CSD pursuant to the Central Securities Depositories and Financial Instruments Accounts Act in which:
- (a) an owner of such security is directly registered; or
- (b) an owner's holding of securities is registered in the name of a nominee.
- "Security" means a mortgage, charge, pledge, lien, security assignment or other security interest securing any obligation of any person, or any other agreement or arrangement having a similar effect.
- "**Special Event**" means any of an Accounting Event, a Substantial Repurchase Event, a Tax Deductibility Event, a Rating Methodology Event. a Withholding Tax Event or any combination of the foregoing.

"STIBOR" means:

- (a) the Stockholm interbank offered rate (STIBOR) administered by the Swedish Financial Benchmark Facility AB (or any person replacing it as administrator) for Swedish Kronor and for a period equal to the relevant Interest Period, as published by Swedish Financial Benchmark Facility AB (or any other person which takes over the publication of that rate) as of or around 11.00 a.m. on the Quotation Day;
- (b) if no rate as described in paragraph (a) above is available for the relevant Interest Period, the rate determined by the Issuing Agent by linear interpolation between the two closest rates for STIBOR fixing, as published by Swedish Financial Benchmark Facility AB (or any other person which takes over the publication of that rate), as of or around 11.00 a.m. on the Quotation Day for Swedish Kronor;
- (c) if no rate as described in paragraph (a) or (b) above is available for the relevant Interest Period, the arithmetic mean of the Stockholm interbank offered rates (rounded upwards to four decimal places) as supplied to the Issuing Agent at its request quoted by leading banks in the Stockholm interbank market reasonably selected by the Issuing Agent for deposits of SEK 100,000,000 for the relevant period; or
- (d) if no rate as described in paragraph (a) or (b) above is available for the relevant Interest Period and no quotation is available pursuant to paragraph (c) above, the interest rate which according to the reasonable assessment of the Issuing Agent best reflects the interest rate for deposits in Swedish Kronor offered in the Stockholm interbank market for the relevant period.
- "Subordinated Indebtedness" means any obligation of the Issuer, whether or not having a fixed maturity, which by its terms is, or is expressed to be, subordinated in the event of an Issuer Windingup to the claims of all unsubordinated creditors of the Issuer but senior to the Capital Securities or to the obligations of the Issuer in respect of any Parity Securities.
- "Subsequent Capital Securities" means any Capital Securities issued after the First Issue Date on one or more occasions.
- "Subsidiary" means, in relation to any person, any Swedish or foreign legal entity (whether incorporated or not), which at the time is a subsidiary (*dotterföretag*) to such person, directly or indirectly, as defined in the Swedish Companies Act (*aktiebolagslagen* (2005:551)).
- "Substantial Repurchase Event" shall be deemed to occur if the Issuer and/or any of its Subsidiaries repurchases and cancels or has at any time repurchased and cancelled, a principal amount of Capital Securities equal to or greater than eighty (80) per cent. of the aggregate principal amount of the Capital Securities issued (which shall include, for these purposes, any Subsequent Capital Securities).
- "Swedish Kronor" and "SEK" means the lawful currency of Sweden.

"Tax Deductibility Event" means the receipt by the Issuer of an opinion of independent and well-reputed counsel in Sweden (experienced in such matters) to the effect that, as a result of a Tax Law Change, any interest payments under the Capital Securities were, but are no longer, tax-deductible by the Issuer for Swedish tax purposes to the same extent as any unsubordinated obligations of the Issuer.

"Tax Law Change" means:

- (a) any amendment to, clarification of, or change in, the laws or treaties (or any regulations thereunder) of Sweden, or any political subdivision or any authority thereof or therein having the power to tax, affecting taxation;
- (b) any governmental action; or
- (c) any amendment to, clarification of, or change in the application, official position or the interpretation of such law, treaty (or regulations thereunder) or governmental action or any interpretation, decision or pronouncement that provides for a position with respect to such law, treaty (or regulations thereunder) or governmental action that differs from the theretofore generally accepted position,

in each case, by any legislative body, court, governmental authority or regulatory body in Sweden, irrespective of the manner in which such amendment, clarification, change, action, pronouncement, interpretation or decision is made known, which amendment, clarification or change is effective or such governmental action, pronouncement, interpretation or decision is announced on or after the date falling five (5) Business Days prior to the First Issue Date.

"Total Nominal Amount" means the total aggregate Nominal Amount of the Capital Securities outstanding at the relevant time.

"Withholding Tax Event" shall be deemed to occur if, as a result of any Tax Law Change, in making any payments on the Capital Securities, the Issuer has paid or will or would on the next Interest Payment Date be required to pay additional amounts on the Capital Securities and the Issuer cannot avoid the foregoing by taking reasonable measures available to it as evidenced by an opinion of an independent and well-reputed counsel in Sweden (experienced in such matters) to that effect.

"Written Procedure" means the written or electronic procedure for decision making among the Holders in accordance with Clauses 15.1 (Request for a decision), 15.3 (Instigation of Written Procedure) and 15.4 (Majority, quorum and other provisions).

1.2 Construction

- 1.2.1 Unless a contrary indication appears, any reference in these Terms and Conditions to:
 - (a) "assets" includes present and future properties, revenues and rights of every description;
 - (b) any agreement or instrument is a reference to that agreement or instrument as supplemented, amended, novated, extended, restated or replaced from time to time;
 - (c) a "**regulation**" includes any law, regulation, rule or official directive (whether or not having the force of law) of any governmental, intergovernmental or supranational body, agency or department;
 - (d) a provision of regulation is a reference to that provision as amended or re-enacted; and
 - (e) a time of day is a reference to Stockholm time.
- 1.2.2 When ascertaining whether a limit or threshold specified in Swedish Kronor has been attained or broken, an amount in another currency shall be counted on the basis of the rate of exchange for such currency against Swedish Kronor for the previous Business Day, as published by the Swedish Central Bank (*Riksbanken*) on its website (www.riksbank.se). If no such rate is available, the most recently published rate shall be used instead.

- 1.2.3 A notice shall be deemed to be sent by way of press release if it is made available to the public within Sweden promptly and in a non-discriminatory manner.
- 1.2.4 No delay or omission of the Agent or of any Holder to exercise any right or remedy under the Terms and Conditions shall impair or operate as a waiver of any such right or remedy.
- 1.2.5 The selling restrictions, the privacy notice and any other information contained in this document before the table of contents section do not form part of these Terms and Conditions and may be updated without the consent of the Holders and the Agent.

2. THE CAPITAL SECURITIES

- 2.1 The Capital Securities are denominated in Swedish Kronor and each Capital Security is constituted by these Terms and Conditions. The Issuer undertakes to make payments in relation to the Capital Securities (subject to and in accordance with these Terms and Conditions) and to comply with these Terms and Conditions.
- 2.2 By subscribing for Capital Securities, each initial Holder agrees that the Capital Securities shall benefit from and be subject to the Terms and Conditions and by acquiring Capital Securities, each subsequent Holder confirms such agreement.
- 2.3 The initial nominal amount of each Capital Security is SEK 1,250,000 (the "Nominal Amount"). The maximum Nominal Amount of the Initial Capital Securities as at the First Issue Date is SEK 750,000,000. All Initial Capital Securities are issued on a fully paid basis at an issue price of one hundred (100) per cent. of the Initial Nominal Amount.
- 2.4 The ISIN for the Capital Securities is SE0026853319
- 2.5 Provided that the relevant conditions precedent have been duly received (or waived) by the Agent in accordance with Clause 5.1.2, the Issuer may, on one or several occasions, issue Subsequent Capital Securities. Subsequent Capital Securities shall benefit from and be subject to the Terms and Conditions, and, for the avoidance of doubt, the ISIN, the interest rate, the currency, the Nominal Amount and the perpetual nature applicable to the Initial Capital Securities shall apply to Subsequent Capital Securities. The issue price of the Subsequent Capital Securities may be set at the Nominal Amount, a discount or a premium compared to the Nominal Amount. The maximum Total Nominal Amount of the Capital Securities (the Initial Capital Securities and all Subsequent Capital Securities) may not exceed SEK 3,000,000,000 unless a consent from the Holders is obtained in accordance with paragraph (c) of Clause 15.4.2. Each Subsequent Capital Security shall entitle its holder to Interest in accordance with Clause 9.1, and otherwise have the same rights as the Initial Capital Securities.
- 2.6 The Capital Securities are freely transferable but the Holders may be subject to purchase or transfer restrictions with regard to the Capital Securities, as applicable, under local regulation to which a Holder may be subject. Each Holder must ensure compliance with such restrictions at its own cost and expense.

3. STATUS OF THE CAPITAL SECURITIES

- 3.1 The Capital Securities, including the obligation to pay interest thereon, constitute direct, general, unconditional, unsecured and subordinated obligations of the Issuer. The rights and claims of the Holders in respect of the Capital Securities against the Issuer, are subordinated as described in Clause 3.2 below.
- 3.2 In the event of:
 - (a) a voluntary or involuntary liquidation (*likvidation*) or bankruptcy (*konkurs*) of the Issuer (each an "**Issuer Winding-up**"), the Holders (or the Agent on their behalf) shall, in respect

of their Capital Securities, have a claim for the principal amount of their Capital Securities and any accrued and unpaid interest (including any Deferred Interest) thereon and such claims will rank:

- (i) pari passu without any preference among themselves and with any present or future claims in respect of obligations of the Issuer in respect of Parity Securities;
- (ii) in priority to all present and future claims in respect of:
 - (A) the ordinary shares of the Issuer; and
 - (B) any other obligation of the Issuer expressed by its terms as at its original issue date to rank junior to the Capital Securities or any Parity Securities; and
- (iii) junior in right of payment to any present or future claims of:
 - (A) all unsubordinated obligations of the Issuer; and
 - (B) all Subordinated Indebtedness; or
- (b) a company re-construction (*företagsrekonstruktion*) of the Issuer under the Swedish Company Reorganisation Act (*lag* (2022:964) om *företagsrekonstruktion*) (an "**Issuer Reconstruction**"), the Holders (or the Agent on their behalf) shall, in respect of their Capital Securities, have a claim for the principal amount of their Capital Securities and any accrued and unpaid interest (including any Deferred Interest) thereon and such claims will rank:
 - (i) pari passu without any preference among themselves and with any present or future claims in respect of obligations of the Issuer in respect of Parity Securities; and
 - (ii) junior in right of payment to any present or future claims of:
 - (A) all unsubordinated obligations of the Issuer; and
 - (B) all Subordinated Indebtedness.

Claims in respect of the share capital of the Issuer are not subject to loss absorbing measures under an Issuer Re-construction.

3.3 Subject to applicable law, no Holder may exercise, claim or plead any right of set-off, compensation or retention in respect of any amount owed to it by the Issuer in respect of, or arising under or in connection with the Capital Securities and each Holder shall, by virtue of its holding of any Capital Security, be deemed to have waived all such rights of set-off, compensation or retention.

4. USE OF PROCEEDS

- 4.1 The Issuer shall use the Net Proceeds from the issue of the Initial Capital Securities for its general corporate purposes, including refinancing financial indebtedness and financing acquisitions and investments.
- 4.2 The Issuer shall use the Net Proceeds from the issue of any Subsequent Capital Securities for its general corporate purposes, including financing acquisitions and investments.

5. CONDITIONS PRECEDENT

5.1 Conditions precedent for disbursement

- 5.1.1 The Issuer shall provide to the Agent, prior to the First Issue Date (or such later time as agreed by the Agent), each document and other evidence listed in Part I of <u>Schedule 1</u> (*Conditions Precedent*) in form and substance satisfactory to the Agent.
- 5.1.2 The Issuer shall provide to the Agent, no later than 9.00 a.m. three (3) Business Days prior to the issue date (or such later time as agreed by the Agent) in respect of Subsequent Capital Securities, each document and other evidence set out in Part II of Schedule 1 (Conditions Precedent) in form and substance satisfactory to the Agent.
- 5.1.3 The Agent shall confirm to the Issuing Agent when it is satisfied that the conditions in Clause 5.1.1 or 5.1.2, as the case may be, have been fulfilled (or amended or waived in accordance with Clause 16 (*Amendments and waivers*)).

5.2 Settlement and disbursement

- 5.2.1 Following receipt by the Issuing Agent of the confirmation in accordance with Clause 5.1.2, the Issuing Agent shall settle the issuance of the Initial Capital Securities and pay the Net Proceeds to the Issuer on the First Issue Date.
- 5.2.2 Following receipt by the Issuing Agent of the confirmation in accordance with Clause 5.1.2, the Issuing Agent shall settle the issuance of the Subsequent Capital Securities and pay the Net Proceeds to the Issuer on the relevant issue date.

5.3 No responsibility for documentation

The Agent may assume that the documentation and evidence delivered to it pursuant to this Clause 5 is accurate, legally valid, enforceable, correct, true and complete unless it has actual knowledge to the contrary and the Agent does not have to verify or assess the contents of any such documentation or evidence. The Agent does not have any obligation to review the documentation and evidence set out in this Clause 5 from a legal or commercial perspective of the Holders.

6. CAPITAL SECURITIES IN BOOK-ENTRY FORM

- The Capital Securities will be registered for the Holders on their respective Securities Accounts and no physical Capital Securities will be issued. Accordingly, the Capital Securities will be registered in accordance with the Central Securities Depositories and Financial Instruments Accounts Act. Registration requests relating to the Capital Securities shall be directed to an Account Operator. The Debt Register shall constitute conclusive evidence of the persons who are Holders and their holdings of Capital Securities.
- Those who according to assignment, Security, the provisions of the Swedish Children and Parents Code (*föräldrabalken* (1949:381)), conditions of will or deed of gift or otherwise have acquired a right to receive payments in respect of a Capital Security shall register their entitlements to receive payment in accordance with the Central Securities Depositories and Financial Instruments Accounts Act.
- 6.3 The Issuer and the Agent shall at all times be entitled to obtain information from the Debt Register. At the request of the Agent, the Issuer shall promptly obtain such information and provide it to the Agent. For the purpose of carrying out any administrative procedure that arises out of the Terms and Conditions, the Issuing Agent shall be entitled to obtain information from the Debt Register.
- The Issuer shall issue any necessary power of attorney to such persons employed by the Agent, as notified by the Agent, in order for such individuals to independently obtain information directly from

the Debt Register. The Issuer may not revoke any such power of attorney unless directed by the Agent or unless consent thereto is given by the Holders.

6.5 The Issuer and the Agent may use the information referred to in Clause 6.3 only for the purposes of carrying out their duties and exercising their rights in accordance with the Terms and Conditions and the Agency Agreement and shall not disclose such information to any Holder or third party unless necessary for such purposes.

7. RIGHT TO ACT ON BEHALF OF A HOLDER

- 7.1 If any person other than a Holder (including the owner of a Capital Security, if such person is not the Holder), wishes to exercise any rights under the Terms and Conditions, it must obtain a power of attorney or other authorisation from the Holder or a successive, coherent chain of powers of attorney or authorisations starting with the Holder and authorising such person.
- 7.2 A Holder may issue one or several powers of attorney or other authorisations to third parties to represent it in relation to some or all of the Capital Securities held by it. Any such representative may act independently under the Terms and Conditions in relation to the Capital Securities for which such representative is entitled to represent the Holder.
- 7.3 The Agent shall only have to examine the face of a power of attorney or other authorisation that has been provided to it pursuant to Clause 7.2 and may assume that such document has been duly authorised, is valid, has not been revoked or superseded and that it is in full force and effect, unless otherwise is apparent from its face or the Agent has actual knowledge to the contrary.
- 7.4 These Terms and Conditions shall not affect the relationship between a Holder who is the nominee (*förvaltare*) with respect to a Capital Security and the owner of such Capital Security, and it is the responsibility of such nominee to observe and comply with any restrictions that may apply to it in this capacity.

8. PAYMENTS IN RESPECT OF THE CAPITAL SECURITIES

- 8.1 Any payment or repayment under the Terms and Conditions, or any amount due in respect of a repurchase of any Capital Securities, shall be made to such person who is registered as a Holder on the Record Date prior to an Interest Payment Date or other relevant payment date, or to such other person who is registered with the CSD on such Record Date as being entitled to receive the relevant payment, repayment or repurchase amount.
- 8.2 Provided that a Holder has registered an income account (avkastningskonto) for the relevant Securities Account on the applicable Record Date, the CSD shall procure that principal, interest and other payments under the Capital Securities are deposited to such income account on the relevant payment date. If an income account has not been registered on the Record Date for the payment, no payment will be effected by the CSD to such Holder. The outstanding amount will instead be held by the Issuer until the person that was registered as a Holder on the relevant Record Date has made a valid request for such amount. Should the CSD, due to a delay on behalf of the Issuer or some other obstacle, not be able to effect payments as aforesaid, the Issuer shall procure that such amounts are paid to the persons who are registered as Holders on the relevant Record Date as soon as possible after such obstacle has been removed.
- 8.3 If, due to any obstacle for the CSD, the Issuer cannot make a payment or repayment, such payment or repayment may be postponed until the obstacle has been removed.
- 8.4 If payment or repayment is made in accordance with this Clause 8, the Issuer shall be deemed to have fulfilled its obligation to pay, irrespective of whether such payment was made to a person not entitled to receive such amount (unless the Issuer has actual knowledge of the fact that the payment was made to the wrong person).

8.5 The Issuer is not liable to gross-up any payments under the Terms and Conditions by virtue of any withholding tax, public levy or the similar.

9. INTEREST

9.1 **Interest**

- 9.1.1 Each Initial Capital Security carries Interest at the applicable Interest Rate applied to the Nominal Amount from (but excluding) the First Issue Date up to (and including) the relevant Redemption Date. Any Subsequent Capital Security will carry Interest at the Interest Rate applied to the Nominal Amount from (but excluding) the Interest Payment Date falling immediately prior to its issuance (or the First Issue Date if there is no such Interest Payment Date) up to (and including) the relevant Redemption Date.
- 9.1.2 Interest accrues during an Interest Period. Subject to Clause 10 (*Optional Interest Deferral*), payment of Interest in respect of the Capital Securities shall be made to the Holders on each Interest Payment Date for the preceding Interest Period.
- 9.1.3 Interest shall be calculated on the basis of the actual number of days in the Interest Period in respect of which payment is being made divided by 360 (actual/360-days basis).
- 9.1.4 Payment of Interest in respect of each Interest Period may be deferred in accordance with Clause 10 (*Optional Interest Deferral*).

9.2 Step-up after a Change of Control Event

Notwithstanding any other provision of this Clause 9, if the Issuer does not elect to redeem the Capital Securities in accordance with Clause 11.5 (*Voluntary redemption due to Change of Control Event*) following the occurrence of a Change of Control Event, the then prevailing Interest Rate, and each subsequent Interest Rate otherwise determined in accordance with the provisions of this Clause 9, on the Capital Securities shall be increased by 500 basis points *per annum* with effect from (but excluding) the Change of Control Step-up Date up to (and including) the relevant Redemption Date.

9.3 **Default interest**

If the Issuer fails to pay any amount payable by it pursuant to Clause 10.3 (Mandatory settlement of Deferred Interest) or Clause 11 (Redemption and repurchase of the Capital Securities) (except for Clause 11.1 (No maturity), Clause 11.2 (Purchase of Capital Securities by Group Companies) and Clause 11.8 (Cancellation of Capital Securities)) on its due date, default interest shall accrue on the overdue amount from (but excluding) the due date up to (and including) the date of actual payment at a rate which is 200 basis points per annum higher than the Interest Rate. Accrued default interest shall not be capitalised. No default interest shall accrue where the failure to pay was solely attributable to the Agent or the CSD, in which case the Interest Rate shall apply instead.

10. OPTIONAL INTEREST DEFERRAL

10.1 **Deferral of Interest Payments**

The Issuer may, at any time and at its sole discretion, elect to defer any Interest Payment, in whole or in part, which is otherwise scheduled to be paid on an Interest Payment Date (except on any Interest Payment Date on which the Capital Securities are to be redeemed) by giving notice of such election to the Holders, the Issuing Agent and the Agent in accordance with Clause 23 (*Notices*) not less than seven (7) Business Days prior to the relevant Interest Payment Date. Notwithstanding the foregoing, failure to give such notice shall not prejudice the right of the Issuer to defer any Interest Payment as described above.

- 10.1.2 If the Issuer makes only a partial payment of interest on an Interest Payment Date, such amount shall be applied equally to each Capital Security.
- Any Interest Payment so deferred pursuant to this Clause 10 shall, from (but excluding) the Interest Payment Date on which such Interest Payment would (but for its deferral) have been payable to (and including) the date on which it is paid in full, itself bear interest at the Interest Rate prevailing from time to time (which interest shall compound on each Interest Payment Date) and, for so long as the same remains unpaid, such deferred interest (together with the interest accrued thereon) shall constitute "**Deferred Interest**".
- 10.1.4 The deferral of an Interest Payment in accordance with this Clause 10 shall not constitute a default pursuant to Clause 14 (*Default and Enforcement*) by the Issuer under the Capital Securities or for any other purpose.

10.2 **Optional settlement of Deferred Interest**

Deferred Interest may be paid, in whole or in part, at any time at the option of the Issuer following delivery of a notice to such effect given by the Issuer to the Holders, the Issuing Agent and the Agent in accordance with Clause 23 (*Notices*) not less than seven (7) Business Days prior to the date (to be specified in such notice) on which the Issuer will pay such Deferred Interest. Any such notice shall state the date fixed for the payment and the relevant Record Date and be irrevocable.

10.3 Mandatory settlement of Deferred Interest

- 10.3.1 The Issuer shall pay any Deferred Interest, in whole but not in part, on the first to occur of the following dates:
 - (a) the tenth (10th) Business Day following the date on which a Deferred Interest Payment Event occurs;
 - (b) any Interest Payment Date in respect of which the Issuer does not elect to defer all of the interest accrued in respect of the relevant Interest Period; and
 - (c) the date on which the Capital Securities are redeemed or repaid in accordance with Clause 11 (Redemption and repurchase of the Capital Securities) or Clause 14 (Default and Enforcement).
- 10.3.2 Notice of any Deferred Interest Payment Event shall be given by the Issuer to the Holders, the Issuing Agent and the Agent in accordance with Clause 23 (*Notices*) within three (3) Business Days of such event.

11. REDEMPTION AND REPURCHASE OF THE CAPITAL SECURITIES

11.1 No maturity

The Capital Securities are perpetual and have no specified maturity date. The Issuer may only redeem the Capital Securities in the circumstances described in this Clause 11 (*Redemption and repurchase of the Capital Securities*). The Capital Securities are not redeemable at the option of the Holders at any time.

11.2 Purchase of Capital Securities by Group Companies

- 11.2.1 The Issuer or any other Group Company may, subject to applicable regulations, at any time and at any price purchase Capital Securities on the market or in any other way.
- Capital Securities held by a Group Company may at such Group Company's discretion be retained or sold but, if held by the Issuer, not cancelled except in connection with a Substantial Repurchase Event.

11.3 Voluntary total redemption (call option)

The Issuer may redeem all, but not some only, of the outstanding Capital Securities in full on the First Call Date or on any Interest Payment Date thereafter at an amount per Capital Security equal to one hundred (100) per cent. of the Nominal Amount together with any Deferred Interest and any other accrued and unpaid interest up to (and including) the Redemption Date.

11.4 Voluntary redemption due to a Special Event

Upon the occurrence of a Special Event which is continuing, the Issuer may, subject to Clause 12 (*Preconditions to Special Event Redemption or Change of Control Event Redemption*), redeem all, but not some only, of the Capital Securities at an amount equal to:

- (a) one hundred and one (101) per cent. of their principal amount, where such redemption occurs before the First Call Date; or
- (b) one hundred (100) per cent. of their principal amount, where such redemption occurs on or after the First Call Date.

together, in each case, with any Deferred Interest and any other accrued and unpaid interest up to (and including) the Redemption Date.

11.5 Voluntary redemption due to a Change of Control Event

- 11.5.1 Upon the occurrence of a Change of Control Event, the Issuer may, no later than the Change of Control Step-up Date, subject to Clause 12 (*Preconditions to Special Event Redemption or Change of Control Event Redemption*), redeem all, but not some only, of the Capital Securities at an amount equal to:
 - one hundred and one (101) per cent. of their principal amount, where such redemption occurs before the First Call Date; or
 - (b) one hundred (100) per cent. of their principal amount, where such redemption occurs on or after the First Call Date,

together, in each case, with any Deferred Interest and any other accrued and unpaid interest up to (and including) the Redemption Date.

11.5.2 Immediately upon the Issuer becoming aware that a Change of Control Event has occurred, the Issuer shall give notice to the Agent, the Issuing Agent and the Holders in accordance with Clause 23 (*Notices*) specifying the nature of the Change of Control Event.

Substitution or variation due to a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event

If a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event occurs, the Issuer may, at its option but subject to giving notice in accordance with Clause 11.7 (*Notice of redemption, substitution or variation*), substitute or vary the terms of all (but not some only) of the outstanding Capital Securities without any requirement for the consent or approval of the Holders, so that they become or remain, as applicable, Qualifying Securities, provided that such substitution or variation does not itself give rise to any right of the Issuer to redeem, substitute or vary the terms of the Capital Securities in accordance with this Clause 11.6 in relation to the Qualifying Securities so substituted or varied.

11.7 Notice of redemption, substitution or variation

Any redemption, substitution or variation in accordance with Clauses 11.3, 11.4, 11.5.1 or 11.6 shall be made by the Issuer giving not less than thirty (30) Business Days' notice and not more than sixty (60) Business Days' notice to the Holders, the Issuing Agent and the Agent, in each case calculated from the effective date of the notice. The Notice from the Issuer shall specify the Redemption Date and also the Record Date on which a person shall be registered as a Holder to receive the amounts

due on such Redemption Date. The notice is irrevocable but may in the case of a redemption in accordance with Clause 11.3, at the Issuer's discretion, contain one or more conditions precedent that shall be satisfied prior to the Record Date. Upon fulfilment of the conditions precedent (if any), the Issuer shall redeem the Capital Securities in full at the applicable amount on the specified Redemption Date.

11.8 Cancellation of Capital Securities

All Capital Securities which are redeemed pursuant to this Clause 11 and all Capital Securities purchased and elected to be cancelled pursuant to Clause 11.2 (*Purchase of Capital Securities by Group Companies*) will be cancelled and may not be reissued or resold. The Issuer shall promptly inform the Holders, the Agent and the Issuing Agent in accordance with Clause 23 (*Notices*) of any such cancellation and for so long as the Capital Securities are admitted to trading on Nasdaq Stockholm or another Regulated Market, as the case may be, and the rules of such exchange so require, the Issuer shall promptly inform Nasdaq Stockholm or the Regulated Market, as the case may be, of the cancellation of any Capital Securities under this Clause 11.

12. PRECONDITIONS TO SPECIAL EVENT REDEMPTION OR CHANGE OF CONTROL EVENT REDEMPTION

- Prior to the publication of any notice of redemption pursuant to Clause 11 (*Redemption and repurchase of the Capital Securities*) (other than redemption pursuant to Clause 11.3 (*Voluntary total redemption (call option)*)), the Issuer shall deliver to the Agent a certificate signed by two authorised signatories of the Issuer stating that the relevant requirement or circumstance giving rise to the right to redeem the Capital Securities is satisfied.
- In addition, in the case of an Accounting Event, a Tax Deductibility Event or a Withholding Tax Event, the Issuer shall deliver to the Agent and the Issuing Agent the relevant opinion of the independent legal, accounting or other tax advisers (as applicable) to the effect that the relevant requirement or circumstance giving rise to the right to redeem is satisfied (save, in the case of a Withholding Tax Event, as to whether reasonable measures to avoid paying additional amounts are available to the Issuer). Such certificate and, if applicable, opinion, shall be conclusive and binding on the Holders provided that the opinion complies with the opinion requirements set out in this Clause 12.2.
- Any redemption of the Capital Securities in accordance with Clause 11 (*Redemption and repurchase of the Capital Securities*) shall be conditional on all Deferred Interest being paid in full in accordance with the provisions of Clause 10.3 (*Mandatory settlement of Deferred Interest*) on or prior to the date of such redemption.

13. ADMISSION TO TRADING

The Issuer has the intention and shall use its best efforts (without assuming any legal or contractual obligation) to ensure that:

- (a) the Initial Capital Securities are admitted to trading on the corporate bond list of Nasdaq Stockholm or, if such admission to trading is not possible to obtain or maintain, admitted to trading on another Regulated Market within thirty (30) days after the First Issue Date;
- (b) any Subsequent Capital Securities are admitted to trading on the corporate bond list of Nasdaq Stockholm or, if such admission to trading is not possible to obtain or maintain, admitted to trading on another Regulated Market within thirty (30) days after relevant issue date of such Subsequent Capital Securities, unless the relevant Subsequent Capital Securities are issued before the expiry of the thirty (30) day period referred to in paragraph (a) above, in which case such Subsequent Capital Securities shall be admitted to trading within thirty (30) days after the First Issue Date together with the Initial Capital Securities; and

(c) the Capital Securities, once admitted to trading on the relevant Regulated Market, continue being listed thereon (however, taking into account the rules and regulations of the relevant Regulated Market and the CSD (as amended from time to time) preventing trading in the Capital Securities in close connection to the redemption of the Capital Securities).

14. DEFAULT AND ENFORCEMENT

14.1 **Proceedings**

- 14.1.1 Without prejudice to the Issuer's right to defer the payment of interest under Clause 10 (*Optional Interest Deferral*), if a default is made by the Issuer for a period of thirty (30) days or more in relation to the payment of any interest, principal or premium in respect of the Capital Securities which is due and payable (a "**Default**"), then the Issuer shall be deemed to be in default under the Capital Securities and the Agent (acting on instructions of the Holders in accordance with these Terms and Conditions) or (subject to Clause 20 (*No direct action by Holders*)) any Holder may institute proceedings for an Issuer Winding-up provided that such Default is still continuing.
- 14.1.2 In the event of an Issuer Winding-up, a Holder may, provided such Holder does not contravene a previously adopted resolution in accordance with Clause 15.4.2 or 15.4.3 (if any), either independently or through the Agent prove and/or claim in such Issuer Winding-up in respect of the Capital Securities, such claim being for such amount, and being subordinated in such manner, as is provided under Clause 3.2.

14.2 **Enforcement**

The Agent (acting on the instructions of the Holders in accordance with these Terms and Conditions) may institute such proceedings against the Issuer as it may think fit to enforce any term or condition binding on the Issuer under the Capital Securities but in no event shall the Issuer, by virtue of the institution of any such proceedings, be obliged to pay any sum or sums sooner than the same would otherwise have been payable by it.

14.3 Extent of Holders' Remedy

No remedy against the Issuer, other than as referred to in this Clause 14, shall be available to the Agent and the Holders in relation to the Capital Securities, whether for the recovery of amounts owing in respect of the Capital Securities or in respect of any breach by the Issuer of any of its other obligations under or in respect of the Capital Securities.

15. DECISIONS BY HOLDERS

15.1 Request for a decision

- 15.1.1 A request by the Agent for a decision by the Holders on a matter relating to the Terms and Conditions shall (at the option of the Agent) be dealt with at a Holders' Meeting or by way of a Written Procedure.
- Any request from the Issuer or a Holder (or Holders) representing at least ten (10) per cent. of the Adjusted Nominal Amount (such request shall, if made by several Holders, be made by them jointly) for a decision by the Holders on a matter relating to the Terms and Conditions shall be directed to the Agent and dealt with at a Holders' Meeting or by way a Written Procedure, as determined by the Agent. The person requesting the decision may suggest the form for decision making, but if it is in the Agent's opinion more appropriate that a matter is dealt with at a Holders' Meeting than by way of a Written Procedure, it shall be dealt with at a Holders' Meeting.
- 15.1.3 The Agent may refrain from convening a Holders' Meeting or instigating a Written Procedure if:
 - (a) the suggested decision must be approved by any person in addition to the Holders and such person has informed the Agent that an approval will not be given; or

- (b) the suggested decision is not in accordance with applicable regulations.
- 15.1.4 The Agent shall not be responsible for the content of a notice for a Holders' Meeting or a communication regarding a Written Procedure unless and to the extent it contains information provided by the Agent.
- 15.1.5 Should the Agent not convene a Holders' Meeting or instigate a Written Procedure in accordance with these Terms and Conditions, without Clause 15.1.3 being applicable, the Issuer or the Holder(s) requesting a decision by the Holders may convene such Holders' Meeting or instigate such Written Procedure, as the case may be, instead. The Issuer or the Issuing Agent shall upon request provide the convening Holder(s) with the information available in the Debt Register in order to convene and hold the Holders' Meeting or instigate and carry out the Written Procedure, as the case may be. The Issuer or Holder(s), as applicable, shall supply to the Agent a copy of the dispatched notice or communication.
- 15.1.6 Should the Issuer want to replace the Agent, it may (i) convene a Holders' Meeting in accordance with Clause 15.2 (*Convening of Holders' Meeting*) or (ii) instigate a Written Procedure by sending communication in accordance with Clause 15.3 (*Instigation of Written Procedure*). After a request from the Holders pursuant to Clause 18.4.3, the Issuer shall no later than ten (10) Business Days after receipt of such request (or such later date as may be necessary for technical or administrative reasons) convene a Holders' Meeting in accordance with Clause 15.2. The Issuer shall inform the Agent before a notice for a Holders' Meeting or communication relating to a Written Procedure where the Agent is proposed to be replaced is sent and supply to the Agent a copy of the dispatched notice or communication.
- 15.1.7 Should the Issuer or any Holder(s) convene a Holders' Meeting or instigate a Written Procedure pursuant to Clause 15.1.5 or 15.1.6, then the Agent shall no later than five (5) Business Days' prior to dispatch of such notice or communication be provided with a draft thereof. The Agent may further append information from it together with the notice or communication, provided that the Agent supplies such information to the Issuer or the Holder(s), as the case may be, no later than one (1) Business Day prior to the dispatch of such notice or communication.

15.2 Convening of Holders' Meeting

- 15.2.1 The Agent shall convene a Holders' Meeting by way of notice to the Holders as soon as practicable and in any event no later than five (5) Business Days after receipt of a complete notice from the Issuer or the Holder(s) (or such later date as may be necessary for technical or administrative reasons).
- 15.2.2 The notice pursuant to Clause 15.2.1 shall include:
 - (a) time for the meeting;
 - (b) place for the meeting;
 - (c) a specification of the Record Date on which a person must be registered as a Bondholder in order to be entitled to exercise voting rights;
 - (d) a form of power of attorney;
 - (e) the agenda for the meeting;
 - (f) any applicable conditions precedent and conditions subsequent;
 - (g) the reasons for, and contents of, each proposal;
 - (h) if the proposal concerns an amendment to the Terms and Conditions, the details of such proposed amendment;

- if a notification by the Holders is required in order to attend the Holders' Meeting, information regarding such requirement; and
- (j) information on where additional information (if any) will be published.
- 15.2.3 The Holders' Meeting shall be held no earlier than ten (10) Business Days and no later than thirty (30) Business Days after the effective date of the notice.
- 15.2.4 Without amending or varying these Terms and Conditions, the Agent may prescribe such further regulations regarding the convening and holding of a Holders' Meeting as the Agent may deem appropriate. Such regulations may include a possibility for Holders to vote without attending the meeting in person.

15.3 **Instigation of Written Procedure**

- 15.3.1 The Agent shall instigate a Written Procedure by way of sending a communication to the Holders as soon as practicable and in any event no later than five (5) Business Days after receipt of a complete communication from the Issuer or the Holder(s) (or such later date as may be necessary for technical or administrative reasons).
- 15.3.2 A communication pursuant to Clause 15.3.1 shall include:
 - (a) a specification of the Record Date on which a person must be registered as a Holder in order to be entitled to exercise voting rights;
 - (b) instructions and directions on where to receive a form for replying to the request (such form to include an option to vote yes or no for each request) as well as a form of power of attorney;
 - (c) the stipulated time period within which the Holder must reply to the request (such time period to last at least ten (10) Business Days and not longer than thirty (30) Business Days from the effective date of the communication pursuant to Clause 15.3.1);
 - (d) any applicable conditions precedent and conditions subsequent;
 - (e) the reasons for, and contents of, each proposal;
 - (f) if a proposal concerns an amendment to the Terms and Conditions, the details of such proposed amendment;
 - (g) if the voting is to be made electronically, instructions for such voting; and
 - (h) information on where additional information (if any) will be published.
- 15.3.3 If so elected by the person requesting the Written Procedure and provided that it is also disclosed in the communication pursuant to Clause 15.3.1, when consents from Holders representing the requisite majority of the total Adjusted Nominal Amount pursuant to Clauses 15.4.2 and 15.4.3 have been received in a Written Procedure, the relevant decision shall be deemed to be adopted pursuant to Clause 15.4.2 or 15.4.3, as the case may be, even if the time period for replies in the Written Procedure has not yet expired.
- 15.3.4 The Agent may, during the Written Procedure, provide information to the Issuer by way of updates whether or not quorum requirements have been met and about the eligible votes received by the Agent, including the portion consenting or not consenting to the proposal(s) or refraining from voting (as applicable).

15.4 Majority, quorum and other provisions

- Only a Holder, or a person who has been provided with a power of attorney or other authorisation pursuant to Clause 7 (*Right to act on behalf of a Holder*) from a Holder:
 - on the Record Date specified in the notice pursuant to Clause 15.2.2, in respect of a Holders' Meeting, or
 - (b) on the Record Date specified in the communication pursuant to Clause 15.3.2, in respect of a Written Procedure,

may exercise voting rights as a Holder at such Holders' Meeting or in such Written Procedure, provided that the relevant Capital Securities are included in the Adjusted Nominal Amount. Each whole Capital Security entitles to one vote and any fraction of a Capital Security voted for by a person shall be disregarded. Such Business Day specified pursuant to paragraph (a) or (b) above must fall no earlier than one (1) Business Day after the effective date of the notice or communication, as the case may be.

- The following matters shall require the consent of Holders representing at least sixty-six and two thirds (66%) per cent. of the Adjusted Nominal Amount for which Holders are voting at a Holders' Meeting or for which Holders reply in a Written Procedure in accordance with the instructions given pursuant to Clause 15.3.2:
 - (a) a change of Issuer;
 - (b) a change to the currency, denomination, status or transferability of the Capital Securities;
 - (c) the issue of any Subsequent Capital Securities, if the Total Nominal Amount of the Capital Securities exceeds, of if such issue would cause the Total Nominal Amount to at any time exceed, SEK 3,000,000,000 (for the avoidance of doubt, for which consent shall be required at each occasion such Subsequent Capital Securities are issued);
 - (d) a change to the Interest Rate (including, for the avoidance of doubt, changes to the Margin or the Base Rate (other than as a result of an application of Clause 17 (*Replacement of Base Rate*)) or the Nominal Amount;
 - (e) any delay of the due date for payment of any interest on the Capital Securities other than as permitted pursuant to Clause 10 (*Optional Interest Deferral*);
 - (f) a reduction of the premium payable upon the redemption or repurchase of any Capital Security pursuant to Clause 11 (*Redemption and repurchase of the Capital Securities*);
 - (g) a change to the terms dealing with the requirements for Holders' consent set out in this Clause 15.4 (Majority, quorum and other provisions);
 - (h) an extension of the tenor of the Capital Securities or any delay of the due date for payment of any principal or interest on the Capital Securities;
 - (i) a mandatory exchange of the Capital Securities for other securities; and
 - early redemption of the Capital Securities, other than as otherwise permitted or required by these Terms and Conditions.
- 15.4.3 Any matter not covered by Clause 15.4.2, including for the avoidance of doubt the initiation of an Issuer Winding-up, shall require the consent of Holders representing more than fifty (50) per cent. of the Adjusted Nominal Amount for which Holders are voting at a Holders' Meeting or for which Holders reply in a Written Procedure in accordance with the instructions given pursuant to Clause 15.3.2. This includes, but is not limited to, any amendment to, or waiver of, the terms of the

Terms and Conditions that does not require a higher majority (other than an amendment permitted pursuant to Clause 16.1(a) or (c)).

- 15.4.4 Quorum at a Holders' Meeting or in respect of a Written Procedure only exists if a Holder (or Holders) representing at least fifty (50) per cent. of the Adjusted Nominal Amount in case of a matter pursuant to Clause 15.4.2, and otherwise twenty (20) per cent. of the Adjusted Nominal Amount:
 - (a) if at a Holders' Meeting, attend the meeting in person or by other means prescribed by the Agent pursuant to Clause 15.2.4 (or appear through duly authorised representatives); or
 - (b) if in respect of a Written Procedure, reply to the request.
- 15.4.5 If a quorum exists for some but not all of the matters to be dealt with at a Holders' Meeting or by a Written Procedure, decisions may be taken in the matters for which a quorum exists.
- 15.4.6 If a quorum does not exist at a Holders' Meeting or in respect of a Written Procedure, the Agent or the Issuer shall convene a second Holders' Meeting (in accordance with Clause 15.2.1) or initiate a second Written Procedure (in accordance with Clause 15.3.1), as the case may be, provided that the person(s) who initiated the procedure for Holders' consent has confirmed that the relevant proposal is not withdrawn. For the purposes of a second Holders' Meeting or second Written Procedure pursuant to this Clause 15.4.6, the date of request of the second Holders' Meeting pursuant to Clause 15.2.1 or second Written Procedure pursuant to Clause 15.3.1, as the case may be, shall be deemed to be the relevant date when the quorum did not exist. The quorum requirement in Clause 15.4.4 shall not apply to such second Holders' Meeting or Written Procedure.
- 15.4.7 Any decision which extends or increases the obligations of the Issuer or the Agent, or limits, reduces or extinguishes the rights or benefits of the Issuer or the Agent, under the Terms and Conditions shall be subject to the Issuer's or the Agent's consent, as applicable.
- 15.4.8 A Holder holding more than one Capital Security need not use all its votes or cast all the votes to which it is entitled in the same way and may in its discretion use or cast some of its votes only.
- The Issuer may not, directly or indirectly, pay or cause to be paid any consideration to or for the benefit of any owner of Capital Securities (irrespective of whether such person is a Holder) for or as inducement to any consent under these Terms and Conditions, unless such consideration is offered to all Holders that vote in respect of the proposal at the relevant Holders' Meeting or in a Written Procedure within the time period stipulated for the consideration to be payable (such time period to be no less than ten (10) Business Days).
- 15.4.10 The Issuer may not, directly or indirectly, pay or cause to be paid any consideration to or for the benefit of any owner of Capital Securities (irrespective of whether such person is a Holder), nor make an offer to repurchase any Capital Securities, if receipt of such consideration or participation in such tender offer (as applicable) is conditional upon the Holder's consent to a proposal at a Holder's Meeting or in a Written Procedure.
- 15.4.11 A matter decided at a duly convened and held Holders' Meeting or by way of Written Procedure is binding on all Holders, irrespective of them being present or represented at the Holders' Meeting or responding in the Written Procedure. The Holders that have not adopted or voted for a decision shall not be liable for any damages that this may cause the Issuer or the other Holders.
- 15.4.12 All costs and expenses incurred by the Issuer or the Agent for the purpose of convening a Holders' Meeting or for the purpose of carrying out a Written Procedure, including reasonable fees to the Agent, shall be paid by the Issuer.
- 15.4.13 If a decision is to be taken by the Holders on a matter relating to the Terms and Conditions, the Issuer shall promptly at the request of the Agent provide the Agent with a certificate specifying the number of Capital Securities owned by Group Companies as per the Record Date for voting, irrespective of whether such person is a Holder. The Agent shall not be responsible for the accuracy of such

certificate or otherwise be responsible for determining whether a Capital Security is owned by a Group Company.

15.4.14 Information about decisions taken at a Holders' Meeting or by way of a Written Procedure shall promptly be published on the websites of the Issuer and the Agent, provided that a failure to do so shall not invalidate any decision made or voting result achieved. The minutes from the relevant Holders' Meeting or Written Procedure shall at the request of a Holder be sent to it by the Issuer or the Agent, as applicable.

16. AMENDMENTS AND WAIVERS

- The Issuer and the Agent (acting on behalf of the Holders) may agree in writing to amend or waive any provision in the Terms and Conditions or any other document relating to the Capital Securities, provided that the Agent is satisfied that such amendment or waiver:
 - (a) is not detrimental to the interest of the Holders as a group;
 - (b) is made solely for the purpose of rectifying obvious errors and mistakes;
 - (c) is required by any applicable regulation, a court ruling or a decision by a relevant authority;
 - (d) is necessary for the purpose of having the Capital Securities admitted to trading on Nasdaq Stockholm (or any other Regulated Market, as applicable), provided that such amendment or waiver does not materially adversely affect the rights of the Holders;
 - (e) is made pursuant to Clause 17 (Replacement of Base Rate); or
 - (f) has been duly approved by the Holders in accordance with Clause 15 (*Decisions by Holders*) and it has received any conditions precedent specified for the effectiveness of the approval by the Holders.
- The Issuer may substitute or vary the terms of all (but not some only) of the outstanding Capital Securities without any requirement for the consent or approval of the Holders, so that they become or remain, as applicable, Qualifying Securities, provided that such substitution or variation does not itself give rise to any right of the Issuer to substitute or vary the terms of the Capital Securities in accordance with Clause 11.6 (Substitution or variation due to a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event) in relation to the Qualifying Securities so substituted or varied.
- 16.3 The consent of the Holders is not necessary to approve the particular form of any amendment or waiver to these Terms and Conditions. It is sufficient if such consent approves the substance of the amendment or waiver.
- The Agent shall promptly notify the Holders of any amendments or waivers made in accordance with Clause 16.1, setting out the date from which the amendment or waiver will be effective, and ensure that any amendments to these Terms and Conditions are available on the websites of the Issuer and the Agent. The Issuer shall ensure that any amendments to these Terms and Conditions are duly registered with the CSD and each other relevant organisation or authority.
- An amendment to the Terms and Conditions shall take effect on the date determined by the Holders' Meeting, in the Written Procedure or by the Agent, as the case may be.

17. REPLACEMENT OF BASE RATE

17.1 General

- 17.1.1 Any determination or election to be made by an Independent Adviser, the Issuer or the Holders in accordance with the provisions of this Clause 17 shall at all times be made by such Independent Adviser, the Issuer or the Holders (as applicable) acting in good faith, in a commercially reasonable manner and by reference to relevant market data.
- 17.1.2 If a Base Rate Event has occurred, this Clause 17 shall take precedent over the fallbacks set out in paragraphs (a) to (d) of the definition of STIBOR.

17.2 **Definitions**

In this Clause 17:

- "Adjustment Spread" means a spread (which may be positive, negative or zero) or a formula or methodology for calculating a spread, or a combination thereof to be applied to a Successor Base Rate and that is:
- (a) formally recommended by any Relevant Nominating Body in relation to the replacement of the Base Rate; or
- (b) if paragraph (a) is not applicable, the adjustment spread that the Independent Adviser determines is reasonable to use in order to eliminate, to the extent possible, any transfer of economic value from one party to another as a result of a replacement of the Base Rate and is customarily applied in comparable debt capital market transactions.

"Base Rate Amendments" has the meaning set forth in Clause 17.3.4.

"Base Rate Event" means one or several of the following circumstances:

- (a) the Base Rate (for the relevant Interest Period) has ceased to exist or ceased to be published for at least five (5) consecutive Business Days as a result of the Base Rate (for the relevant Interest Period) ceasing to be calculated or administered;
- (b) a public statement or publication of information by (i) the supervisor of the Base Rate Administrator or (ii) the Base Rate Administrator that the Base Rate Administrator ceases to provide the applicable Base Rate (for the relevant Interest Period) permanently or indefinitely and, at the time of the statement or publication, no successor administrator has been appointed or is expected to be appointed to continue to provide the Base Rate;
- (c) a public statement or publication of information in each case by the supervisor of the Base Rate Administrator that the Base Rate (for the relevant Interest Period) is no longer representative of the underlying market which the Base Rate is intended to represent and the representativeness of the Base Rate will not be restored in the opinion of the supervisor of the Base Rate Administrator;
- (d) a public statement or publication of information in each case by the supervisor of the Base Rate Administrator with the consequence that it is unlawful for the Issuer or the Issuing Agent to calculate any payments due to be made to any Holder using the applicable Base Rate (for the relevant Interest Period) or it has otherwise become prohibited to use the applicable Base Rate (for the relevant Interest Period);
- (e) a public statement or publication of information in each case by the bankruptcy trustee of the Base Rate Administrator or by the trustee under the bank recovery and resolution framework (*krishanteringsregelverket*) containing the information referred to in paragraph (b) above; or
- (f) a Base Rate Event Announcement has been made and the announced Base Rate Event as set out in paragraphs (b) to (e) above will occur within six (6) months.

- "Base Rate Event Announcement" means a public statement or published information as set out in paragraph (b) to (e) of the definition of Base Rate Event that any event or circumstance specified therein will occur.
- "Independent Adviser" means an independent financial institution or adviser of repute in the debt capital markets where the Base Rate is commonly used.
- "Relevant Nominating Body" means, subject to applicable law, firstly any relevant supervisory authority, secondly any applicable central bank, or any working group or committee of any of them, or thirdly the Financial Stability Council (*Finansiella stabilitetsrådet*) or any part thereof.

"Successor Base Rate" means:

- (a) a screen or benchmark rate, including the methodology for calculating term structure and calculation methods in respect of debt instruments with similar interest rate terms as the Capital Securities, which is formally recommended as a successor to or replacement of the Base Rate by a Relevant Nominating Body; or
- (b) if there is no such rate as described in paragraph (a), such other rate as the Independent Adviser determines is most comparable to the Base Rate.

For the avoidance of doubt, in the event that a Successor Base Rate ceases to exist, this definition shall apply *mutatis mutandis* to such new Successor Base Rate.

17.3 Determination of Base Rate, Adjustment Spread and Base Rate Amendments

- 17.3.1 Without prejudice to Clause 17.3.2, upon a Base Rate Event Announcement, the Issuer may, if it is possible to determine a Successor Base Rate at such point of time, at any time before the occurrence of the relevant Base Rate Event at the Issuer's expense appoint an Independent Adviser to initiate the procedure to determine a Successor Base Rate, the Adjustment Spread and any Base Rate Amendments for purposes of determining, calculating and finally deciding the applicable Base Rate. For the avoidance of doubt, the Issuer will not be obliged to take any such actions until obliged to do so pursuant to Clause 17.3.2.
- 17.3.2 If a Base Rate Event has occurred, the Issuer shall use all commercially reasonable endeavours to, as soon as reasonably practicable and at the Issuer's expense, appoint an Independent Adviser to initiate the procedure to determine, as soon as commercially reasonable, a Successor Base Rate, the Adjustment Spread and any Base Rate Amendments for purposes of determining, calculating, and finally deciding the applicable Base Rate.
- 17.3.3 If the Issuer fails to appoint an Independent Adviser in accordance with Clause 17.3.2, the Holders shall, if so decided at a Holders' Meeting or by way of Written Procedure, be entitled to appoint an Independent Adviser (at the Issuer's expense) for the purposes set forth in Clause 17.3.2. If the Issuer fails to carry out any other actions set forth in Clause 17.3 to 17.6, the Agent (acting on the instructions of the Holders) may to the extent necessary effectuate any Base Rate Amendments without the Issuer's cooperation.
- 17.3.4 The Independent Adviser shall also initiate the procedure to determine any technical, administrative or operational changes required to ensure the proper operation of a Successor Base Rate or to reflect the adoption of such Successor Base Rate in a manner substantially consistent with market practice ("Base Rate Amendments").
- 17.3.5 Provided that a Successor Base Rate, the applicable Adjustment Spread and any Base Rate Amendments have been finally decided no later than prior to the relevant Quotation Day in relation to the next succeeding Interest Period, they shall become effective with effect from and including the commencement of the next succeeding Interest Period, always subject to any technical limitations of the CSD and any calculations methods applicable to such Successor Base Rate.

17.4 **Interim measures**

- 17.4.1 If a Base Rate Event set out in any of the paragraphs (a) to (e) of the Base Rate Event definition has occurred but no Successor Base Rate and Adjustment Spread have been finally decided prior to the relevant Quotation Day in relation to the next succeeding Interest Period or if such Successor Base Rate and Adjustment Spread have been finally decided but due to technical limitations of the CSD, cannot be applied in relation to the relevant Quotation Day, the Interest Rate applicable to the next succeeding Interest Period shall be:
 - (a) if the previous Base Rate is available, determined pursuant to the terms that would apply to the determination of the Base Rate as if no Base Rate Event had occurred; or
 - (b) if the previous Base Rate is no longer available or cannot be used in accordance with applicable law or regulation, equal to the Interest Rate determined for the immediately preceding Interest Period.
- 17.4.2 For the avoidance of doubt, Clause 17.4.1 shall apply only to the relevant next succeeding Interest Period and any subsequent Interest Periods are subject to the subsequent operation of, and to adjustments as provided in, this Clause 17. This will however not limit the application of Clause 17.4.1 for any subsequent Interest Periods, should all relevant actions provided in this Clause 17 have been taken, but without success.

17.5 **Notices etc.**

Prior to the Successor Base Rate, the applicable Adjustment Spread and any Base Rate Amendments become effective the Issuer shall promptly, following the final decision by the Independent Adviser of any Successor Base Rate, Adjustment Spread and any Base Rate Amendments, give notice thereof to the Agent, the Issuing Agent and the Holders in accordance with Clause 23 (*Notices*) and the CSD. The notice shall also include information about the effective date of the amendments. If the Capital Securities are admitted to trading on a stock exchange, the Issuer shall also give notice of the amendments to the relevant stock exchange.

17.6 Variation upon replacement of Base Rate

- 17.6.1 No later than giving the Agent notice pursuant to Clause 17.5, the Issuer shall deliver to the Agent a certificate signed by the Independent Adviser and the CEO, CFO or any other duly authorised signatory of the Issuer (subject to Clause 17.3.3) confirming the relevant Successor Base Rate, the Adjustment Spread and any Base Rate Amendments, in each case as determined and decided in accordance with the provisions of this Clause 17. The Successor Base Rate the Adjustment Spread and any Base Rate Amendments (as applicable) specified in such certificate will, in the absence of manifest error or bad faith in any decision, be binding on the Issuer, the Agent, the Issuing Agent and the Holders.
- 17.6.2 Subject to receipt by the Agent of the certificate referred to in Clause 17.6.1, the Issuer and the Agent shall, at the request and expense of the Issuer, without the requirement for any consent or approval of the Holders, without undue delay effect such amendments to these Terms and Conditions as may be required by the Issuer in order to give effect to this Clause 17.
- 17.6.3 The Agent and the Issuing Agent shall always be entitled to consult with external experts prior to amendments are effected pursuant to this Clause 17. Neither the Agent nor the Issuing Agent shall be obliged to concur if in the reasonable opinion of the Agent or the Issuing Agent (as applicable), doing so would impose more onerous obligations upon it or expose it to any additional duties, responsibilities or liabilities or reduce or amend the protective provisions afforded to the Agent or the Issuing Agent in these Terms and Conditions.

17.7 Limitation of liability for the Independent Adviser

Any Independent Adviser appointed pursuant to Clause 17.3 shall not be liable whatsoever for damage or loss caused by any determination, action taken or omitted by it under or in connection with these Terms and Conditions, unless directly caused by its gross negligence or wilful misconduct. The Independent Adviser shall never be responsible for indirect or consequential loss.

18. THE AGENT

18.1 **Appointment of the Agent**

- 18.1.1 By subscribing for Capital Securities, each initial Holder appoints the Agent to act as its agent in all matters relating to the Capital Securities and the Terms and Conditions, and authorises the Agent to act on its behalf (without first having to obtain its consent, unless such consent is specifically required by these Terms and Conditions) in any legal or arbitration proceedings relating to the Capital Securities held by such Holder, including the winding-up, dissolution, liquidation, company reorganisation (*företagsrekonstruktion*) or bankruptcy (*konkurs*) (or its equivalent in any other jurisdiction) of the Issuer and in relation to any mandatory exchange of the Capital Securities for other securities (including, for the avoidance of doubt, a right for the Agent to subscribe for any such new securities on behalf of the relevant Holder). By acquiring Capital Securities, each subsequent Holder confirms such appointment and authorisation for the Agent to act on its behalf.
- 18.1.2 Each Holder shall immediately upon request provide the Agent with any such documents, including a written power of attorney (in form and substance satisfactory to the Agent), that the Agent deems necessary for the purpose of exercising its rights and/or carrying out its duties under the Terms and Conditions. The Agent is under no obligation to represent a Holder which does not comply with such request.
- 18.1.3 The Issuer shall promptly upon request provide the Agent with any documents and other assistance (in form and substance satisfactory to the Agent), that the Agent deems necessary for the purpose of exercising its rights and/or carrying out its duties under the Terms and Conditions.
- 18.1.4 The Agent is entitled to fees for all its work in such capacity and to be indemnified for costs, losses and liabilities on the terms set out in the Terms and Conditions and the Agency Agreement and the Agent's obligations as Agent under the Terms and Conditions are conditioned upon the due payment of such fees and indemnifications.
- 18.1.5 The Agent may act as agent or trustee for several issues of securities or other loans issued by or relating to the Issuer and other Group Companies notwithstanding potential conflicts of interest.

18.2 **Duties of the Agent**

- 18.2.1 The Agent shall represent the Holders in accordance with the Terms and Conditions.
- 18.2.2 When acting pursuant to the Terms and Conditions, the Agent is always acting with binding effect on behalf of the Holders. The Agent is never acting as an advisor to the Holders or the Issuer. Any advice or opinion from the Agent does not bind the Holders or the Issuer.
- 18.2.3 When acting pursuant to the Terms and Conditions, the Agent shall carry out its duties with reasonable care and skill in a proficient and professional manner.
- 18.2.4 The Agent shall treat all Holders equally and, when acting pursuant to the Terms and Conditions, act with regard only to the interests of the Holders as a group and shall not be required to have regard to the interests or to act upon or comply with any direction or request of any other person, other than as explicitly stated in the Terms and Conditions.

- 18.2.5 The Agent is always entitled to delegate its duties to other professional parties and to engage external experts when carrying out its duties as agent, without having to first obtain any consent from the Holders or the Issuer. The Agent shall however remain liable for any actions of such parties if such parties are performing duties of the Agent under the Terms and Conditions.
- The Issuer shall on demand by the Agent pay all costs for external experts engaged by it in connection with (i) matters relating to the Issuer or the Terms and Conditions which the Agent reasonably believes may be detrimental to the interests of the Holders under the Terms and Conditions, (ii) any Holders' Meeting or Written Procedure, or (iii) any amendment (whether contemplated by the Terms and Conditions or not) or waiver under the Terms and Conditions.
- 18.2.7 The Agent shall, as applicable, enter into agreements with the CSD, and comply with such agreement and the CSD Regulations applicable to the Agent, as may be necessary in order for the Agent to carry out its duties under the Terms and Conditions.
- 18.2.8 Other than as specifically set out in the Terms and Conditions, the Agent shall not be obliged to monitor (i) the financial condition of the Issuer and the Group, (ii) the performance, default or any breach by the Issuer or any other party of its obligations under the Terms and Conditions, or (iii) whether any other event specified in the Terms and Conditions has occurred or is expected to occur. Should the Agent not receive such information, the Agent is entitled to assume that no such event or circumstance exists or can be expected to occur, provided that the Agent does not have actual knowledge of such event or circumstance.
- 18.2.9 The Agent shall ensure that it receives evidence satisfactory to it that the Terms and Conditions are duly authorised and executed. The Issuer shall promptly upon request provide the Agent with such documents and evidence as the Agent reasonably considers necessary for the purpose of being able to comply with this Clause 18.2.9. Other than as set out above, the Agent shall neither be liable to the Issuer or the Holders for damage due to any documents and information delivered to the Agent not being accurate, correct and complete, unless it has actual knowledge to the contrary, nor be liable for the content, validity, perfection or enforceability of such documents.
- 18.2.10 Notwithstanding any other provision of the Terms and Conditions to the contrary, the Agent is not obliged to do or omit to do anything if it would or might in its reasonable opinion constitute a breach of any regulation.
- 18.2.11 If in the Agent's reasonable opinion the cost, loss or liability which it may incur (including reasonable fees to the Agent) in complying with instructions of the Holders, or taking any action at its own initiative, will not be covered by the Issuer, the Agent may refrain from acting in accordance with such instructions, or taking such action, until it has received such funding or indemnities (or adequate Security has been provided therefore) as it may reasonably require.
- 18.2.12 The Agent shall give a notice to the Holders:
 - (a) before it ceases to perform its obligations under the Terms and Conditions by reason of the non-payment by the Issuer of any fee or indemnity due to the Agent under the Terms and Conditions or the Agency Agreement; or
 - (b) if it refrains from acting for any reason described in Clause 18.2.11.
- 18.2.13 Subject to applicable regulations, the Agent shall promptly upon the reasonable request by a Holder forward by post any information from such Holder to the Holders which relates to the Capital Securities. The Agent may require that the requesting Holder reimburses any costs or expenses incurred, or to be incurred, by it in doing so (including a reasonable fee for its work).
- 18.2.14 Subject to the restrictions of a non-disclosure agreement entered into by the Agent in connection with these Terms and Conditions, the Agent is entitled to disclose to the Holders any document, information, event or circumstance directly or indirectly relating to the Issuer or the Capital Securities.

Notwithstanding the foregoing, the Agent may if it considers it to be beneficial to the interests of the Holders delay disclosure or refrain from disclosing certain information.

18.2.15 If a committee representing the Holders' interests under the Terms and Conditions has been appointed by the Holders in accordance with Clause 15 (*Decisions by Holders*), the members of such committee may agree with the Issuer not to disclose information received from the Issuer, provided that it, in the reasonable opinion of such members, is beneficial to the interests of the Holders. The Agent shall be a party to such agreement and receive the same information from the Issuer as the members of the committee.

18.3 Liability for the Agent

- 18.3.1 The Agent will not be liable to the Holders for damage or loss caused by any action taken or omitted by it under or in connection with these Terms and Conditions, unless directly caused by its negligence or wilful misconduct. The Agent shall never be responsible for indirect or consequential loss.
- 18.3.2 The Agent shall not be considered to have acted negligently if it has acted in accordance with advice from or opinions of reputable external experts provided to the Agent or if the Agent has acted with reasonable care in a situation when the Agent considers that it is detrimental to the interests of the Holders to delay the action in order to first obtain instructions from the Holders.
- 18.3.3 The Agent shall not be liable for any delay (or any related consequences) in crediting an account with an amount required pursuant to the Terms and Conditions to be paid by the Agent to the Holders, provided that the Agent has taken all necessary steps as soon as reasonably practicable to comply with the regulations or operating procedures of any recognised clearing or settlement system used by the Agent for that purpose.
- 18.3.4 The Agent shall have no liability to the Issuer or the Holders for damage caused by the Agent acting in accordance with instructions of the Holders given in accordance with the Terms and Conditions.
- 18.3.5 Any liability towards the Issuer which is incurred by the Agent in acting under, or in relation to, the Terms and Conditions shall not be subject to set-off against the obligations of the Issuer to the Holders under the Terms and Conditions.

18.4 **Replacement of the Agent**

- 18.4.1 Subject to Clause 18.4.6, the Agent may resign by giving notice to the Issuer and the Holders, in which case the Holders shall appoint a successor Agent at a Holders' Meeting convened by the retiring Agent or by way of Written Procedure initiated by the retiring Agent.
- 18.4.2 Subject to Clause 18.4.6, if the Agent is Insolvent, the Agent shall be deemed to resign as Agent and the Issuer shall within ten (10) Business Days appoint a successor Agent which shall be an independent financial institution or other reputable company which regularly acts as agent under debt issuances.
- 18.4.3 A Holder (or Holders) representing at least ten (10) per cent. of the Adjusted Nominal Amount may, by notice to the Issuer (such notice shall, if given by several Holders, be given by them jointly), require that a Holders' Meeting is held for the purpose of dismissing the Agent and appointing a new Agent. The Issuer may, at a Holders' Meeting convened by it or by way of Written Procedure initiated by it, propose to the Holders that the Agent be dismissed and a new Agent appointed.
- 18.4.4 If the Holders have not appointed a successor Agent within ninety (90) days after:
 - (a) the earlier of the notice of resignation was given or the resignation otherwise took place; or

(b) the Agent was dismissed through a decision by the Holders,

the Issuer shall within thirty (30) days thereafter appoint a successor Agent which shall be an independent financial institution or other reputable company with the necessary resources to act as agent under debt issuances.

- 18.4.5 The retiring Agent shall, at its own cost, make available to the successor Agent such documents and records and provide such assistance as the successor Agent may reasonably request for the purposes of performing its functions as Agent under the Terms and Conditions.
- 18.4.6 The Agent's resignation or dismissal shall only take effect upon the earlier of:
 - (a) the appointment of a successor Agent and acceptance by such successor Agent of such appointment and the execution of all necessary documentation to effectively substitute the retiring Agent; and
 - (b) the period pursuant to Clause 18.4.4 having lapsed.
- 18.4.7 Upon the appointment of a successor, the retiring Agent shall be discharged from any further obligation in respect of the Terms and Conditions but shall remain entitled to the benefit of the Terms and Conditions and remain liable under the Terms and Conditions in respect of any action which it took or failed to take whilst acting as Agent. Its successor, the Issuer and each of the Holders shall have the same rights and obligations amongst themselves under the Terms and Conditions as they would have had if such successor had been the original Agent.
- In the event that there is a change of the Agent in accordance with this Clause 18.4, the Issuer shall execute such documents and take such actions as the new Agent may reasonably require for the purpose of vesting in such new Agent the rights, powers and obligation of the Agent and releasing the retiring Agent from its further obligations under the Terms and Conditions and the Agency Agreement. Unless the Issuer and the new Agent agree otherwise, the new Agent shall be entitled to the same fees and the same indemnities as the retiring Agent.

19. THE ISSUING AGENT

- 19.1 The Issuer shall when necessary appoint an Issuing Agent to manage certain specified tasks under these Terms and Conditions and in accordance with the legislation, rules and regulations applicable to and/or issued by the CSD and relating to the Capital Securities. The Issuing Agent shall be a commercial bank or securities institution approved by the CSD.
- The Issuer shall ensure that the Issuing Agent enters into agreements with the CSD, and comply with such agreement and the CSD Regulations applicable to the Issuing Agent, as may be necessary in order for the Issuing Agent to carry out its duties relating to the Capital Securities.
- 19.3 The Issuing Agent will not be liable to the Holders for damage or loss caused by any action taken or omitted by it under or in connection with the Terms and Conditions, unless directly caused by its gross negligence or wilful misconduct. The Issuing Agent shall never be responsible for indirect or consequential loss.

20. THE CSD

- The Issuer has appointed the CSD to manage certain tasks under these Terms and Conditions and in accordance with the CSD Regulations and the other regulations applicable to the Capital Securities.
- 20.2 The CSD may retire from its assignment or be dismissed by the Issuer, provided that the Issuer has effectively appointed a replacement CSD that accedes as CSD at the same time as the old CSD retires or is dismissed and provided also that the replacement does not have a negative effect on any Holder

or the admission to trading of the Capital Securities on the Regulated Market. The replacing CSD must be authorised to professionally conduct clearing operations pursuant to the Central Securities Depository Regulation (Regulation (EU) No 909/2014) and be authorised as a central securities depository in accordance with the Central Securities Depositories and Financial Instruments Accounts Act.

21. NO DIRECT ACTIONS BY HOLDERS

- A Holder may not take any steps whatsoever against the Issuer to enforce or recover any amount due or owing to it pursuant to the Terms and Conditions, or to initiate, support or procure the winding-up, dissolution, liquidation, company reorganisation or bankruptcy in any jurisdiction of any Group Company in relation to any of the obligations and liabilities of the Issuer or any Subsidiary under the Terms and Conditions. Such steps may only be taken by the Agent.
- Clause 21.1 shall not apply if the Agent has been instructed by the Holders in accordance with the Terms and Conditions to take certain actions but fails for any reason to take, or is unable to take (for any reason other than a failure by a Holder to provide documents in accordance with Clause 17.1.2), such actions within a reasonable period of time and such failure or inability is continuing. However, if the failure to take certain actions is caused by the non-payment of any fee or indemnity due to the Agent under the Terms and Conditions or the Agency Agreement or by any reason described in Clause 17.2.11, such failure must continue for at least forty (40) Business Days after notice pursuant to Clause 17.2.12 before a Holder may take any action referred to in Clause 21.1.

22. TIME-BAR

- The right to receive repayment of the principal of the Capital Securities shall be time-barred and become void ten (10) years from the relevant Redemption Date. The right to receive payment of interest (excluding any capitalised interest) shall be time-barred and become void three (3) years from the relevant due date for payment. The Issuer is entitled to any funds set aside for payments in respect of which the Holders' right to receive payment has been time-barred and has become void.
- If a limitation period is duly interrupted in accordance with the Swedish Act on Limitations (preskriptionslag (1981:130)), a new limitation period of ten (10) years with respect to the right to receive repayment of the principal of the Capital Securities, and of three (3) years with respect to receive payment of interest (excluding capitalised interest) will commence, in both cases calculated from the date of interruption of the limitation period, as such date is determined pursuant to the provisions of the Swedish Act on Limitations.

23. NOTICES

- 23.1 Any notice or other communication to be made under or in connection with the Terms and Conditions:
 - (a) if to the Agent, shall be given at the address specified on its website www.nordictrustee.com on the Business Day prior to dispatch or, if sent by email by the Issuer, to the email address notified by the Agent to the Issuer from time to time;
 - (b) if to the Issuer, shall be given at the address registered with the Swedish Companies Registration Office on the Business Day prior to dispatch or, if sent by email by the Agent, to the email address notified by the Issuer to the Agent from time to time; and
 - (c) if to the Holders, shall be given at their addresses registered with the CSD on a date selected by the sending person which falls no more than five (5) Business Days prior to the date on which the notice or communication is sent, and by either courier delivery (if practically possible) or letter for all Holders. A Notice to the Holders shall also be published on the websites of the Issuer and the Agent.

- Any notice or other communication made by one person to another under or in connection with the Terms and Conditions shall be sent by way of courier, personal delivery or letter, or, if between the Issuer and the Agent, by email, and will only be effective:
 - (a) in case of courier or personal delivery, when it has been left at the address specified in Clause 23.1;
 - (b) in case of letter, three (3) Business Days after being deposited postage prepaid in an envelope addressed to the address specified in Clause 23.1; or
 - (c) in case of email, when received in readable form by the email recipient.
- 23.3 Any notice or other communication pursuant to the Terms and Conditions shall be in English.
- Failure to send a notice or other communication to a Holder or any defect in it shall not affect its sufficiency with respect to other Holders.

24. FORCE MAJEURE

- 24.1 Neither the Agent nor the Issuing Agent shall be held responsible for any damage arising out of any legal enactment, or any measure taken by a public authority, or war, strike, lockout, boycott, blockade, natural disaster, insurrection, civil commotion, terrorism or any other similar circumstance (a "Force Majeure Event"). The reservation in respect of strikes, lockouts, boycotts and blockades applies even if the Agent or the Issuing Agent itself takes such measures, or is subject to such measures.
- 24.2 Should a Force Majeure Event arise which prevents the Agent or the Issuing Agent from taking any action required to comply with these Terms and Conditions, such action may be postponed until the obstacle has been removed.
- 24.3 The provisions in this Clause 24 apply unless they are inconsistent with the provisions of the Central Securities Depositories and Financial Instruments Accounts Act which provisions shall take precedence.

25. GOVERNING LAW AND JURISDICTION

- 25.1 These Terms and Conditions, and any non-contractual obligations arising out of or in connection therewith, shall be governed by and construed in accordance with the laws of Sweden.
- The Issuer submits to the non-exclusive jurisdiction of the District Court of Stockholm (*Stockholms tingsrätt*).

SCHEDULE 1

CONDITIONS PRECEDENT

PART I – CONDITIONS PRECEDENT FOR INITIAL CAPITAL SECURITIES

1. CORPORATE DOCUMENTS

- (d) Copies of the articles of association and an up-to-date certificate of registration of the Issuer.
- (e) A copy of a resolution from the board of directors of the Issuer:
 - (i) approving the issue of the Initial Capital Securities, the terms of the Terms and Conditions and the Agency Agreement, and resolving to enter into such documents and any other documents necessary in connection therewith;
 - (ii) authorising a specified person or persons to execute the Terms and Conditions and the Agency Agreement; and
 - (iii) authorising a specified person or persons, on its behalf, to execute all documents and notices to be executed by it or in connection with the Terms and Conditions or the Agency Agreement
- (f) A copy of a passport or driver's license which includes a specimen of the signature of each person authorised by the resolution referred to in paragraph (b) above and actually signing the Terms and Conditions and/or the Agency Agreement.

2. FINANCE DOCUMENTS

- (g) A copy of the Terms and Conditions duly executed by the Issuer.
- (h) A copy of the Agency Agreement duly executed by the Issuer.

3. OTHER DOCUMENTS AND EVIDENCE

Such other documents and evidence as is agreed between the Agent and the Issuer.

PART II – CONDITIONS PRECEDENT FOR SUBSEQUENT CAPITAL SECURITIES

1. CORPORATE DOCUMENTS

- (i) Copies of the articles of association and an up-to-date certificate of registration of the Issuer.
- (j) A copy of a resolution from the board of directors of the Issuer:
 - (i) approving the issue of the Subsequent Capital Securities and resolving to enter into such documents and any other documents necessary in connection therewith;
 - (ii) authorising a specified person or persons to execute the documents necessary in connection with the Subsequent Capital Securities to which it is a party on its behalf; and
 - (iii) authorising a specified person or persons, on its behalf, to sign and/or dispatch all documents and notices to be signed and/or despatched by it under or in connection with the Subsequent Capital Securities.

TERMS AND CONDITIONS OF THE FIXED RATE CAPITAL SECURITIES



TERMS AND CONDITIONS FOR SWEDAVIA AB (PUBL)

UP TO SEK 3,000,000,000 SUBORDINATED PERPETUAL FIXED RATE CALLABLE CAPITAL SECURITIES

ISIN: SE0026853285

First Issue Date: 19 November 2025

SELLING RESTRICTIONS

No action is being taken that would or is intended to permit a public offering of the Capital Securities or the possession, circulation or distribution of this document or any other material relating to the Issuer or the Capital Securities in any jurisdiction other than Sweden, where action for that purpose is required. Persons into whose possession this document comes are required by the Issuer to inform themselves about, and to observe, any applicable restrictions.

PRIVACY NOTICE

The Issuer, the Agent and the Issuing Agent may collect and process personal data relating to the Holders, the Holders' representatives or agents, and other persons nominated to act on behalf of the Holders pursuant to the Terms and Conditions (name, contact details and, when relevant, holding of Capital Securities). The personal data relating to the Holders is primarily collected from the registry kept by the CSD. The personal data relating to other persons is primarily collected directly from such persons.

The personal data collected will be processed by the Issuer, the Agent and the Issuing Agent for the following purposes:

- (a) to exercise their respective rights and fulfil their respective obligations under the Terms and Conditions;
- (b) to manage the administration of the Capital Securities and payments under the Capital Securities;
- (c) to enable the Holders' to exercise their rights under the Terms and Conditions; and
- (d) to comply with their obligations under applicable laws and regulations.

The processing of personal data by the Issuer, the Agent and the Issuing Agent in relation to items (a) to (c) (inclusive) is based on their legitimate interest to exercise their respective rights and to fulfil their respective obligations under the Terms and Conditions. In relation to item (d), the processing is based on the fact that such processing is necessary for compliance with a legal obligation incumbent on the Issuer, the Agent or the Issuing Agent. Unless otherwise required or permitted by law, the personal data collected will not be kept longer than necessary given the purpose of the processing.

Personal data collected may be shared with third parties, such as the CSD, when necessary to fulfil the purpose for which such data is processed.

Subject to any legal preconditions, the applicability of which have to be assessed in each individual case, data subjects have the rights as follows. Data subjects have right to get access to their personal data and may request the same in writing at the address of the Issuer, the Agent and the Issuing Agent, respectively. In addition, data subjects have the right to (i) request that personal data is rectified or erased, (ii) object to specific processing, (iii) request that the processing be restricted and (iv) receive personal data provided by themselves in machine-readable format.

Data subjects are also entitled to lodge complaints with the relevant supervisory authority if dissatisfied with the processing carried out.

The Issuer's, the Agent's and the Issuing Agent's addresses, and the contact details for their respective Data Protection Officers (if applicable), are found on their websites www.swedavia.se, www.nordictrustee.com and www.nordea.com.

TABLE OF CONTENTS

1.	DEFINITIONS AND CONSTRUCTION	1
2.	THE CAPITAL SECURITIES	8
3.	STATUS OF THE CAPITAL SECURITIES	8
4.	USE OF PROCEEDS	9
5.	CONDITIONS PRECEDENT	10
6.	CAPITAL SECURITIES IN BOOK-ENTRY FORM	10
7.	RIGHT TO ACT ON BEHALF OF A HOLDER	11
8.	PAYMENTS IN RESPECT OF THE CAPITAL SECURITIES	11
9.	INTEREST	12
10.	OPTIONAL INTEREST DEFERRAL	13
11.	REDEMPTION AND REPURCHASE OF THE CAPITAL SECURITIES	14
12.	PRECONDITIONS TO SPECIAL EVENT REDEMPTION OR CHANGE OF CO EVENT REDEMPTION	
13.	ADMISSION TO TRADING	16
14.	DEFAULT AND ENFORCEMENT	17
15.	DECISIONS BY HOLDERS	17
16.	AMENDMENTS AND WAIVERS	23
17.	THE AGENT	23
18.	THE ISSUING AGENT	27
19.	THE CSD	28
20.	NO DIRECT ACTIONS BY HOLDERS	28
21.	TIME-BAR	28
22.	NOTICES	29
23.	FORCE MAJEURE	29
24.	GOVERNING LAW AND JURISDICTION	30
SCH	HEDULE	
1.	CONDITIONS PRECEDENT	31

1. DEFINITIONS AND CONSTRUCTION

1.1 **Definitions**

In these terms and conditions (the "Terms and Conditions"):

- "Account Operator" means a bank or other party duly authorised to operate as an account operator pursuant to the Central Securities Depositories and Financial Instruments Accounts Act and through which a Holder has opened a Securities Account in respect of its Capital Securities.
- "Accounting Event" means the receipt by the Issuer of an opinion of an independent and authorised accountant (*auktoriserad revisor*) in Sweden (experienced in such matters) to the effect that, as a result of a change in the Accounting Principles or interpretation thereof, the equity treatment of the Capital Securities as "equity" in full in the Issuer's consolidated financial statements has or will cease.
- "Accounting Principles" means the international financial reporting standards (IFRS) within the meaning of Regulation 1606/2002/EC (or as otherwise adopted or amended from time to time).
- "Adjusted Nominal Amount" means the Total Nominal Amount less the aggregate Nominal Amount of all Capital Securities owned by a Group Company, irrespective of whether such person is directly registered as owner of such Capital Securities.
- "Agency Agreement" means the agency agreement entered into before the First Issue Date, between the Issuer and the Agent, or any replacement agency agreement entered into after the First Issue Date between the Issuer and an agent.
- "Agent" means Nordic Trustee & Agency AB (publ), Swedish Reg. No. 556882-1879, or another party replacing it, as Holders' agent, in accordance with these Terms and Conditions.
- "Business Day" means a day in Sweden other than a Sunday or other public holiday. Saturdays, Midsummer Eve (*midsommarafton*), Christmas Eve (*julafton*) and New Year's Eve (*nyårsafton*) shall for the purpose of this definition be deemed to be public holidays.
- "Business Day Convention" means the first following day that is a Business Day.
- "Capital Security" means a debt instrument (*skuldförbindelse*) for the Nominal Amount and of the type set forth in Chapter 1 Section 3 of the Central Securities Depositories and Financial Instruments Accounts Act and which are governed by and issued by the Issuer under these Terms and Conditions, including the Initial Capital Securities and any Subsequent Capital Securities.
- "Central Securities Depositories and Financial Instruments Accounts Act" means the Swedish Central Securities Depositories and Financial Instruments Accounts Act (lag (1998:1479) om värdepapperscentraler och kontoföring av finansiella instrument).
- "Change of Control Event" means an event where the Government of Sweden ceases to, directly or indirectly, own or control less than 100 per cent. of the shares and the votes in the Issuer.
- "Change of Control Step-up Date" means the date falling six (6) months after the date on which a Change of Control Event has occurred.
- "CSD" means the Issuer's central securities depository and registrar in respect of the Capital Securities, Euroclear Sweden AB, Swedish Reg. No. 556112-8074, P.O. Box 191, 101 23 Stockholm, Sweden, or another party replacing it, as CSD, in accordance with these Terms and Conditions.
- "CSD Regulations" means the CSD's rules and regulations applicable to the Issuer, the Agent and the Capital Securities from time to time.
- "**Debt Register**" means the debt register (*skuldbok*) kept by the CSD in respect of the Capital Securities in which a Holder is registered.
- "Deferred Interest" has the meaning ascribed to it in Clause 10.1.3.

"Deferred Interest Payment Event" means any one or more of the following events:

- (a) declaration or payment of any distribution or dividend or any other payment made by the Issuer on its share capital or any other obligation of the Issuer which ranks or is expressed by its terms to rank junior to the Capital Securities or any Parity Securities;
- (b) declaration or payment of any distribution or dividend or any other payment made by the Issuer or any Subsidiary of the Issuer, as the case may be, on any Parity Securities;
- (c) redemption, repurchase, repayment, cancellation, reduction or other acquisition by the Issuer or any Subsidiary of the Issuer of any shares of the Issuer or any other obligation of the Issuer which ranks or is expressed by its terms to rank junior to the Capital Securities or any Parity Securities; and/or
- (d) redemption, repurchase, repayment, cancellation, reduction or other acquisition by the Issuer or any Subsidiary of the Issuer of any Parity Securities,

save for:

- (i) in each case, any compulsory distribution, dividend, other payment, redemption, repurchase, repayment, cancellation, reduction or other acquisition required by the terms of such securities or by mandatory operation of applicable law; and
- (ii) in the case of paragraph (c) above only, any redemption, repurchase, repayment, cancellation, reduction or other acquisition that is executed in connection with, or for the purpose of:
 - (A) any reduction of the quota value of the share capital of the Issuer without a corresponding return of cash, capital or assets to shareholders of the Issuer: or
 - (B) any share buyback programme then in force and duly approved by the shareholders' general meeting of the Issuer or the relevant Subsidiary of the Issuer (as applicable) or any existing or future stock option plan or free share allocation plan or other incentive plan, in all cases, reserved for directors, officers and/or employees of the Issuer or the relevant Subsidiary of the Issuer or any associated hedging transaction.

"First Call Date" means the date falling seven (7) years after the Issue Date or, to the extent such day is not a Business Day, the Business Day following from an application of the Business Day Convention.

"First Issue Date" means 19 November 2025.

"Force Majeure Event" has the meaning set forth in Clause 24.1.

"Group" means the Issuer and its Subsidiaries from time to time (each a "Group Company").

"Holder" means the person who is registered on a Securities Account as direct registered owner (direktregistrerad ägare) or nominee (förvaltare) with respect to a Capital Security.

"Holders' Meeting" means a meeting among the Holders held in accordance with Clauses 15.1 (Request for a decision), 15.2 (Convening of Holders' Meeting) and 15.4 (Majority, quorum and other provisions).

"Initial Capital Securities" means the Capital Securities issued on the First Issue Date, being in a Nominal Amount of SEK 500,000,000.

"**Insolvent**" means, in respect of a relevant person, that it is deemed to be insolvent, or admits inability to pay its debts as they fall due, in each case within the meaning of Chapter 2, Sections 7-9 of the Swedish Bankruptcy Act (*konkurslagen* (1987:672)) (or its equivalent in any other relevant jurisdiction).

"**Interest**" means the interest on the Capital Securities calculated in accordance with Clauses 9.1.1 to 9.1.3.

- "Interest Payment" means, in respect the payment of interest on an Interest Payment Date, the amount of interest payable for the relevant Interest Period in accordance with Clause 9 (Interest).
- "Interest Payment Date" means, subject to Clause 10 (*Optional Interest Deferral*), 19 November of each year or, to the extent such day is not a Business Day, the Business Day following from an application of the Business Day Convention. The first Interest Payment Date for the Capital Securities shall be 19 November 2026 and the last Interest Payment Date shall be the relevant Redemption Date.
- "Interest Period" means (i) in respect of the first Interest Period, the period from (but excluding) the First Issue Date to (and including) the first Interest Payment Date or, if the Capital Securities are redeemed before, the Redemption Date, and (ii) in respect of subsequent Interest Periods, the period from (but excluding) an Interest Payment Date to (and including) the next succeeding Interest Payment Date or, if the Capital Securities are redeemed before, the Redemption Date. An Interest Period shall not be adjusted due to an application of the Business Day Convention.

"Interest Rate" means:

- (a) in respect of the period from (but excluding) the First Issue Date to (and including) the Interest Step-up Date, a fixed rate of 4.815 per cent. *per annum*; and
- (b) in respect of the period from (but excluding) the Interest Rate Step-up Date, a fixed rate of 5.815 per cent. *per annum*.
- "Interest Rate Step-up Date" means the Interest Payment Date falling in November 2035.
- "Issuer" means Swedavia AB (publ), a public limited liability company incorporated under the laws of Sweden with Reg. No. 556797-0818.
- "Issuer Winding-up" has the meaning set forth in Clause 3.2.
- "Issuer Re-Construction" has the meaning set forth in Clause 3.2.
- "Issuing Agent" means, initially, Nordea Bank Abp, filial i Sverige, Reg. No. 516411-1683, and thereafter each other party appointed as Issuing Agent in accordance with these Terms and Conditions and the CSD Regulations.
- "Net Proceeds" means the gross proceeds from the offering of the relevant Capital Securities, minus the costs incurred by the Issuer in conjunction with the issuance and listing thereof.
- "Nominal Amount" has the meaning set forth in Clause 2.3.
- "Parity Securities" means any obligations of:
- (a) the Issuer, issued directly or indirectly by it, which rank, or are expressed to rank, *pari passu* with the Capital Securities; and
- (b) any Subsidiary of the Issuer having the benefit of a guarantee or support agreement from the Issuer which ranks or is expressed to rank *pari passu* with the Capital Securities.
- "Qualifying Securities" means securities issued directly by the Issuer following a substitution or variation of the Capital Securities in accordance with Clause 11.6 (Substitution or variation due to a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event) that have terms not materially less favourable to investors, certified by the Issuer acting reasonably (having consulted with an independent investment bank or independent financial adviser), than the terms of the Capital Securities (immediately prior to the relevant substitution or variation), provided that they:
- (a) shall include a ranking at least equal to that of the Capital Securities;
- (b) shall have at least the same Interest Rate and the same Interest Payment Dates as those applying to the Capital Securities;
- (c) shall have the same redemption rights as the Capital Securities;
- (d) shall preserve any existing rights under the Capital Securities to any accrued interest which has not been paid but which has not been cancelled in respect of the period from (and

- including) the Interest Payment Date last preceding the date of the relevant substitution or variation of the Capital Securities;
- (e) if the Capital Securities were admitted to trading on a Regulated Market immediately prior to the relevant substitution or variation, the Issuer shall use reasonable efforts to ensure that the Capital Securities are admitted to trading on a Regulated Market within thirty (30) days from their issuance (noting that no investor in the relevant Qualifying Securities (or its representative) has the right to accelerate the relevant Qualifying Securities or otherwise request a prepayment or redemption of the relevant Qualifying Securities upon a failure to admit the relevant Qualifying Securities to trading).
- "Rating Agency" means Moody's and any other rating agency of equivalent international standing requested by the Issuer to grant a corporate credit rating to the Issuer and, in each case, their successors or affiliates.
- "Rating Methodology Event" shall be deemed to occur if the Issuer has received confirmation from any Rating Agency either directly or via a publication by such Rating Agency, that due to any amendment to, clarification of or change in hybrid capital methodology or a change in the interpretation thereof, in each case occurring or becoming effective after the Issue Date:
- (a) all or any of the Capital Securities will no longer be eligible for the same or a higher amount of "equity credit" (or such other nomenclature that the relevant Rating Agency may then use to describe the degree to which an instrument exhibits the characteristics of an ordinary share) attributed to the Capital Securities at the Issue Date (or, if "equity credit" is not assigned to the Capital Securities by the relevant Rating Agency on the Issue Date, at the date on which "equity credit" is assigned by such Rating Agency for the first time); or
- (b) the length of time the Capital Securities are assigned a particular level of "equity credit" by the relevant Rating Agency is shortened as compared to the length of time that the Capital Securities were assigned that level of "equity credit" by the relevant Rating Agency under its prevailing methodology on the Issue Date.
- "Record Date" means the fifth (5) Business Day prior to:
- (a) an Interest Payment Date;
- (b) a Redemption Date; or
- (c) another relevant date,

or in each case such other Business Day falling prior to a relevant date if generally applicable on the Swedish bond market.

- "Redemption Date" means the date on which the relevant Capital Securities are to be redeemed or repurchased in accordance with Clause 11 (*Redemption and repurchase of the Capital Securities*).
- "Regulated Market" means any regulated market (as defined in Directive 2014/65/EU on markets in financial instruments).
- "Securities Account" means the account for dematerialised securities (avstämningsregister) maintained by the CSD pursuant to the Central Securities Depositories and Financial Instruments Accounts Act in which:
- (a) an owner of such security is directly registered; or
- (b) an owner's holding of securities is registered in the name of a nominee.
- "Security" means a mortgage, charge, pledge, lien, security assignment or other security interest securing any obligation of any person, or any other agreement or arrangement having a similar effect.
- "**Special Event**" means any of an Accounting Event, a Substantial Repurchase Event, a Tax Deductibility Event, a Rating Methodology Event. a Withholding Tax Event or any combination of the foregoing.

- "Subordinated Indebtedness" means any obligation of the Issuer, whether or not having a fixed maturity, which by its terms is, or is expressed to be, subordinated in the event of an Issuer Windingup to the claims of all unsubordinated creditors of the Issuer but senior to the Capital Securities or to the obligations of the Issuer in respect of any Parity Securities.
- "Subsequent Capital Securities" means any Capital Securities issued after the First Issue Date on one or more occasions.
- "Subsidiary" means, in relation to any person, any Swedish or foreign legal entity (whether incorporated or not), which at the time is a subsidiary (*dotterföretag*) to such person, directly or indirectly, as defined in the Swedish Companies Act (*aktiebolagslagen* (2005:551)).
- "Substantial Repurchase Event" shall be deemed to occur if the Issuer and/or any of its Subsidiaries repurchases and cancels or has at any time repurchased and cancelled, a principal amount of Capital Securities equal to or greater than eighty (80) per cent. of the aggregate principal amount of the Capital Securities issued (which shall include, for these purposes, any Subsequent Capital Securities).
- "Swedish Kronor" and "SEK" means the lawful currency of Sweden.
- "Tax Deductibility Event" means the receipt by the Issuer of an opinion of independent and well-reputed counsel in Sweden (experienced in such matters) to the effect that, as a result of a Tax Law Change, any interest payments under the Capital Securities were, but are no longer, tax-deductible by the Issuer for Swedish tax purposes to the same extent as any unsubordinated obligations of the Issuer.

"Tax Law Change" means:

- (a) any amendment to, clarification of, or change in, the laws or treaties (or any regulations thereunder) of Sweden, or any political subdivision or any authority thereof or therein having the power to tax, affecting taxation;
- (b) any governmental action; or
- (c) any amendment to, clarification of, or change in the application, official position or the interpretation of such law, treaty (or regulations thereunder) or governmental action or any interpretation, decision or pronouncement that provides for a position with respect to such law, treaty (or regulations thereunder) or governmental action that differs from the theretofore generally accepted position,

in each case, by any legislative body, court, governmental authority or regulatory body in Sweden, irrespective of the manner in which such amendment, clarification, change, action, pronouncement, interpretation or decision is made known, which amendment, clarification or change is effective or such governmental action, pronouncement, interpretation or decision is announced on or after the date falling five (5) Business Days prior to the First Issue Date.

- "Total Nominal Amount" means the total aggregate Nominal Amount of the Capital Securities outstanding at the relevant time.
- "Withholding Tax Event" shall be deemed to occur if, as a result of any Tax Law Change, in making any payments on the Capital Securities, the Issuer has paid or will or would on the next Interest Payment Date be required to pay additional amounts on the Capital Securities and the Issuer cannot avoid the foregoing by taking reasonable measures available to it as evidenced by an opinion of an independent and well-reputed counsel in Sweden (experienced in such matters) to that effect.
- "Written Procedure" means the written or electronic procedure for decision making among the Holders in accordance with Clauses 15.1 (Request for a decision), 15.3 (Instigation of Written Procedure) and 15.4 (Majority, quorum and other provisions).

1.2 Construction

- 1.2.1 Unless a contrary indication appears, any reference in these Terms and Conditions to:
 - (a) "assets" includes present and future properties, revenues and rights of every description;
 - (b) any agreement or instrument is a reference to that agreement or instrument as supplemented, amended, novated, extended, restated or replaced from time to time;

- (c) a "**regulation**" includes any law, regulation, rule or official directive (whether or not having the force of law) of any governmental, intergovernmental or supranational body, agency or department;
- (d) a provision of regulation is a reference to that provision as amended or re-enacted; and
- (e) a time of day is a reference to Stockholm time.
- 1.2.2 When ascertaining whether a limit or threshold specified in Swedish Kronor has been attained or broken, an amount in another currency shall be counted on the basis of the rate of exchange for such currency against Swedish Kronor for the previous Business Day, as published by the Swedish Central Bank (*Riksbanken*) on its website (www.riksbank.se). If no such rate is available, the most recently published rate shall be used instead.
- 1.2.3 A notice shall be deemed to be sent by way of press release if it is made available to the public within Sweden promptly and in a non-discriminatory manner.
- 1.2.4 No delay or omission of the Agent or of any Holder to exercise any right or remedy under the Terms and Conditions shall impair or operate as a waiver of any such right or remedy.
- 1.2.5 The selling restrictions, the privacy notice and any other information contained in this document before the table of contents section do not form part of these Terms and Conditions and may be updated without the consent of the Holders and the Agent.

2. THE CAPITAL SECURITIES

- 2.1 The Capital Securities are denominated in Swedish Kronor and each Capital Security is constituted by these Terms and Conditions. The Issuer undertakes to make payments in relation to the Capital Securities (subject to and in accordance with these Terms and Conditions) and to comply with these Terms and Conditions.
- 2.2 By subscribing for Capital Securities, each initial Holder agrees that the Capital Securities shall benefit from and be subject to the Terms and Conditions and by acquiring Capital Securities, each subsequent Holder confirms such agreement.
- 2.3 The initial nominal amount of each Capital Security is SEK 1,250,000 (the "Nominal Amount"). The maximum Nominal Amount of the Initial Capital Securities as at the First Issue Date is SEK 500,000,000. All Initial Capital Securities are issued on a fully paid basis at an issue price of one hundred (100) per cent. of the Initial Nominal Amount.
- 2.4 The ISIN for the Capital Securities is SE0026853285.
- Provided that the relevant conditions precedent have been duly received (or waived) by the Agent in accordance with Clause 5.1.2, the Issuer may, on one or several occasions, issue Subsequent Capital Securities. Subsequent Capital Securities shall benefit from and be subject to the Terms and Conditions, and, for the avoidance of doubt, the ISIN, the interest rate, the currency, the Nominal Amount and the perpetual nature applicable to the Initial Capital Securities shall apply to Subsequent Capital Securities. The issue price of the Subsequent Capital Securities may be set at the Nominal Amount, a discount or a premium compared to the Nominal Amount. The maximum Total Nominal Amount of the Capital Securities (the Initial Capital Securities and all Subsequent Capital Securities) may not exceed SEK 3,000,000,000 unless a consent from the Holders is obtained in accordance with paragraph (c) of Clause 15.4.2. Each Subsequent Capital Security shall entitle its holder to Interest in accordance with Clause 9.1, and otherwise have the same rights as the Initial Capital Securities.
- 2.6 The Capital Securities are freely transferable but the Holders may be subject to purchase or transfer restrictions with regard to the Capital Securities, as applicable, under local regulation to which a Holder may be subject. Each Holder must ensure compliance with such restrictions at its own cost and expense.

3. STATUS OF THE CAPITAL SECURITIES

3.1 The Capital Securities, including the obligation to pay interest thereon, constitute direct, general, unconditional, unsecured and subordinated obligations of the Issuer. The rights and claims of the Holders in respect of the Capital Securities against the Issuer, are subordinated as described in Clause 3.2 below.

3.2 In the event of:

- (a) a voluntary or involuntary liquidation (*likvidation*) or bankruptcy (*konkurs*) of the Issuer (each an "Issuer Winding-up"), the Holders (or the Agent on their behalf) shall, in respect of their Capital Securities, have a claim for the principal amount of their Capital Securities and any accrued and unpaid interest (including any Deferred Interest) thereon and such claims will rank:
 - (i) *pari passu* without any preference among themselves and with any present or future claims in respect of obligations of the Issuer in respect of Parity Securities;
 - (ii) in priority to all present and future claims in respect of:
 - (A) the ordinary shares of the Issuer; and
 - (B) any other obligation of the Issuer expressed by its terms as at its original issue date to rank junior to the Capital Securities or any Parity Securities; and
 - (iii) junior in right of payment to any present or future claims of:
 - (A) all unsubordinated obligations of the Issuer; and
 - (B) all Subordinated Indebtedness; or
- (b) a company re-construction (*företagsrekonstruktion*) of the Issuer under the Swedish Company Reorganisation Act (*lag* (2022:964) om *företagsrekonstruktion*) (an "**Issuer Reconstruction**"), the Holders (or the Agent on their behalf) shall, in respect of their Capital Securities, have a claim for the principal amount of their Capital Securities and any accrued and unpaid interest (including any Deferred Interest) thereon and such claims will rank:
 - (i) pari passu without any preference among themselves and with any present or future claims in respect of obligations of the Issuer in respect of Parity Securities; and
 - (ii) junior in right of payment to any present or future claims of:
 - (A) all unsubordinated obligations of the Issuer; and
 - (B) all Subordinated Indebtedness.

Claims in respect of the share capital of the Issuer are not subject to loss absorbing measures under an Issuer Re-construction.

3.3 Subject to applicable law, no Holder may exercise, claim or plead any right of set-off, compensation or retention in respect of any amount owed to it by the Issuer in respect of, or arising under or in connection with the Capital Securities and each Holder shall, by virtue of its holding of any Capital Security, be deemed to have waived all such rights of set-off, compensation or retention.

4. USE OF PROCEEDS

- 4.1 The Issuer shall use the Net Proceeds from the issue of the Initial Capital Securities for its general corporate purposes, including refinancing financial indebtedness and financing acquisitions and investments.
- 4.2 The Issuer shall use the Net Proceeds from the issue of any Subsequent Capital Securities for its general corporate purposes, including financing acquisitions and investments.

5. CONDITIONS PRECEDENT

5.1 Conditions precedent for disbursement

- 5.1.1 The Issuer shall provide to the Agent, prior to the First Issue Date (or such later time as agreed by the Agent), each document and other evidence listed in Part I of <u>Schedule 1</u> (*Conditions Precedent*) in form and substance satisfactory to the Agent.
- 5.1.2 The Issuer shall provide to the Agent, no later 9.00 a.m. three (3) Business Days prior to the issue date (or such later time as agreed by the Agent) in respect of Subsequent Capital Securities, each document and other evidence set out in Part II of Schedule 1 (Conditions Precedent) in form and substance satisfactory to the Agent.
- 5.1.3 The Agent shall confirm to the Issuing Agent when it is satisfied that the conditions in Clause 5.1.1 or, 5.1.2 as the case may be, have been fulfilled (or amended or waived in accordance with Clause 16 (*Amendments and waivers*)).

5.2 Settlement and disbursement

- 5.2.1 Following receipt by the Issuing Agent of the confirmation in accordance with Clause 5.1.3, the Issuing Agent shall settle the issuance of the Initial Capital Securities and pay the Net Proceeds to the Issuer on the First Issue Date.
- 5.2.2 Following receipt by the Issuing Agent of the confirmation in accordance with Clause 5.1.3, the Issuing Agent shall settle the issuance of the Subsequent Capital Securities and pay the Net Proceeds to the Issuer on the relevant issue date.

5.3 No responsibility for documentation

The Agent may assume that the documentation and evidence delivered to it pursuant to this Clause 5 is accurate, legally valid, enforceable, correct, true and complete unless it has actual knowledge to the contrary and the Agent does not have to verify or assess the contents of any such documentation or evidence. The Agent does not have any obligation to review the documentation and evidence set out in this Clause 5 from a legal or commercial perspective of the Holders.

6. CAPITAL SECURITIES IN BOOK-ENTRY FORM

- The Capital Securities will be registered for the Holders on their respective Securities Accounts and no physical Capital Securities will be issued. Accordingly, the Capital Securities will be registered in accordance with the Central Securities Depositories and Financial Instruments Accounts Act. Registration requests relating to the Capital Securities shall be directed to an Account Operator. The Debt Register shall constitute conclusive evidence of the persons who are Holders and their holdings of Capital Securities.
- Those who according to assignment, Security, the provisions of the Swedish Children and Parents Code (*föräldrabalken* (1949:381)), conditions of will or deed of gift or otherwise have acquired a right to receive payments in respect of a Capital Security shall register their entitlements to receive payment in accordance with the Central Securities Depositories and Financial Instruments Accounts Act.

- 6.3 The Issuer and the Agent shall at all times be entitled to obtain information from the Debt Register. At the request of the Agent, the Issuer shall promptly obtain such information and provide it to the Agent. For the purpose of carrying out any administrative procedure that arises out of the Terms and Conditions, the Issuing Agent shall be entitled to obtain information from the Debt Register.
- The Issuer shall issue any necessary power of attorney to such persons employed by the Agent, as notified by the Agent, in order for such individuals to independently obtain information directly from the Debt Register. The Issuer may not revoke any such power of attorney unless directed by the Agent or unless consent thereto is given by the Holders.
- 6.5 The Issuer and the Agent may use the information referred to in Clause 6.3 only for the purposes of carrying out their duties and exercising their rights in accordance with the Terms and Conditions and the Agency Agreement and shall not disclose such information to any Holder or third party unless necessary for such purposes.

7. RIGHT TO ACT ON BEHALF OF A HOLDER

- 7.1 If any person other than a Holder (including the owner of a Capital Security, if such person is not the Holder), wishes to exercise any rights under the Terms and Conditions, it must obtain a power of attorney or other authorisation from the Holder or a successive, coherent chain of powers of attorney or authorisations starting with the Holder and authorising such person.
- 7.2 A Holder may issue one or several powers of attorney or other authorisations to third parties to represent it in relation to some or all of the Capital Securities held by it. Any such representative may act independently under the Terms and Conditions in relation to the Capital Securities for which such representative is entitled to represent the Holder.
- 7.3 The Agent shall only have to examine the face of a power of attorney or other authorisation that has been provided to it pursuant to Clause 7.2 and may assume that such document has been duly authorised, is valid, has not been revoked or superseded and that it is in full force and effect, unless otherwise is apparent from its face or the Agent has actual knowledge to the contrary.
- 7.4 These Terms and Conditions shall not affect the relationship between a Holder who is the nominee (*förvaltare*) with respect to a Capital Security and the owner of such Capital Security, and it is the responsibility of such nominee to observe and comply with any restrictions that may apply to it in this capacity.

8. PAYMENTS IN RESPECT OF THE CAPITAL SECURITIES

- 8.1 Any payment or repayment under the Terms and Conditions, or any amount due in respect of a repurchase of any Capital Securities, shall be made to such person who is registered as a Holder on the Record Date prior to an Interest Payment Date or other relevant payment date, or to such other person who is registered with the CSD on such Record Date as being entitled to receive the relevant payment, repayment or repurchase amount.
- 8.2 Provided that a Holder has registered an income account (avkastningskonto) for the relevant Securities Account on the applicable Record Date, the CSD shall procure that principal, interest and other payments under the Capital Securities are deposited to such income account on the relevant payment date. If an income account has not been registered on the Record Date for the payment, no payment will be effected by the CSD to such Holder. The outstanding amount will instead be held by the Issuer until the person that was registered as a Holder on the relevant Record Date has made a valid request for such amount. Should the CSD, due to a delay on behalf of the Issuer or some other obstacle, not be able to effect payments as aforesaid, the Issuer shall procure that such amounts are paid to the persons who are registered as Holders on the relevant Record Date as soon as possible after such obstacle has been removed.
- 8.3 If, due to any obstacle for the CSD, the Issuer cannot make a payment or repayment, such payment or repayment may be postponed until the obstacle has been removed.

- 8.4 If payment or repayment is made in accordance with this Clause 8, the Issuer shall be deemed to have fulfilled its obligation to pay, irrespective of whether such payment was made to a person not entitled to receive such amount (unless the Issuer has actual knowledge of the fact that the payment was made to the wrong person).
- 8.5 The Issuer is not liable to gross-up any payments under the Terms and Conditions by virtue of any withholding tax, public levy or the similar.

9. INTEREST

9.1 Interest

- 9.1.1 Each Initial Capital Security carries Interest at the applicable Interest Rate applied to the Nominal Amount from (but excluding) the First Issue Date up to (and including) the relevant Redemption Date. Any Subsequent Capital Security will carry Interest at the Interest Rate applied to the Nominal Amount from (but excluding) the Interest Payment Date falling immediately prior to its issuance (or the First Issue Date if there is no such Interest Payment Date) up to (and including) the relevant Redemption Date.
- 9.1.2 Interest accrues during an Interest Period. Subject to Clause 10 (*Optional Interest Deferral*), payment of Interest in respect of the Capital Securities shall be made to the Holders on each Interest Payment Date for the preceding Interest Period.
- 9.1.3 Interest shall be calculated on the basis of a 360-day year comprised of twelve (12) months of thirty (30) days each and, in case of an incomplete month, the actual number of days elapsed (30/360-days basis).
- 9.1.4 Payment of Interest in respect of each Interest Period may be deferred in accordance with Clause 10 (*Optional Interest Deferral*).

9.2 **Step-up after a Change of Control Event**

Notwithstanding any other provision of this Clause 9, if the Issuer does not elect to redeem the Capital Securities in accordance with Clause 11.5 (*Voluntary redemption due to Change of Control Event*) following the occurrence of a Change of Control Event, the then prevailing Interest Rate, and each subsequent Interest Rate otherwise determined in accordance with the provisions of this Clause 9, on the Capital Securities shall be increased by 500 basis points *per annum* with effect from (but excluding) the Change of Control Step-up Date up to (and including) the relevant Redemption Date.

9.3 **Default interest**

If the Issuer fails to pay any amount payable by it pursuant to Clause 10.3 (Mandatory settlement of Deferred Interest) or Clause 11 (Redemption and repurchase of the Capital Securities) (except for Clause 11.1 (No maturity), Clause 11.2 (Purchase of Capital Securities by Group Companies) and Clause 11.8 (Cancellation of Capital Securities)) on its due date, default interest shall accrue on the overdue amount from (but excluding) the due date up to (and including) the date of actual payment at a rate which is 200 basis points per annum higher than the Interest Rate. Accrued default interest shall not be capitalised. No default interest shall accrue where the failure to pay was solely attributable to the Agent or the CSD, in which case the Interest Rate shall apply instead.

10. OPTIONAL INTEREST DEFERRAL

10.1 **Deferral of Interest Payments**

10.1.1 The Issuer may, at any time and at its sole discretion, elect to defer any Interest Payment, in whole or in part, which is otherwise scheduled to be paid on an Interest Payment Date (except on any Interest Payment Date on which the Capital Securities are to be redeemed) by giving notice of such election to the Holders, the Issuing Agent and the Agent in accordance with Clause 23 (*Notices*) not less than seven (7) Business Days prior to the relevant Interest Payment Date. Notwithstanding the foregoing,

failure to give such notice shall not prejudice the right of the Issuer to defer any Interest Payment as described above.

- 10.1.2 If the Issuer makes only a partial payment of interest on an Interest Payment Date, such amount shall be applied equally to each Capital Security.
- Any Interest Payment so deferred pursuant to this Clause 10 shall, from (but excluding) the Interest Payment Date on which such Interest Payment would (but for its deferral) have been payable to (and including) the date on which it is paid in full, itself bear interest at the Interest Rate prevailing from time to time (which interest shall compound on each Interest Payment Date) and, for so long as the same remains unpaid, such deferred interest (together with the interest accrued thereon) shall constitute "Deferred Interest".
- 10.1.4 The deferral of an Interest Payment in accordance with this Clause 10 shall not constitute a default pursuant to Clause 14 (*Default and Enforcement*) by the Issuer under the Capital Securities or for any other purpose.

10.2 **Optional settlement of Deferred Interest**

Deferred Interest may be paid, in whole or in part, at any time at the option of the Issuer following delivery of a notice to such effect given by the Issuer to the Holders, the Issuing Agent and the Agent in accordance with Clause 23 (*Notices*) not less than seven (7) Business Days prior to the date (to be specified in such notice) on which the Issuer will pay such Deferred Interest. Any such notice shall state the date fixed for the payment and the relevant Record Date and be irrevocable.

10.3 Mandatory settlement of Deferred Interest

- 10.3.1 The Issuer shall pay any Deferred Interest, in whole but not in part, on the first to occur of the following dates:
 - (a) the tenth (10th) Business Day following the date on which a Deferred Interest Payment Event occurs:
 - (b) any Interest Payment Date in respect of which the Issuer does not elect to defer all of the interest accrued in respect of the relevant Interest Period; and
 - (c) the date on which the Capital Securities are redeemed or repaid in accordance with Clause 11 (*Redemption and repurchase of the Capital Securities*) or Clause 14 (*Default and Enforcement*).
- 10.3.2 Notice of any Deferred Interest Payment Event shall be given by the Issuer to the Holders, the Issuing Agent and the Agent in accordance with Clause 23 (*Notices*) within three (3) Business Days of such event.

11. REDEMPTION AND REPURCHASE OF THE CAPITAL SECURITIES

11.1 No maturity

The Capital Securities are perpetual and have no specified maturity date. The Issuer may only redeem the Capital Securities in the circumstances described in this Clause 11 (*Redemption and repurchase of the Capital Securities*). The Capital Securities are not redeemable at the option of the Holders at any time.

11.2 Purchase of Capital Securities by Group Companies

- 11.2.1 The Issuer or any other Group Company may, subject to applicable regulations, at any time and at any price purchase Capital Securities on the market or in any other way.
- 11.2.2 Capital Securities held by a Group Company may at such Group Company's discretion be retained or sold but, if held by the Issuer, not cancelled except in connection with a Substantial Repurchase Event.

11.3 Voluntary total redemption (call option)

The Issuer may redeem all, but not some only, of the outstanding Capital Securities in full on the First Call Date or on any Interest Payment Date thereafter at an amount per Capital Security equal to one hundred (100) per cent. of the Nominal Amount together with any Deferred Interest and any other accrued and unpaid interest up to (and including) the Redemption Date.

11.4 Voluntary redemption due to a Special Event

Upon the occurrence of a Special Event which is continuing, the Issuer may, subject to Clause 12 (*Preconditions to Special Event Redemption or Change of Control Event Redemption*), redeem all, but not some only, of the Capital Securities at an amount equal to:

- (c) one hundred and one (101) per cent. of their principal amount, where such redemption occurs before the First Call Date; or
- (d) one hundred (100) per cent. of their principal amount, where such redemption occurs on or after the First Call Date,

together, in each case, with any Deferred Interest and any other accrued and unpaid interest up to (and including) the Redemption Date.

11.5 Voluntary redemption due to a Change of Control Event

- 11.5.1 Upon the occurrence of a Change of Control Event, the Issuer may, no later than the Change of Control Step-up Date, subject to Clause 12 (*Preconditions to Special Event Redemption or Change of Control Event Redemption*), redeem all, but not some only, of the Capital Securities at an amount equal to:
 - (e) one hundred and one (101) per cent. of their principal amount, where such redemption occurs before the First Call Date; or
 - (f) one hundred (100) per cent. of their principal amount, where such redemption occurs on or after the First Call Date,

together, in each case, with any Deferred Interest and any other accrued and unpaid interest up to (and including) the Redemption Date.

Immediately upon the Issuer becoming aware that a Change of Control Event has occurred, the Issuer shall give notice to the Agent, the Issuing Agent and the Holders in accordance with Clause 23 (*Notices*) specifying the nature of the Change of Control Event.

Substitution or variation due to a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event

If a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event occurs, the Issuer may, at its option but subject to giving notice in accordance with Clause 11.7 (*Notice of redemption, substitution or variation*), substitute or vary the terms of all (but not some only) of the outstanding Capital Securities without any requirement for the consent or approval of the Holders, so that they become or remain, as applicable, Qualifying Securities, provided that such substitution or variation does not itself give rise to any right of the Issuer to redeem, substitute or vary the terms of the Capital Securities in accordance with this Clause 11.6 in relation to the Qualifying Securities so substituted or varied.

11.7 Notice of redemption, substitution or variation

Any redemption, substitution or variation in accordance with Clauses 11.3, 11.4, 11.5.1 or 11.6 shall be made by the Issuer giving not less than thirty (30) Business Days' notice and not more than sixty (60) Business Days' notice to the Holders, the Issuing Agent and the Agent, in each case calculated from the effective date of the notice. The Notice from the Issuer shall specify the Redemption Date and also the Record Date on which a person shall be registered as a Holder to receive the amounts due on such Redemption Date. The notice is irrevocable but may in the case of a redemption in accordance with Clause 11.3, at the Issuer's discretion, contain one or more conditions precedent that

shall be satisfied prior to the Record Date. Upon fulfilment of the conditions precedent (if any), the Issuer shall redeem the Capital Securities in full at the applicable amount on the specified Redemption Date.

11.8 Cancellation of Capital Securities

All Capital Securities which are redeemed pursuant to this Clause 11 and all Capital Securities purchased and elected to be cancelled pursuant to Clause 11.2 (*Purchase of Capital Securities by Group Companies*) will be cancelled and may not be reissued or resold. The Issuer shall promptly inform the Holders, the Agent and the Issuing Agent in accordance with Clause 23 (*Notices*) of any such cancellation and for so long as the Capital Securities are admitted to trading on Nasdaq Stockholm or another Regulated Market, as the case may be, and the rules of such exchange so require, the Issuer shall promptly inform Nasdaq Stockholm or the Regulated Market, as the case may be, of the cancellation of any Capital Securities under this Clause 11.

12. PRECONDITIONS TO SPECIAL EVENT REDEMPTION OR CHANGE OF CONTROL EVENT REDEMPTION

- Prior to the publication of any notice of redemption pursuant to Clause 11 (*Redemption and repurchase of the Capital Securities*) (other than redemption pursuant to Clause 11.3 (*Voluntary total redemption (call option)*)), the Issuer shall deliver to the Agent a certificate signed by two authorised signatories of the Issuer stating that the relevant requirement or circumstance giving rise to the right to redeem the Capital Securities is satisfied.
- In addition, in the case of an Accounting Event, a Tax Deductibility Event or a Withholding Tax Event, the Issuer shall deliver to the Agent and the Issuing Agent the relevant opinion of the independent legal, accounting or other tax advisers (as applicable) to the effect that the relevant requirement or circumstance giving rise to the right to redeem is satisfied (save, in the case of a Withholding Tax Event, as to whether reasonable measures to avoid paying additional amounts are available to the Issuer). Such certificate and, if applicable, opinion, shall be conclusive and binding on the Holders provided that the opinion complies with the opinion requirements set out in this Clause 12.2.
- 12.3 Any redemption of the Capital Securities in accordance with Clause 11 (*Redemption and repurchase of the Capital Securities*) shall be conditional on all Deferred Interest being paid in full in accordance with the provisions of Clause 10.3 (*Mandatory settlement of Deferred Interest*) on or prior to the date of such redemption.

13. ADMISSION TO TRADING

The Issuer has the intention and shall use its best efforts (without assuming any legal or contractual obligation) to ensure that:

- (a) the Initial Capital Securities are admitted to trading on the corporate bond list of Nasdaq Stockholm or, if such admission to trading is not possible to obtain or maintain, admitted to trading on another Regulated Market within thirty (30) days after the First Issue Date;
- (b) any Subsequent Capital Securities are admitted to trading on the corporate bond list of Nasdaq Stockholm or, if such admission to trading is not possible to obtain or maintain, admitted to trading on another Regulated Market within thirty (30) days after the relevant issue date of such Subsequent Capital Securities, unless the relevant Subsequent Capital Securities are issued before the expiry of the thirty (30) day period referred to in paragraph (a) above, in which case such Subsequent Capital Securities shall be admitted to trading within thirty (30) days after the First Issue Date together with the Initial Capital Securities; and
- (c) the Capital Securities, once admitted to trading on the relevant Regulated Market, continue being listed thereon (however, taking into account the rules and regulations of the relevant

Regulated Market and the CSD (as amended from time to time) preventing trading in the Capital Securities in close connection to the redemption of the Capital Securities).

14. DEFAULT AND ENFORCEMENT

14.1 **Proceedings**

- 14.1.1 Without prejudice to the Issuer's right to defer the payment of interest under Clause 10 (*Optional Interest Deferral*), if a default is made by the Issuer for a period of thirty (30) days or more in relation to the payment of any interest, principal or premium in respect of the Capital Securities which is due and payable (a "**Default**"), then the Issuer shall be deemed to be in default under the Capital Securities and the Agent (acting on instructions of the Holders in accordance with these Terms and Conditions) or (subject to Clause 20 (*No direct action by Holders*)) any Holder may institute proceedings for an Issuer Winding-up provided that such Default is still continuing.
- In the event of an Issuer Winding-up, a Holder may, provided such Holder does not contravene a previously adopted resolution in accordance with Clause 15.4.2 or 15.4.3 (if any), either independently or through the Agent prove and/or claim in such Issuer Winding-up in respect of the Capital Securities, such claim being for such amount, and being subordinated in such manner, as is provided under Clause 3.2.

14.2 **Enforcement**

The Agent (acting on the instructions of the Holders in accordance with these Terms and Conditions) may institute such proceedings against the Issuer as it may think fit to enforce any term or condition binding on the Issuer under the Capital Securities but in no event shall the Issuer, by virtue of the institution of any such proceedings, be obliged to pay any sum or sums sooner than the same would otherwise have been payable by it.

14.3 Extent of Holders' Remedy

No remedy against the Issuer, other than as referred to in this Clause 14, shall be available to the Agent and the Holders in relation to the Capital Securities, whether for the recovery of amounts owing in respect of the Capital Securities or in respect of any breach by the Issuer of any of its other obligations under or in respect of the Capital Securities.

15. DECISIONS BY HOLDERS

15.1 Request for a decision

- 15.1.1 A request by the Agent for a decision by the Holders on a matter relating to the Terms and Conditions shall (at the option of the Agent) be dealt with at a Holders' Meeting or by way of a Written Procedure.
- Any request from the Issuer or a Holder (or Holders) representing at least ten (10) per cent. of the Adjusted Nominal Amount (such request shall, if made by several Holders, be made by them jointly) for a decision by the Holders on a matter relating to the Terms and Conditions shall be directed to the Agent and dealt with at a Holders' Meeting or by way a Written Procedure, as determined by the Agent. The person requesting the decision may suggest the form for decision making, but if it is in the Agent's opinion more appropriate that a matter is dealt with at a Holders' Meeting than by way of a Written Procedure, it shall be dealt with at a Holders' Meeting.
- 15.1.3 The Agent may refrain from convening a Holders' Meeting or instigating a Written Procedure if:
 - (a) the suggested decision must be approved by any person in addition to the Holders and such person has informed the Agent that an approval will not be given; or
 - (b) the suggested decision is not in accordance with applicable regulations.

- 15.1.4 The Agent shall not be responsible for the content of a notice for a Holders' Meeting or a communication regarding a Written Procedure unless and to the extent it contains information provided by the Agent.
- 15.1.5 Should the Agent not convene a Holders' Meeting or instigate a Written Procedure in accordance with these Terms and Conditions, without Clause 15.1.3 being applicable, the Issuer or the Holder(s) requesting a decision by the Holders may convene such Holders' Meeting or instigate such Written Procedure, as the case may be, instead. The Issuer or the Issuing Agent shall upon request provide the convening Holder(s) with the information available in the Debt Register in order to convene and hold the Holders' Meeting or instigate and carry out the Written Procedure, as the case may be. The Issuer or Holder(s), as applicable, shall supply to the Agent a copy of the dispatched notice or communication.
- Should the Issuer want to replace the Agent, it may (i) convene a Holders' Meeting in accordance with Clause 15.2 (*Convening of Holders' Meeting*) or (ii) instigate a Written Procedure by sending communication in accordance with Clause 15.3 (*Instigation of Written Procedure*). After a request from the Holders pursuant to Clause 18.4.3, the Issuer shall no later than ten (10) Business Days after receipt of such request (or such later date as may be necessary for technical or administrative reasons) convene a Holders' Meeting in accordance with Clause 15.2. The Issuer shall inform the Agent before a notice for a Holders' Meeting or communication relating to a Written Procedure where the Agent is proposed to be replaced is sent and supply to the Agent a copy of the dispatched notice or communication.
- 15.1.7 Should the Issuer or any Holder(s) convene a Holders' Meeting or instigate a Written Procedure pursuant to Clause 15.1.5 or 15.1.6, then the Agent shall no later than five (5) Business Days' prior to dispatch of such notice or communication be provided with a draft thereof. The Agent may further append information from it together with the notice or communication, provided that the Agent supplies such information to the Issuer or the Holder(s), as the case may be, no later than one (1) Business Day prior to the dispatch of such notice or communication.

15.2 Convening of Holders' Meeting

- 15.2.1 The Agent shall convene a Holders' Meeting by way of notice to the Holders as soon as practicable and in any event no later than five (5) Business Days after receipt of a complete notice from the Issuer or the Holder(s) (or such later date as may be necessary for technical or administrative reasons).
- 15.2.2 The notice pursuant to Clause 15.2.1 shall include:
 - (a) time for the meeting;
 - (b) place for the meeting;
 - (c) a specification of the Record Date on which a person must be registered as a Bondholder in order to be entitled to exercise voting rights;
 - (d) a form of power of attorney;
 - (e) the agenda for the meeting;
 - (f) any applicable conditions precedent and conditions subsequent;
 - (g) the reasons for, and contents of, each proposal;
 - (h) if the proposal concerns an amendment to the Terms and Conditions, the details of such proposed amendment;
 - (i) if a notification by the Holders is required in order to attend the Holders' Meeting, information regarding such requirement; and
 - (j) information on where additional information (if any) will be published.

- 15.2.3 The Holders' Meeting shall be held no earlier than ten (10) Business Days and no later than thirty (30) Business Days after the effective date of the notice.
- 15.2.4 Without amending or varying these Terms and Conditions, the Agent may prescribe such further regulations regarding the convening and holding of a Holders' Meeting as the Agent may deem appropriate. Such regulations may include a possibility for Holders to vote without attending the meeting in person.

15.3 **Instigation of Written Procedure**

- 15.3.1 The Agent shall instigate a Written Procedure by way of sending a communication to the Holders as soon as practicable and in any event no later than five (5) Business Days after receipt of a complete communication from the Issuer or the Holder(s) (or such later date as may be necessary for technical or administrative reasons).
- 15.3.2 A communication pursuant to Clause 15.3.1 shall include:
 - (a) a specification of the Record Date on which a person must be registered as a Holder in order to be entitled to exercise voting rights;
 - (b) instructions and directions on where to receive a form for replying to the request (such form to include an option to vote yes or no for each request) as well as a form of power of attorney;
 - (c) the stipulated time period within which the Holder must reply to the request (such time period to last at least ten (10) Business Days and not longer than thirty (30) Business Days from the effective date of the communication pursuant to Clause 15.3.1);
 - (d) any applicable conditions precedent and conditions subsequent;
 - (e) the reasons for, and contents of, each proposal;
 - (f) if a proposal concerns an amendment to the Terms and Conditions, the details of such proposed amendment;
 - (g) if the voting is to be made electronically, instructions for such voting; and
 - (h) information on where additional information (if any) will be published.
- 15.3.3 If so elected by the person requesting the Written Procedure and provided that it is also disclosed in the communication pursuant to Clause 15.3.1, when consents from Holders representing the requisite majority of the total Adjusted Nominal Amount pursuant to Clauses 15.4.2 and 15.4.3 have been received in a Written Procedure, the relevant decision shall be deemed to be adopted pursuant to Clause 15.4.2 or 15.4.3, as the case may be, even if the time period for replies in the Written Procedure has not yet expired.
- 15.3.4 The Agent may, during the Written Procedure, provide information to the Issuer by way of updates whether or not quorum requirements have been met and about the eligible votes received by the Agent, including the portion consenting or not consenting to the proposal(s) or refraining from voting (as applicable).

15.4 Majority, quorum and other provisions

- Only a Holder, or a person who has been provided with a power of attorney or other authorisation pursuant to Clause 7 (*Right to act on behalf of a Holder*) from a Holder:
 - (a) on the Record Date specified in the notice pursuant to Clause 15.2.2, in respect of a Holders' Meeting, or

(b) on the Record Date specified in the communication pursuant to Clause 15.3.2, in respect of a Written Procedure,

may exercise voting rights as a Holder at such Holders' Meeting or in such Written Procedure, provided that the relevant Capital Securities are included in the Adjusted Nominal Amount. Each whole Capital Security entitles to one vote and any fraction of a Capital Security voted for by a person shall be disregarded. Such Business Day specified pursuant to paragraph (a) or (b) above must fall no earlier than one (1) Business Day after the effective date of the notice or communication, as the case may be.

- 15.4.2 The following matters shall require the consent of Holders representing at least sixty-six and two thirds (66%) per cent. of the Adjusted Nominal Amount for which Holders are voting at a Holders' Meeting or for which Holders reply in a Written Procedure in accordance with the instructions given pursuant to Clause 15.3.2:
 - (a) a change of Issuer;
 - (b) a change to the currency, denomination, status or transferability of the Capital Securities;
 - (c) the issue of any Subsequent Capital Securities, if the Total Nominal Amount of the Capital Securities exceeds, or if such issue would cause the Total Nominal Amount to at any time exceed, SEK 3,000,000,000 (for the avoidance of doubt, for which consent shall be required at each occasion such Subsequent Capital Securities are issued);
 - (d) a change to the Interest Rate or the Nominal Amount;
 - (e) any delay of the due date for payment of any interest on the Capital Securities other than as permitted pursuant to Clause 10 (*Optional Interest Deferral*);
 - (f) a reduction of the premium payable upon the redemption or repurchase of any Capital Security pursuant to Clause 11 (*Redemption and repurchase of the Capital Securities*);
 - (g) a change to the terms dealing with the requirements for Holders' consent set out in this Clause 15.4 (Majority, quorum and other provisions);
 - (h) an extension of the tenor of the Capital Securities or any delay of the due date for payment of any principal or interest on the Capital Securities;
 - (i) a mandatory exchange of the Capital Securities for other securities; and
 - (j) early redemption of the Capital Securities, other than as otherwise permitted or required by these Terms and Conditions.
- Any matter not covered by Clause 15.4.2, including for the avoidance of doubt the initiation of an Issuer Winding-up, shall require the consent of Holders representing more than fifty (50) per cent. of the Adjusted Nominal Amount for which Holders are voting at a Holders' Meeting or for which Holders reply in a Written Procedure in accordance with the instructions given pursuant to Clause 15.3.2. This includes, but is not limited to, any amendment to, or waiver of, the terms of the Terms and Conditions that does not require a higher majority (other than an amendment permitted pursuant to Clause 16.1(a) or (c)).
- 15.4.4 Quorum at a Holders' Meeting or in respect of a Written Procedure only exists if a Holder (or Holders) representing at least fifty (50) per cent. of the Adjusted Nominal Amount in case of a matter pursuant to Clause 15.4.2, and otherwise twenty (20) per cent. of the Adjusted Nominal Amount:
 - (a) if at a Holders' Meeting, attend the meeting in person or by other means prescribed by the Agent pursuant to Clause 15.2.4 (or appear through duly authorised representatives); or
 - (b) if in respect of a Written Procedure, reply to the request.

- 15.4.5 If a quorum exists for some but not all of the matters to be dealt with at a Holders' Meeting or by a Written Procedure, decisions may be taken in the matters for which a quorum exists.
- 15.4.6 If a quorum does not exist at a Holders' Meeting or in respect of a Written Procedure, the Agent or the Issuer shall convene a second Holders' Meeting (in accordance with Clause 15.2.1) or initiate a second Written Procedure (in accordance with Clause 15.3.1), as the case may be, provided that the person(s) who initiated the procedure for Holders' consent has confirmed that the relevant proposal is not withdrawn. For the purposes of a second Holders' Meeting or second Written Procedure pursuant to this Clause 15.4.6, the date of request of the second Holders' Meeting pursuant to Clause 15.2.1 or second Written Procedure pursuant to Clause 15.3.1, as the case may be, shall be deemed to be the relevant date when the quorum did not exist. The quorum requirement in Clause 15.4.4 shall not apply to such second Holders' Meeting or Written Procedure.
- 15.4.7 Any decision which extends or increases the obligations of the Issuer or the Agent, or limits, reduces or extinguishes the rights or benefits of the Issuer or the Agent, under the Terms and Conditions shall be subject to the Issuer's or the Agent's consent, as applicable.
- 15.4.8 A Holder holding more than one Capital Security need not use all its votes or cast all the votes to which it is entitled in the same way and may in its discretion use or cast some of its votes only.
- The Issuer may not, directly or indirectly, pay or cause to be paid any consideration to or for the benefit of any owner of Capital Securities (irrespective of whether such person is a Holder) for or as inducement to any consent under these Terms and Conditions, unless such consideration is offered to all Holders that vote in respect of the proposal at the relevant Holders' Meeting or in a Written Procedure within the time period stipulated for the consideration to be payable (such time period to be no less than ten (10) Business Days).
- 15.4.10 The Issuer may not, directly or indirectly, pay or cause to be paid any consideration to or for the benefit of any owner of Capital Securities (irrespective of whether such person is a Holder), nor make an offer to repurchase any Capital Securities, if receipt of such consideration or participation in such tender offer (as applicable) is conditional upon the Holder's consent to a proposal at a Holder's Meeting or in a Written Procedure.
- 15.4.11 A matter decided at a duly convened and held Holders' Meeting or by way of Written Procedure is binding on all Holders, irrespective of them being present or represented at the Holders' Meeting or responding in the Written Procedure. The Holders that have not adopted or voted for a decision shall not be liable for any damages that this may cause the Issuer or the other Holders.
- 15.4.12 All costs and expenses incurred by the Issuer or the Agent for the purpose of convening a Holders' Meeting or for the purpose of carrying out a Written Procedure, including reasonable fees to the Agent, shall be paid by the Issuer.
- 15.4.13 If a decision is to be taken by the Holders on a matter relating to the Terms and Conditions, the Issuer shall promptly at the request of the Agent provide the Agent with a certificate specifying the number of Capital Securities owned by Group Companies as per the Record Date for voting, irrespective of whether such person is a Holder. The Agent shall not be responsible for the accuracy of such certificate or otherwise be responsible for determining whether a Capital Security is owned by a Group Company.
- 15.4.14 Information about decisions taken at a Holders' Meeting or by way of a Written Procedure shall promptly be published on the websites of the Issuer and the Agent, provided that a failure to do so shall not invalidate any decision made or voting result achieved. The minutes from the relevant Holders' Meeting or Written Procedure shall at the request of a Holder be sent to it by the Issuer or the Agent, as applicable.

16. AMENDMENTS AND WAIVERS

- The Issuer and the Agent (acting on behalf of the Holders) may agree in writing to amend or waive any provision in the Terms and Conditions or any other document relating to the Capital Securities, provided that the Agent is satisfied that such amendment or waiver:
 - (a) is not detrimental to the interest of the Holders as a group;
 - (b) is made solely for the purpose of rectifying obvious errors and mistakes;
 - (c) is required by any applicable regulation, a court ruling or a decision by a relevant authority;
 - (d) is necessary for the purpose of having the Capital Securities admitted to trading on Nasdaq Stockholm (or any other Regulated Market, as applicable), provided that such amendment or waiver does not materially adversely affect the rights of the Holders; or
 - (e) has been duly approved by the Holders in accordance with Clause 15 (*Decisions by Holders*) and it has received any conditions precedent specified for the effectiveness of the approval by the Holders.
- The Issuer may substitute or vary the terms of all (but not some only) of the outstanding Capital Securities without any requirement for the consent or approval of the Holders, so that they become or remain, as applicable, Qualifying Securities, provided that such substitution or variation does not itself give rise to any right of the Issuer to substitute or vary the terms of the Capital Securities in accordance with Clause 11.6 (Substitution or variation due to a Tax Deductibility Event, a Rating Methodology Event or a Withholding Tax Event) in relation to the Qualifying Securities so substituted or varied.
- The consent of the Holders is not necessary to approve the particular form of any amendment or waiver to these Terms and Conditions. It is sufficient if such consent approves the substance of the amendment or waiver.
- The Agent shall promptly notify the Holders of any amendments or waivers made in accordance with Clause 16.1, setting out the date from which the amendment or waiver will be effective, and ensure that any amendments to these Terms and Conditions are available on the websites of the Issuer and the Agent. The Issuer shall ensure that any amendments to these Terms and Conditions are duly registered with the CSD and each other relevant organisation or authority.
- An amendment to the Terms and Conditions shall take effect on the date determined by the Holders' Meeting, in the Written Procedure or by the Agent, as the case may be.

17. THE AGENT

17.1 **Appointment of the Agent**

- 17.1.1 By subscribing for Capital Securities, each initial Holder appoints the Agent to act as its agent in all matters relating to the Capital Securities and the Terms and Conditions, and authorises the Agent to act on its behalf (without first having to obtain its consent, unless such consent is specifically required by these Terms and Conditions) in any legal or arbitration proceedings relating to the Capital Securities held by such Holder, including the winding-up, dissolution, liquidation, company reorganisation (*företagsrekonstruktion*) or bankruptcy (*konkurs*) (or its equivalent in any other jurisdiction) of the Issuer and in relation to any mandatory exchange of the Capital Securities for other securities (including, for the avoidance of doubt, a right for the Agent to subscribe for any such new securities on behalf of the relevant Holder). By acquiring Capital Securities, each subsequent Holder confirms such appointment and authorisation for the Agent to act on its behalf.
- 17.1.2 Each Holder shall immediately upon request provide the Agent with any such documents, including a written power of attorney (in form and substance satisfactory to the Agent), that the Agent deems necessary for the purpose of exercising its rights and/or carrying out its duties under the Terms and

- Conditions. The Agent is under no obligation to represent a Holder which does not comply with such request.
- 17.1.3 The Issuer shall promptly upon request provide the Agent with any documents and other assistance (in form and substance satisfactory to the Agent), that the Agent deems necessary for the purpose of exercising its rights and/or carrying out its duties under the Terms and Conditions.
- 17.1.4 The Agent is entitled to fees for all its work in such capacity and to be indemnified for costs, losses and liabilities on the terms set out in the Terms and Conditions and the Agency Agreement and the Agent's obligations as Agent under the Terms and Conditions are conditioned upon the due payment of such fees and indemnifications.
- 17.1.5 The Agent may act as agent or trustee for several issues of securities or other loans issued by or relating to the Issuer and other Group Companies notwithstanding potential conflicts of interest.

17.2 **Duties of the Agent**

- 17.2.1 The Agent shall represent the Holders in accordance with the Terms and Conditions.
- 17.2.2 When acting pursuant to the Terms and Conditions, the Agent is always acting with binding effect on behalf of the Holders. The Agent is never acting as an advisor to the Holders or the Issuer. Any advice or opinion from the Agent does not bind the Holders or the Issuer.
- 17.2.3 When acting pursuant to the Terms and Conditions, the Agent shall carry out its duties with reasonable care and skill in a proficient and professional manner.
- 17.2.4 The Agent shall treat all Holders equally and, when acting pursuant to the Terms and Conditions, act with regard only to the interests of the Holders as a group and shall not be required to have regard to the interests or to act upon or comply with any direction or request of any other person, other than as explicitly stated in the Terms and Conditions.
- 17.2.5 The Agent is always entitled to delegate its duties to other professional parties and to engage external experts when carrying out its duties as agent, without having to first obtain any consent from the Holders or the Issuer. The Agent shall however remain liable for any actions of such parties if such parties are performing duties of the Agent under the Terms and Conditions.
- The Issuer shall on demand by the Agent pay all costs for external experts engaged by it in connection with (i) matters relating to the Issuer or the Terms and Conditions which the Agent reasonably believes may be detrimental to the interests of the Holders under the Terms and Conditions, (ii) any Holders' Meeting or Written Procedure, or (iii) any amendment (whether contemplated by the Terms and Conditions or not) or waiver under the Terms and Conditions.
- 17.2.7 The Agent shall, as applicable, enter into agreements with the CSD, and comply with such agreement and the CSD Regulations applicable to the Agent, as may be necessary in order for the Agent to carry out its duties under the Terms and Conditions.
- Other than as specifically set out in the Terms and Conditions, the Agent shall not be obliged to monitor (i) the financial condition of the Issuer and the Group, (ii) the performance, default or any breach by the Issuer or any other party of its obligations under the Terms and Conditions, or (iii) whether any other event specified in the Terms and Conditions has occurred or is expected to occur. Should the Agent not receive such information, the Agent is entitled to assume that no such event or circumstance exists or can be expected to occur, provided that the Agent does not have actual knowledge of such event or circumstance.
- 17.2.9 The Agent shall ensure that it receives evidence satisfactory to it that the Terms and Conditions are duly authorised and executed. The Issuer shall promptly upon request provide the Agent with such documents and evidence as the Agent reasonably considers necessary for the purpose of being able to comply with this Clause 17.2.9. Other than as set out above, the Agent shall neither be liable to the Issuer or the Holders for damage due to any documents and information delivered to the Agent not

being accurate, correct and complete, unless it has actual knowledge to the contrary, nor be liable for the content, validity, perfection or enforceability of such documents.

- 17.2.10 Notwithstanding any other provision of the Terms and Conditions to the contrary, the Agent is not obliged to do or omit to do anything if it would or might in its reasonable opinion constitute a breach of any regulation.
- 17.2.11 If in the Agent's reasonable opinion the cost, loss or liability which it may incur (including reasonable fees to the Agent) in complying with instructions of the Holders, or taking any action at its own initiative, will not be covered by the Issuer, the Agent may refrain from acting in accordance with such instructions, or taking such action, until it has received such funding or indemnities (or adequate Security has been provided therefore) as it may reasonably require.
- 17.2.12 The Agent shall give a notice to the Holders:
 - (a) before it ceases to perform its obligations under the Terms and Conditions by reason of the non-payment by the Issuer of any fee or indemnity due to the Agent under the Terms and Conditions or the Agency Agreement; or
 - (b) if it refrains from acting for any reason described in Clause 17.2.11.
- 17.2.13 Subject to applicable regulations, the Agent shall promptly upon the reasonable request by a Holder forward by post any information from such Holder to the Holders which relates to the Capital Securities. The Agent may require that the requesting Holder reimburses any costs or expenses incurred, or to be incurred, by it in doing so (including a reasonable fee for its work).
- 17.2.14 Subject to the restrictions of a non-disclosure agreement entered into by the Agent in connection with these Terms and Conditions, the Agent is entitled to disclose to the Holders any document, information, event or circumstance directly or indirectly relating to the Issuer or the Capital Securities. Notwithstanding the foregoing, the Agent may if it considers it to be beneficial to the interests of the Holders delay disclosure or refrain from disclosing certain information.
- 17.2.15 If a committee representing the Holders' interests under the Terms and Conditions has been appointed by the Holders in accordance with Clause 15 (*Decisions by Holders*), the members of such committee may agree with the Issuer not to disclose information received from the Issuer, provided that it, in the reasonable opinion of such members, is beneficial to the interests of the Holders. The Agent shall be a party to such agreement and receive the same information from the Issuer as the members of the committee.

17.3 Liability for the Agent

- 17.3.1 The Agent will not be liable to the Holders for damage or loss caused by any action taken or omitted by it under or in connection with these Terms and Conditions, unless directly caused by its negligence or wilful misconduct. The Agent shall never be responsible for indirect or consequential loss.
- 17.3.2 The Agent shall not be considered to have acted negligently if it has acted in accordance with advice from or opinions of reputable external experts provided to the Agent or if the Agent has acted with reasonable care in a situation when the Agent considers that it is detrimental to the interests of the Holders to delay the action in order to first obtain instructions from the Holders.
- 17.3.3 The Agent shall not be liable for any delay (or any related consequences) in crediting an account with an amount required pursuant to the Terms and Conditions to be paid by the Agent to the Holders, provided that the Agent has taken all necessary steps as soon as reasonably practicable to comply with the regulations or operating procedures of any recognised clearing or settlement system used by the Agent for that purpose.
- 17.3.4 The Agent shall have no liability to the Issuer or the Holders for damage caused by the Agent acting in accordance with instructions of the Holders given in accordance with the Terms and Conditions.

17.3.5 Any liability towards the Issuer which is incurred by the Agent in acting under, or in relation to, the Terms and Conditions shall not be subject to set-off against the obligations of the Issuer to the Holders under the Terms and Conditions.

17.4 **Replacement of the Agent**

- 17.4.1 Subject to Clause 17.4.6, the Agent may resign by giving notice to the Issuer and the Holders, in which case the Holders shall appoint a successor Agent at a Holders' Meeting convened by the retiring Agent or by way of Written Procedure initiated by the retiring Agent.
- 17.4.2 Subject to Clause 17.4.6, if the Agent is Insolvent, the Agent shall be deemed to resign as Agent and the Issuer shall within ten (10) Business Days appoint a successor Agent which shall be an independent financial institution or other reputable company which regularly acts as agent under debt issuances.
- 17.4.3 A Holder (or Holders) representing at least ten (10) per cent. of the Adjusted Nominal Amount may, by notice to the Issuer (such notice shall, if given by several Holders, be given by them jointly), require that a Holders' Meeting is held for the purpose of dismissing the Agent and appointing a new Agent. The Issuer may, at a Holders' Meeting convened by it or by way of Written Procedure initiated by it, propose to the Holders that the Agent be dismissed and a new Agent appointed.
- 17.4.4 If the Holders have not appointed a successor Agent within ninety (90) days after:
 - (a) the earlier of the notice of resignation was given or the resignation otherwise took place; or
 - (b) the Agent was dismissed through a decision by the Holders,

the Issuer shall within thirty (30) days thereafter appoint a successor Agent which shall be an independent financial institution or other reputable company with the necessary resources to act as agent under debt issuances.

- 17.4.5 The retiring Agent shall, at its own cost, make available to the successor Agent such documents and records and provide such assistance as the successor Agent may reasonably request for the purposes of performing its functions as Agent under the Terms and Conditions.
- 17.4.6 The Agent's resignation or dismissal shall only take effect upon the earlier of:
 - (a) the appointment of a successor Agent and acceptance by such successor Agent of such appointment and the execution of all necessary documentation to effectively substitute the retiring Agent; and
 - (b) the period pursuant to Clause 17.4.4 having lapsed.
- 17.4.7 Upon the appointment of a successor, the retiring Agent shall be discharged from any further obligation in respect of the Terms and Conditions but shall remain entitled to the benefit of the Terms and Conditions and remain liable under the Terms and Conditions in respect of any action which it took or failed to take whilst acting as Agent. Its successor, the Issuer and each of the Holders shall have the same rights and obligations amongst themselves under the Terms and Conditions as they would have had if such successor had been the original Agent.
- 17.4.8 In the event that there is a change of the Agent in accordance with this Clause 17.4, the Issuer shall execute such documents and take such actions as the new Agent may reasonably require for the purpose of vesting in such new Agent the rights, powers and obligation of the Agent and releasing the retiring Agent from its further obligations under the Terms and Conditions and the Agency Agreement. Unless the Issuer and the new Agent agree otherwise, the new Agent shall be entitled to the same fees and the same indemnities as the retiring Agent.

18. THE ISSUING AGENT

- The Issuer shall when necessary appoint an Issuing Agent to manage certain specified tasks under these Terms and Conditions and in accordance with the legislation, rules and regulations applicable to and/or issued by the CSD and relating to the Capital Securities. The Issuing Agent shall be a commercial bank or securities institution approved by the CSD.
- The Issuer shall ensure that the Issuing Agent enters into agreements with the CSD, and comply with such agreement and the CSD Regulations applicable to the Issuing Agent, as may be necessary in order for the Issuing Agent to carry out its duties relating to the Capital Securities.
- The Issuing Agent will not be liable to the Holders for damage or loss caused by any action taken or omitted by it under or in connection with the Terms and Conditions, unless directly caused by its gross negligence or wilful misconduct. The Issuing Agent shall never be responsible for indirect or consequential loss.

19. THE CSD

- 19.1 The Issuer has appointed the CSD to manage certain tasks under these Terms and Conditions and in accordance with the CSD Regulations and the other regulations applicable to the Capital Securities.
- The CSD may retire from its assignment or be dismissed by the Issuer, provided that the Issuer has effectively appointed a replacement CSD that accedes as CSD at the same time as the old CSD retires or is dismissed and provided also that the replacement does not have a negative effect on any Holder or the admission to trading of the Capital Securities on the Regulated Market. The replacing CSD must be authorised to professionally conduct clearing operations pursuant to the Central Securities Depository Regulation (Regulation (EU) No 909/2014) and be authorised as a central securities depository in accordance with the Central Securities Depositories and Financial Instruments Accounts

20. NO DIRECT ACTIONS BY HOLDERS

- A Holder may not take any steps whatsoever against the Issuer to enforce or recover any amount due or owing to it pursuant to the Terms and Conditions, or to initiate, support or procure the winding-up, dissolution, liquidation, company reorganisation or bankruptcy in any jurisdiction of any Group Company in relation to any of the obligations and liabilities of the Issuer or any Subsidiary under the Terms and Conditions. Such steps may only be taken by the Agent.
- Clause 20.1 shall not apply if the Agent has been instructed by the Holders in accordance with the Terms and Conditions to take certain actions but fails for any reason to take, or is unable to take (for any reason other than a failure by a Holder to provide documents in accordance with Clause 17.1.2), such actions within a reasonable period of time and such failure or inability is continuing. However, if the failure to take certain actions is caused by the non-payment of any fee or indemnity due to the Agent under the Terms and Conditions or the Agency Agreement or by any reason described in Clause 17.2.11, such failure must continue for at least forty (40) Business Days after notice pursuant to Clause 17.2.12 before a Holder may take any action referred to in Clause 20.1.

21. TIME-BAR

- The right to receive repayment of the principal of the Capital Securities shall be time-barred and become void ten (10) years from the relevant Redemption Date. The right to receive payment of interest (excluding any capitalised interest) shall be time-barred and become void three (3) years from the relevant due date for payment. The Issuer is entitled to any funds set aside for payments in respect of which the Holders' right to receive payment has been time-barred and has become void.
- 21.2 If a limitation period is duly interrupted in accordance with the Swedish Act on Limitations (preskriptionslag (1981:130)), a new limitation period of ten (10) years with respect to the right to

receive repayment of the principal of the Capital Securities, and of three (3) years with respect to receive payment of interest (excluding capitalised interest) will commence, in both cases calculated from the date of interruption of the limitation period, as such date is determined pursuant to the provisions of the Swedish Act on Limitations.

22. NOTICES

- 22.1 Any notice or other communication to be made under or in connection with the Terms and Conditions:
 - (a) if to the Agent, shall be given at the address specified on its website www.nordictrustee.com on the Business Day prior to dispatch or, if sent by email by the Issuer, to the email address notified by the Agent to the Issuer from time to time;
 - (b) if to the Issuer, shall be given at the address registered with the Swedish Companies Registration Office on the Business Day prior to dispatch or, if sent by email by the Agent, to the email address notified by the Issuer to the Agent from time to time; and
 - (c) if to the Holders, shall be given at their addresses registered with the CSD on a date selected by the sending person which falls no more than five (5) Business Days prior to the date on which the notice or communication is sent, and by either courier delivery (if practically possible) or letter for all Holders. A Notice to the Holders shall also be published on the websites of the Issuer and the Agent.
- Any notice or other communication made by one person to another under or in connection with the Terms and Conditions shall be sent by way of courier, personal delivery or letter, or, if between the Issuer and the Agent, by email, and will only be effective:
 - (a) in case of courier or personal delivery, when it has been left at the address specified in Clause 22.1;
 - (b) in case of letter, three (3) Business Days after being deposited postage prepaid in an envelope addressed to the address specified in Clause 22.1; or
 - (c) in case of email, when received in readable form by the email recipient.
- 22.3 Any notice or other communication pursuant to the Terms and Conditions shall be in English.
- Failure to send a notice or other communication to a Holder or any defect in it shall not affect its sufficiency with respect to other Holders.

23. FORCE MAJEURE

- Neither the Agent nor the Issuing Agent shall be held responsible for any damage arising out of any legal enactment, or any measure taken by a public authority, or war, strike, lockout, boycott, blockade, natural disaster, insurrection, civil commotion, terrorism or any other similar circumstance (a "Force Majeure Event"). The reservation in respect of strikes, lockouts, boycotts and blockades applies even if the Agent or the Issuing Agent itself takes such measures, or is subject to such measures.
- 23.2 Should a Force Majeure Event arise which prevents the Agent or the Issuing Agent from taking any action required to comply with these Terms and Conditions, such action may be postponed until the obstacle has been removed.
- 23.3 The provisions in this Clause 23 apply unless they are inconsistent with the provisions of the Central Securities Depositories and Financial Instruments Accounts Act which provisions shall take precedence.

24. GOVERNING LAW AND JURISDICTION

- These Terms and Conditions, and any non-contractual obligations arising out of or in connection therewith, shall be governed by and construed in accordance with the laws of Sweden.
- The Issuer submits to the non-exclusive jurisdiction of the District Court of Stockholm (*Stockholms tingsrätt*).

SCHEDULE 1

CONDITIONS PRECEDENT

PART I – CONDITIONS PRECEDENT FOR INITIAL CAPITAL SECURITIES

1. CORPORATE DOCUMENTS

- (a) Copies of the articles of association and an up-to-date certificate of registration of the Issuer.
- (b) A copy of a resolution from the board of directors of the Issuer:
 - (i) approving the issue of the Initial Capital Securities, the terms of the Terms and Conditions and the Agency Agreement, and resolving to enter into such documents and any other documents necessary in connection therewith;
 - (ii) authorising a specified person or persons to execute the Terms and Conditions and the Agency Agreement; and
 - (iii) authorising a specified person or persons, on its behalf, to execute all documents and notices to be executed by it or in connection with the Terms and Conditions or the Agency Agreement
- (c) A copy of a passport or driver's license which includes a specimen of the signature of each person authorised by the resolution referred to in paragraph (b) above and actually signing the Terms and Conditions and/or the Agency Agreement.

2. FINANCE DOCUMENTS

- (a) A copy of the Terms and Conditions duly executed by the Issuer.
- (b) A copy of the Agency Agreement duly executed by the Issuer.

3. OTHER DOCUMENTS AND EVIDENCE

Such other documents and evidence as is agreed between the Agent and the Issuer.

PART II – CONDITIONS PRECEDENT FOR SUBSEQUENT CAPITAL SECURITIES

4. CORPORATE DOCUMENTS

- (c) Copies of the articles of association and an up-to-date certificate of registration of the Issuer.
- (d) A copy of a resolution from the board of directors of the Issuer:
 - (i) approving the issue of the Subsequent Capital Securities and resolving to enter into such documents and any other documents necessary in connection therewith;
 - (ii) authorising a specified person or persons to execute the documents necessary in connection with the Subsequent Capital Securities to which it is a party on its behalf; and
 - (iii) authorising a specified person or persons, on its behalf, to sign and/or dispatch all documents and notices to be signed and/or despatched by it under or in connection with the Subsequent Capital Securities.

DESCRIPTION OF SWEDAVIA AB (PUBL)

Business

The Company, Swedavia AB (publ), has been Sweden's state airport group since 1 April 2010, when the Swedish Civil Aviation Administration (*Luftfartsverket*) was divided into two parts: the airports and the air traffic service. The background to the Swedish Parliament's decision is the different business conditions that apply to each business. For Swedavia, the incorporation creates better business opportunities. The corporate form has a regulatory system that is adapted for companies that act in a commercial environment, it provides the conditions for a faster decision-making process and facilitates collaboration with other parties.

Swedavia is wholly owned by the Swedish state. Swedavia has the mission to operate and develop the ten airports that are part of the national basic supply and to contribute to good accessibility for Sweden and the country's regions. Swedavia's role is to create the accessibility that Sweden needs to facilitate travel, business and meetings – within Sweden, Europe and the world. Safety and sustainable development with the customer in focus is the starting point for everything Swedavia does, both in its own operations and in society at large. Together with its partners, Swedavia creates added value for its customers by offering attractive airports and accessibility that provide smooth and inspiring travel experiences. Swedavia's airports aims to be Scandinavia's most important meeting places.

In 2024, the Group had net revenue of SEK 6,393 million and had approximately an average number of 2,769 employees.

January-September 2025 in brief1

- During the first nine months of the year, Swedavia's airports had 25,2 million passengers (24,9 million), which is an increase of 1,2 per cent.
- Net revenue was SEK 5,099 million (SEK 4,814 million).
- Operating income was SEK 292 million (SEK 179).
- The Group's investments totalled SEK 1,553 million (SEK 1,335).

2024 in brief2

- During the year, Swedavia's airports had 32,5 million (32,1 million) passengers, which is an increase of 1,2 per cent compared with 2023, but only to 80 per cent. of the volume for the same period in 2019 prior to the COVID-19 pandemic.
- Net revenue totalled SEK 6,393 million (SEK 5,931 million).
- Operating income was SEK 26 million (SEK -318 million).
- Net income for the period was SEK -227 million (SEK -353 million). During the fourth quarter, the net income for the period was SEK -216 million (SEK -252 million).
- Cash flow from operating activities during the year was positive at SEK 1,071 million (SEK 766 million). During the fourth quarter, the cash flow from operating activities was SEK 293 million (SEK 286 million).
- Investments for the Group totalled SEK 1,790 million (SEK 2,010 million).
- The Board proposed that no dividend should be paid by Swedavia for the financial year 2024.

2023 in brief³

- During the year, Swedavia's airports had 32,1 million (27,6 million) passengers, which is an increase of 16,5 per cent compared with the same period in 2022. This corresponds to approximately 80 per cent. of the passenger volume for the same period in 2019 prior to the COVID-19 pandemic.
- Net revenue totalled SEK 5,931 million (SEK 4,846 million).
- Operating income was SEK -318 million (SEK -783 million).

¹ Figures in parentheses are results for the corresponding period for the previous year, except for liquidity, financial position, and contingent liabilities and pledged assets, where the comparison is with the opening balance for the previous year.

² Figures in parentheses are results for the corresponding period for the previous year, except for liquidity, financial position, and contingent liabilities and pledged assets, where the comparison is with the opening balance for the previous year.

³ Figures in parentheses are results for the corresponding period for the previous year, except for liquidity, financial position, and contingent liabilities and pledged assets, where the comparison is with the opening balance for the previous year.

- Net income for the period was SEK -353 million (SEK -709 million). During the fourth quarter, the net income for the period was SEK -252 million (SEK -543).
- Cash flow from operating activities during the year was positive at SEK 766 million (SEK 541 million). During the fourth quarter, the cash flow from operating activities was SEK 286 million (SEK 234 million).
- Investments for the Group totalled SEK 2,010 million (SEK 1,814 million).
- Swedavia issued capital securities in November in an aggregate amount of SEK 2,000 million.
- The Board proposed that no dividend should be paid by Swedavia for the financial year 2023.

The Issuer

Swedavia AB (publ), Swedish Corporate ID No. 556797-0818 and LEI code 529900ZERGNJBEQEX179, based in Sigtuna Municipality, Sweden, is a Swedish public limited company governed by the Swedish Companies Act and registered at the Swedish Companies Registration Office (*Bolagsverket*). The Company was incorporated in Sweden on 1 April 2010. The Company's website is www.swedavia.se. The information on the Issuer's website, and on any other website being referred to in this Prospectus, does not form part of this Prospectus unless such information is incorporated by reference into this Prospectus and has not been reviewed or approved by the SFSA.

Address

Swedavia AB (publ) SE-190 45 Stockholm-Arlanda, Sweden Visiting address: Flygvägen 1

Telephone and fax Tel +46 (0)10 109 00 00 Fax: +46 (0)10 109 05 00

Legal structure of the Group

Swedavia is the parent company in the Group and has one subsidiary; Swedavia Real Estate AB which is 100 per cent. owned as of the date of this Prospectus. Swedavia Real Estate AB is in turn the parent company of a property group. In addition, Swedavia has an associated company through Swedavia Real Estate AB; Swedish Airport Infrastructure AB, which is a strategic collaboration in the form of joint venture together with the occupational pension provider Alecta, which means that it is a jointly owned company where Swedavia does not have a controlling interest.

The parent company's operations consist of airport operations and group-wide functions and organisation for the management of the subsidiaries. Most of the borrowing takes place in the parent company.

Principal shareholder

All shares in the Company are held by the Swedish state. The state ownership in Swedavia is managed by the Ministry of Finance (*Finansdepartementet*).

Share capital

The Company's share capital amounts to SEK 1,441,403,026 divided into 1,441,403,026 shares with a quotient value of SEK 1 per share. The Company has only one class of shares. The principal property of the shares is to give holders the right to a share of the Company's equity and the right to vote at the General Meeting, where each share gives right to dividend and one vote.

Credit rating

The Issuer has on 5 November 2025 been assigned a long term issuer rating of Baa1 with positive outlook from Moody's. Moody's is established in the European Union and is registered under Regulation (EC) No. 1060/2009 (as amended) (the EU CRA Regulation). A credit rating is not a recommendation to buy, sell or hold securities and may be subject to revision, suspension, reduction or withdrawal at any time by the assigning rating agency.

The following table sets out the possible long-term ratings assigned by Moody's.

Aaa	
Aa1	
Aa2	
Aa3	
A1	
A2	
A3	
Baa1	
Baa2	
Baa3	
Ba1	
Ba2	
Ba3	
B1	
B2	
В3	
Caa1	
Caa2	
Caa3	
Ca	
С	

BOARD OF DIRECTORS, GROUP MANAGEMENT AND AUDITOR

Board of Directors

The Board of Directors of the Issuer consists of nine members elected by the Annual General Meeting, two regular employee representatives and their two deputies, elected by the employee organisations.

Ulrika Francke

Born in 1956. Chairman of the Board (elected 2024). Chairman of the Remuneration Committee, member of the Finance- and Audit Committee.

Other on-going assignments/positions: Chairman of the Board of Vasakronan AB and Svenska Institutet för Standarder (SIS), board member of Circura Holding AB, Liquid Wind AB, Sven Tyrén's Foundation, KREAB, Stiftelsen Stockholms Framtid and Brf Blåbäret.

Josefin Karlsson

Born in 1984. Board member (elected 2025). Member of the Finance and Audit Committee

Other on-going assignments/positions: Board member of Svevia AB (publ).

Karl Sandlund

Born in 1977. Board member (elected 2025).

Other on-going assignments/positions: None. Byggmax Group (President & CEO Byggmax Group).

Tor Clausen

Born 1962. Board member (elected 2020). Member of the Finance- and Audit Committee.

Other on-going assignments/positions: Chairman of the Board of Wexthuset AB, board member of Goldmarsh AB and Paradisparkering AB.

Nina Linander

Born 1959. Board member (elected 2020). Chairman of the Finance- and Audit Committee.

Other on-going assignments/positions: Board member of Asker Healthcare Group, Suominen Corporation and Vattenfall AB.

Eva Nygren

Born 1955. Board member (elected 2020).

Other on-going assignments/positions: Board member of Troax Group AB, NRC Group ASA, Prince Eugen's Waldemarsudde Foundation and Brekke & Strand Akustikk AS.

Per Sjödell

Born 1972. Board member (elected 2020). Member of the Remuneration Committee

Other on-going assignments/positions: Chairman of the Board of Identity Works AB, IW Family AB, IW Edition AB, Equestrian Stockholm AB, Parfym Sverige AB / Parfym Sverige Holding AB and OBAYATY AB, board member of Spendrups Bryggeriaktiebolag, Visit Sweden, Geins (vice chairman) and PostNord AB.

Annica Ånäs

Born 1971. Board member (elected 2020).

Other on-going assignments/positions: CEO of Atrium Ljungberg AB and board member of Kojamo Oyj and Fastighetsägarna Sverige AB.

Lars Mydland

Born 1954. Board member (elected 2014).

Other on-going assignments/positions: Special Adviser to the Board of Directors of Veling Ltd.

Robert Olsson

Born 1966. Employee representative (elected 2010).

Other on-going assignments/positions: Section Chairman of ST.

Agne Lindbom

Born 1961. Employee representative (elected 2018).

Other on-going assignments/positions: Representative of the negotiation organisation of SEKO Swedavia.

Jessica Lindstedt

Born 1993. Deputy employee representative.

Ida Sjögren

Born 1979. Deputy employee representative.

Group Management

On the date of this Prospectus, the Group Management consists of nine persons.

Name	Position	
Jonas Abrahamsson ⁴	Chief Executive Officer (CEO)	
Ali Sadeghi	Director of Facilities and Systems	
Anna Bovaller	Chief Legal Counsel (CLO)	
Charlotte Ljunggren	Marketing Director	
Marie Wiksborg	Director Business Support	
Susanne Norman	Operational Director	
Karin Öhrström	Director Regional Airports	
Kristina Ferenius	Chief Financial Officer (CFO)	
Fredrik Jaresved	Director Strategic Initiatives & Innovation	

Conflicts of interest

No member of the Board of Directors or the Group Management or any other senior executive has any private interest that might conflict with the Issuer's interests.

Business address

The office address for all board members and all members of the Group Management is Swedavia Arlanda Airport, SE-190 45 Stockholm-Arlanda, Sweden.

Auditor

KPMG AB (Vasagatan 16, Box 382, SE-101 27 Stockholm, Sweden) is Swedavia's auditor since the 2019 annual general meeting. On 25 April 2025, at the 2025 annual general meeting, KPMG was re-elected as Swedavia's auditor until the 2026 annual general meeting. Tomas Gerhardsson is auditor in charge. Tomas Gerhardsson is an authorised public accountant and member of FAR, the professional institute for accountants in Sweden.

⁴ Jonas Abrahamsson has announced his decision to leave Swedavia, with his resignation taking effect at the end of February 2026. Swedavia's Board of Directors has appointed Mats Johannesson as the new CEO, and he will assume the role no later than the beginning of May 2026.

LEGAL CONSIDERATIONS AND SUPPLEMENTARY INFORMATION

Approval by the Swedish Financial Supervisory Authority

The Prospectus has been approved by SFSA as competent authority under Regulation (EU) 2017/1129. The SFSA only approves this Prospectus as meeting the standards of completeness, comprehensibility and consistency imposed by Regulation (EU) 2017/1129. The SFSA's approval should not be considered as an endorsement of the Issuer that is the subject of this Prospectus, nor should it be considered as an endorsement of the quality of the securities that are the subject of this Prospectus. Investors should make their own assessment as to the suitability of investing in the securities.

This Prospectus is valid for a maximum of twelve (12) after the date of the approval of the Prospectus, provided that it is completed by any supplement pursuant to Article 23 of the Prospectus Regulation. The obligation to supplement this Prospectus in the event of significant new factors, material mistakes or material inaccuracies does not apply from the time when trading of the Capital Securities on a Regulated Market begins.

Authorisation and responsibility

The Issuer has obtained all necessary resolutions, authorisations and approvals required in conjunction with the Capital Securities and the performance of its obligations relating thereto. The issuance of the Capital Securities on 19 November 2025 was authorised by a resolution by the Board of Directors of the Issuer on 30 October 2025.

The Issuer accepts responsibility for the information contained in this Prospectus and declares that, to the best of its knowledge, the information contained in this Prospectus is in accordance with the facts and the Prospectus makes no omission likely to affect its import. The Board of Directors of the Issuer is, to the extent provided by law, responsible for the information contained in this Prospectus and declares that, to the best of its knowledge, the information contained in this Prospectus is in accordance with the facts and the Prospectus makes no omission likely to affect its import.

Material contracts

There are no material contracts that Swedavia has entered into outside of the ordinary course of its business, which could result in Swedavia being under an obligation or entitlement that is material to its ability to meet its obligations to Holders in respect of the Capital Securities.

Governmental, legal and arbitration proceedings

Swedavia neither is nor has been a party to any governmental, legal or arbitration proceedings (including any such proceedings which are pending or threatened of which the Issuer is aware) during the previous twelve months which may have, or have had in the recent past, significant effects on the Issuer's and/or the Group's financial position or profitability. Swedavia is, however, from time to time subject to disputes and legal proceedings in the ordinary course of its business.

Certain material interests

The Joint Bookrunners (and closely related companies) have provided, and may in the future provide, certain investment banking and/or commercial banking and other services to the Issuer and the Group for which they have received, or will receive, remuneration. Accordingly, conflicts of interest may exist or may arise as a result of the Joint Bookrunners having previously engaged, or in the future engaging, in transactions with other parties, having multiple roles or carrying out other transactions for third parties.

Mannheimer Swartling Advokatbyrå is Swedavia's legal advisor in connection with the issuance and admission to trading of the Capital Securities.

Trend information

There has been no material adverse change in the prospects of the Issuer since 31 December 2024, being the date of the last published audited annual financial statements of the Issuer.

There has been no significant change in the financial performance of the Group since 30 September 2025, being the end of the last financial period for which financial information has been published to the date of this Prospectus.

Significant changes since 30 September 2025

Apart from the issuance of the Capital Securities, there have been no significant changes in the financial position of the Group since 30 September 2025, being the end of the last financial period for which interim financial information of the Issuer has been published.

Incorporation by reference

The following information has been incorporated into this Prospectus by reference and should be read as part of the Prospectus.

The Group's Annual Reports have been prepared in accordance with IFRS and the Swedish Annual Accounts Act (*årsredovisningslagen*) and the Group's interim report for the period January–September 2025 has been prepared in accordance with IAS 34 Interim Financial Reporting and the Swedish Annual Accounts Act. The Group's interim report for the period January–September 2023 has not been audited, but reviewed by the Company's current auditor, KPMG AB.

With the exception of the abovementioned Annual Reports and interim report, no information in this Prospectus has been audited or reviewed by the Company's current or previous auditor.

Non-incorporated parts of the abovementioned reports contain information presented elsewhere in this Prospectus or which is deemed not relevant to investors.

Certain financial and other information presented in this Prospectus has been rounded off for the purpose of making this Prospectus more easily accessible for the reader. As a result, the figures in tables may not tally with the stated totals.

The following sections of the Group's interim report for the period January–September 2025 (https://mb.cision.com/Main/22898/4259274/3754806.pdf) are incorporated:

- Consolidated income statement, p. 12
- Consolidated balance sheet, summary p. 13
- Consolidated change in equity, p. 14
- Consolidated cash flow statement, p. 14
- Notes, p. 17-22
- Auditor's review report, p. 26

The following sections of the Group's Annual Report for the financial year 2024 (https://www.swedavia.com/globalassets/ahr/2025/swedavia-annual-and-sustainability-report-2024.pdf) are incorporated:

- Consolidated income statement, p. 66
- Consolidated balance sheet, p. 67
- Consolidated change in equity, p. 68
- Consolidated cash flow statement, p. 69
- Notes, p. 74–107
- Auditor's report, p. 109-111

The following sections of the Group's Annual Report for the financial year 2023 (https://www.swedavia.com/globalassets/ahr/2024/swedavia-annual-and-sustainability-report-2023.pdf) are incorporated:

- Consolidated income statement, p. 70
- Consolidated balance sheet, p. 71
- Consolidated change in equity, p. 72
- Consolidated cash flow statement, p. 73
- Notes, p. 78–112

• Auditor's report, p. 114-116

Third party information

The Issuer confirms that the information sourced from third parties has been accurately reproduced and that as far the Issuer is aware and is able to ascertain from information published by the relevant third parties, no facts have been omitted which would render the reproduced information inaccurate or misleading. The information sourced from third parties has not been audited and has not been scrutinised or approved by the SFSA.

Documents available

Swedavia's articles of association and certificate of registration are available on Swedavia's website, https://www.swedavia.com/about-swedavia/corporate-governance/ (the information on the website does not form part of this Prospectus unless that information is incorporated by reference into this Prospectus).

The Terms and Conditions are available on Swedavia's website, https://www.swedavia.com/about-swedavia/financial-information/ (the information on the website does not form part of this Prospectus unless that information is incorporated by reference into this Prospectus).

THE ISSUER

Swedavia AB (publ)

Visiting address: Flygvägen 1, 190 60, Stockholm-Arlanda, Sweden Postal address: SE-190 45 Stockholm-Arlanda, Sweden

Telephone: +46 (0)10 109 00 00 www.swedavia.se

ISSUING AGENT

Nordea Bank Abp, filial i Sverige

Smålandsgatan 17 SE-105 71 Stockholm www.nordea.com

JOINT BOOKRUNNERS

Nordea Bank Abp

Skandinaviska Enskilda Banken AB (publ)

Satamaradankatu 5 FI-00020 Nordea, Helsinki, Finland www.nordea.com Kungsträdgården 8 SE-106 40 Stockholm, Sweden www.sebgroup.com

LEGAL ADVISOR TO THE ISSUER

Mannheimer Swartling Advokatbyrå

Norrlandsgatan 21 SE-111 87 Stockholm, Sweden www.mannheimerswartling.se

AGENT

THE CSD

Nordic Trustee & Agency AB (publ)

Euroclear Sweden AB

Norrlandsgatan 23 SE-111 43 Stockholm, Sweden www.nordictrustee.com Klarabergsviadukten 63 SE-101 23 Stockholm, Sweden www.euroclear.com/sweden